Editors Janez Prašnikar Tjaša Redek Fatmir Memaj

Albania: The Role of Intangible Capital in Future Growth

Ljubljana, 2012

Albania: The Role of Intangible Capital in Future Growth

Faculty of Economics - monography Code: PMR12ZM112

Publisher:	Faculty of Economics Ljubljana Publishing Office
Printed and Bound by:	Copis, Ljubljana, Dunajska 158
First edition:	200 copies
Design by:	Robert Ilovar
Reviewers:	prof. dr. Andreja Cirman
	prof. dr. Primož Dolenc
Proofread by:	Shanna Pafford Groznik

CIP - Kataložni zapis o publikaciji Narodna in univerzitetna knjižnica, Ljubljana

316.42(496.5)

Albania: The Role of Intangible Capital in Future Growth / editors Janez Prašnikar, Tjaša Redek, Fatmir Memaj. - 1st ed. - Ljubljana : Faculty of Economics, 2012

ISBN 978-961-240-239-6

1. Prašnikar, Janez, 1950-

262919680

All rights reserved. No part of this publication may be reproduced or transmitted in any form by any means, electronic, mechanical or otherwise, including (but not limited to) photocopy, recordings or any information or retrieval system, without the express written permission of the author or copyright holder.

Acknowledgements

The research presented in this book is a result of intense research work within the scope of several projects. Public funding supported these projects. The following research projects were partially supported by the Slovenian Research Agency (ARRS):

- 1. J5-4169 (A) Analysis of firm-level investment in tangible and intangible capital from the perspective of future competitive advantages of Slovene firms
- 2. J5-2227 (B) The influence of ownership and employee participation in equity, profit and management on the economic performance of firms in countries of the former Yugoslavia

The following study was partially financed by the Ministry of Foreign Affairs of the Republic of Slovenia:

1. The influence of ownership and employee participation in equity, profit and management on the economic performance of firms in countries of the former Yugoslavia

We would like to thank the Albanian Agency for Research and Information Technology, who helped finance the survey among Albanian companies.

A team of researchers from Albania also supported our research work: prof. dr. Kozeta Sevrani (Bylykbashi), prof. dr. Andrea Koxhaj, prof. dr. Mimoza Skënderi (Kasimati), dr. Petrit Dollani. We are grateful for their assistance.

We would also like to thank both reviewers. Their comments helped us improve the book.

i

Table of contents

Janez Prašnikar, Tjaša Redek, Fatmir Memaj: INTRODUCTION1
<i>Tjaša Redek, Fatmir Memaj, Janez Prašnikar, Domen Trobec:</i> ALBANIA: TWO DECADES OF ECONOMIC DEVELOPMENT AT A GLANCE 4
Janez Prašnikar, Tjaša Redek: INTANGIBLE CAPITAL IN ALBANIA: RESEARCH METHODOLOGY
Matjaž Koman, Gordana Lalović: RELATIONAL, INFORMATIONAL, AND ITC CAPITAL
Vesna Žabkar, Fatmir Memaj: BRANDING AND BRAND CAPITAL83
<i>Tjaša Redek:</i> THE CASE OF R&D99
Janez Prašnikar, Fatmir Memaj, Damjan Voje: SOCIAL CAPITAL AND CORPORATE GOVERNANCE124
Nada Zupan, Daša Farčnik: INVESTMENT IN HUMAN CAPITAL 141
Janez Prašnikar, Aljoša Valentinčič, Andraž Ušlakar: FIRMS' FINANCIAL POLICIES

ii

INTRODUCTION

Albania has made significant progress since the beginning of its transition. Its GDP per capita has risen several times and the country has restructured a large share of its productive means. If economic growth has been, until now, achieved primarily through restructuring and export orientation of cost competitive low value added industries, its future growth strategy, still export oriented, will have to shift towards higher value added production in order to further excel GDP growth and the transition of the country towards middle income countries.

Successful economic growth in transition economies and many other emerging economies largely depends not only on increasing export orientation of the economy, but also on moving production towards more demanding segments. For example in Korea, industrialization was based on producing low value added products like simple products in electronics and machinery. But the strategic orientation of both companies and, more importantly, the state, helped Korea become a leading producer in many segments of the global market, thus taking the primary position from developed countries for some products (such as LCD television screens produced by LG, formerly led by the Japanese and also Samsung; steel-making by the company POSCO) and strengthening its position with other products (such as car industry companies Hyundai and Kia) with the desire to further excel based on its global growth. The trickle-down impact of a few strong, more technologically advanced export-oriented companies is also far from negligible. Domestic value chains can benefit from higher demand for their

products, but the impact of these companies primarily opens opportunities in terms of future sales in global markets, access to new knowledge, and technology, thus increasing growth potential.

Export orientation is often first based on cost competition, but successful experiences of Asian and European economies show that own product development and global market penetration with own brands pushes the economy's developmental frontier up. Both innovation and branding represent key components of intangible capital. Intangible capital is a major source of productivity growth and increased value added. According to the most recent research in the field, intangible capital contributes up to one-third of productivity growth (e.g. Corrado et al., 2009¹). Also, it is one of the major explanatory factors of 'new economy' success. Although the majority of research is being done for developed economies, developing economies also benefit from intangible capital, although the awareness of its importance and actual investment in companies are generally lower (Prašnikar (ed.), 2010²).

The research of intangible capital in Albania is the first of its kind in the country, but it is a part of the research on intangible capital in the Western Balkan region. The methodology applied across the region was developed with a focus on the characteristics and role of intangible capital in developing countries. Consequently, the study in many aspects contributes to the theoretical, methodological and empirical literature.

² Prašnikar, J. (ed.) (2010). The role of intangible assets in exiting the crisis. Ljubljana: Časnik Finance.



¹ Corrado, C., Hulten, C., & Sichel, D. (2009). Intangible Capital and U.S. Economic Growth. Review of Income and Wealth, 55, 661-685

The book is divided into two major parts: an overview of economic development in Albania after the Second World War with a focus on transition and a study of intangible capital in Albanian companies, and an empirical analysis of the situation in 2011.

The book comprises nine chapters. Following this introduction, the second chapter deals with the macroeconomic development in Albania, in the last 20 years. Although the progress has been significant, both Albanian companies and, consequently, the entire economy are still quite inward oriented despite the obvious reorientation towards global markets. Investment in intangible capital would significantly contribute to future restructuring and growth of the Albanian economy.

Investment in intangible capital is the focus of the second part of the book. In the third chapter, first the methodology of the analysis is briefly discussed, defining the components of intangible capital and describing the questionnaire, the sample and data collection. The chapters that follow, chapters four to nine, provide a detailed analysis of the intangible capital components. Chapter four analyzes information and relationship capital and IT, and chapter five discusses branding. Chapter six provides analysis of research and development activity in Albanian firms, chapter seven deals with social capital, focusing on owners-workers-managers relationships. The analysis of human resource management in chapter eight provides additional insight on the topic with also a focus on organization itself. Last, the characteristics of financial activities are analyzed in chapter nine.



Tjaša Redek, Fatmir Memaj, Janez Prašnikar, Domen Trobec

ALBANIA: TWO DECADES OF ECONOMIC DEVELOPMENT AT A GLANCE

1 Introduction

Albania is a small Balkan country with a very rich and diverse history. The 'land of the eagle' has, according to Jeffries (1990, p. 77), felt a profound impact of foreign domination, from Roman and Byzantinian rule, through almost five centuries of Ottoman domination ending in 1912, to Italian invasion in 1939. In 1944 the country attained independence, and in 1946 The People's Republic, led by Enver Hoxha, was founded and communism, following the Soviet model, was implemented.

The pre-war Albania was one of the least developed countries in Europe. According to Schnytzer (1982, p.1, in Jeffries, 1990, p. 78), in 1938 industry accounted only for 4.4 percent of national income, agriculture was the major source of value and exports; about two-thirds of it was primarily livestock. There were only 150 industrial enterprises and 50 percent of them employed less than 10 employees. The majority of the population, 85 percent, was rural with a life

expectancy of 38 years. The country is richly endowed with natural resources, among which Jeffries (1990, p. 78) lists primarily fertile land and energy resources (hydroelectric and also coal, oil and gas). The country is also rich with other mineral resources, such as chrome, nickel, copper, iron and manganese (National Agency of Mineral Resources, 2010).

Natural resources partially were an important base for the fast, primarily industrial growth, during communism. But the period of communism in Albania is remembered largely for other reasons. According to Zickel and Iwaskiw (1992, p.xxxv), Albania was 'for more than forty years one of the most obscure and reclusive countries in the world. A totalitarian communist regime, led by party founder and first secretary Enver Hoxha from 1944 until his death in 1985, maintained strict control over every facet of the country's internal affairs, while implementing a staunchly idiosyncratic foreign policy.' Jeffries (1990, p. 84) also adds that Albania banned all religious institutions and practices in 1967 and became the first atheist state in the world. There was no legal private sector in manufacturing (in other socialist¹ economies a limited amount was usually allowed), but there was one of the most rigid of its kind. Its characteristics severely marked Albanian development in the past and also set the foundations for its post-socialist development (Jeffries, 1990, p. 84).

Albania was the last country to step on the path of transition (Muco, 1997, p. 18). The first step towards changes, in at the time very rigid communism was marked with the death of Enver Hoxha in 1985. The results of the reforms were poor. But

¹ The term 'socialist' will be used to denote the period after the Second World War. This is the predominant expression used in Albania. The more extreme notation 'communism' is only used by the most vigorous critics. Also, communism never existed in real life, at least not as Marx envisaged it. 'Communism' will consequently be used only when giving direct citation of sources.

in 1990 the political elite, led by Ramiz Alia and 'swayed by large-scale student demonstrations, strikes, and the exodus of thousands of Albanians to Italy and Greece, and fearing the prospect of a violent overthrow,' allowed a multiparty system (Zickel and Iwaskiw, 1992, p. xxxv). The first more serious steps towards reforms were made, according to Muco (1997, p. 18) in 1991, after the first democratic election. After initial political difficulties a reform plan, comprised of typical transition reforms (a series of laws about price liberalization, investment deregulation, macroeconomic intervention, as a tight monetary policy and budgetary austerity, land distribution and small-scale privatization), was prepared. The results of the reforms were dramatic, as in many other countries. Output declined by 50 percent in just two years (1990-92) and the country's budget deficit also reached staggering heights in both 1991 and 1992 with 44 and 50 percent of GDP. The country went through an unstable period during the 1990s, weathering a second decline at the end of the decade. But the new millennium brought about strong growth ranging between five to seven percent per year. Despite the initial shock, the reforms eventually brought results.

Albania, a parliamentary democracy, remains for the moment one of the minor countries in the region. With a surface of roughly 27 thousand square kilometers (World Atlas, 2011), it represents about 0.2 percent of Europe. The majority of its approximately 3.2 million inhabitants is rural population (World Bank, 2012). With a per capita GNI of 4,000 US\$ in 2010, the country was ranked 124th out of 213 places by the World Bank among countries like Bosnia and Herzegovina (110th, 4,790 US\$), Macedonia (116th, 4,520 US\$), Peru and the Maldives, and well behind the richest countries like Monaco (1st, 197,590 US\$), Liechtenstein (2nd, 136,540 US\$) and Norway (3rd, 85,380 US\$). In purchasing power terms, the country ranked 109th among 215 countries in 2010 with a per capita GNI of 8,840 US\$ (World Bank, 2011). Despite a relatively strong performance after transition,

poverty is still an important problem with 12 percent of the population living below the poverty line (Table 1). The developmental lag with the richer population is evident also from the low urban population share. Agriculture remains an important sector in Albania with over one-third of GDP.

	Albania	Europe and Central Asia	Upper-middle income countries				
	2009						
Population (2010, mid-year, millions)	3.2	405	2452				
GNI per capita (2010, Atlas method, US\$)	3960	7272	5884				
GNI (2010, Atlas method, US\$ bn)	12.7	2947	14429				
Most recent available estimate, 2004-2010							
Poverty (% of population below national poverty line)	12	-	-				
Urban population (% of total)	48	64	57				
Life expectancy at birth (in years)	77	71	73				
Literacy (in the age group 15 years or older, %)	96	98	93				

Table 1: Albania at a glance

Source: World Bank, 2012.

Currently, the country is still undergoing the process of economic restructuring, but the outlook is quite positive with strong growth in the past decade. Even during the general economic decline due to financial turmoil in global markets, Albania continued to record positive growth between 2007 and 2010. Its future depends largely on the success of continuing reforms and investment patterns.

To understand the current economic processes in Albania, the general macroeconomic setting must be available. In continuing, the general macroeconomic developments are first described, followed by a more detailed

analysis of the reforms with a focus on privatization and new entry and industrial developments. The analysis is further illustrated by several company case studies, which explain the situation in terms of intangible capital in Albanian companies.

2 Macroeconomic developments

Albania is one of the transition economies. Its developmental characteristics and problems are largely linked to its past, and its comparative performance in reform patterns are comparable to other transition countries.

Figure 1: GDP per capita in US dollars in 2009 in transition countries*



* Data for Moldova was not available (0.0 value was provided). Data: EBRD, 2011a.

The country ranked 9th poorest among 28 transition economies in 2009 with roughly 3,800 US\$² of GDP per capita, which is among the poorest third of transition countries (Figure 1). It was also the second poorest economy among Central and Eastern (CEE) and South-Eastern European (SEE) countries; only Kosovo was poorer with slightly less than 1,800 US\$ per capita.

8

² Data from the EBRD and the World Bank differ, because the World Bank uses the Atlas method of currency conversion (for details, see the World Bank's web page), while the EBRD uses current exchange rates.

Nonetheless, Albania has made significant progress since the beginning of transition; its GDP in 1989 was 730 US\$ (Figure 2) (all data EBRD, 2011a, in current prices). Economic growth followed the Southeast European transition pattern: initial decline was followed by a period of growth, but the end of 1990s was again marked by a decline. This brief downturn, though, was later compensated by strong growth (for reforms see Table 3).



Figure 2: GDP per capita in US dollars, 1989-2010

The economic situation in Albania is a result of the country's specific economic situation before transition and its transition reform packages. In order to understand the current position, let us briefly analyze the past.

2.1 Pre transition situation

Muco (1997) claims that despite being a socialist and centralized economy, Albania did not consider itself as a consistent part of the 'Eastern socialist bloc'. The

country remained an economic mystery to researchers due to the lack of data and research, especially to foreign observers. But it was obvious that the authorities implemented a rigid socialist model based on Marxist ideas and Stalin's implementations (Schnytzer, 1982 and 1992 in Muco, 1997, p. 6) that were governing the socio-economic life of the country. After the Second World War and during the 45 years of communism, the country was organized in a rigorously centralized manner: 'all economic decisions on production, pricing, wage setting, investment and external trade were centralized, while changes between the plans were generally minimal' (Muco, 1997, p. 7). According to Zickel and Iwaskiw (1992, pp. 105-106), the resulting poverty was actually an ironic outcome, a monument to socialist mismanagement. Namely, Albania has significant fossil fuel and mineral deposits and huge hydroelectric potential. Also, the fertile lowlands offer ideal conditions for cultivation of fruits and vegetables. Unfortunately, the mismanaged industrial expansion resulted in poverty and staggering unemployment in the country with the highest birth rate in Europe, and in the beginning of the 1990s numerous people fled abroad seeking opportunities.

The economic situation in Albania was, according to Muco (1997, pp. 8-12), marked by several deeply-rooted problems arising from socialist practices. The most prominent feature of the Albanian economy was its desire for self-sufficiency (see Jeffries, 1990, Muco, 1997), which led to extreme production diversification and a complete ignorance of the comparative advantages principle. Lacking suitable technological support and tradition in industrial branches, manufacturing was marked by inefficiency, low productivity and high costs. If industry represented much less than 10 percent of output prior to World War II (4.4 percent in the 1920s according to Jeffries, 1990), a strategy based on industrialization led to the rise of the industrial sector, which in 1989 represented 41.7 percent of output, and agriculture kept a solid one-third of output with over 35 percent (EBRD, 2011a).

The industry employed one-fifth of the population and absorbed over 40 percent of investment. The strategic goal was to develop heavy industry, electricity and mineral production (which were very important export products), machinery, metallurgical and chemical industry, while light industries and food processing were given less attention (Muco, 1997, p. 10). Agriculture was comparatively neglected despite its size in terms of already mentioned output and employment (over 50 percent), although efforts had been made in increasing arable land. The country was relatively self-sufficient in food production (85 percent of total food consumption, but bread production remained a problem (Muco, 1997, p. 11), but due to backward technology the production was inefficient. Also, given the rugged mountain terrain, the country was too focused on crops and much too little on livestock. Private property in both agriculture and industry was virtually nonexistent³, although some form of private property in both sectors was normally allowed in socialism, even in the Soviet Union (for country details see Jeffries, 1990).

The economic results of socialism are displayed in Table 2. Initially, high investment rate growth led to high growth of industrial production and high output growth, but growth was clearly extensive, as evidenced by the industrial labor productivity data. Agricultural growth was also significant. But, given the socialist emphasis on industrial growth, agricultural growth, as expected, lagged behind industrial growth. In the 1980s growth slowed down as economic problems culminated.

³ In agriculture, a private plot of 200 square meters and a certain number of live-stock was allowed. In the 1980s there was much experimentation with private plots, collectivization, brigade plots and cattle, and a reversal of collectivization in the second half of the 1980s (see Jeffries, 1990, and Muco, 1997). Each household could have 10 animals, provided these animals were sheep or goats. One cow was equivalent to 10 sheep/goats. Chickens were not counted.

¹¹

	1961-70	1971-80	1981-88
Net material product	7.4	4.6	1.7
Net material product per capita	4.4	2.2	-0.3
Gross industrial production	9.8	7.5	2.8
Industrial labor productivity	1.5	1.8	1.3
Gross agricultural production	6.0	3.8	1.5
Agricultural labor productivity	1.0	-0.2	-2.0
Gross investment	8.4	4.9	1.5

Table 2: Key economic indicators, 1961-88, average annual increase in percentage

Source: Zickel and Iwaskiw, 1992, p. 244.

According to Figure 3, the structure of the economy at the end of the socialist era in 1986 was clearly biased towards industry. Net industrial production and transportation together accounted for 66 percent of total output in 1986; agriculture still occupies a remarkable 25 percent share. The share of foreign trade in 1986, when the country was already opening up, was two percent and the main trading partner was Yugoslavia, followed by the Eastern European countries of Romania, Poland, Bulgaria and Czechoslovakia, while Italy and Western Germany were its most important partners from the West (Zickel and Iwaskiw, 1992, p. 248).

'Albania's communist economic system, with its strict central controls, egalitarian incentive system, and bias toward heavy industry, collapsed in the early 1990s, idling almost all of the country's production lines' (Zickel and Iwaskiw, 1992, p. 106). Between 1985 and 1990 the government, pressured by the poor results of the existing model, allowed some decentralization and, therefore, in 1990 'a new economic mechanism' was adopted (Muco, 1997, p. 7).

12

Figure 3: Net Material Product by branch of origin, 1986, in millions of Lek and percent of total NMP



Source: Zickel and Iwaskiw, 1992, p. 244.

After the demise of the socialist ideology across Europe, Albania was pushed onto a path of transition. Similar to other countries in transition, the country launched a reform package that typically involved (Gomulka, 2000): macroeconomic stabilization, micro liberalization, structural changes, building new market institutions and safety nets. But the country additionally faced specific internal and external shocks (Muco, 1997, p. 6).

2.2 Transition

Due to political tensions first transition reforms in Albania were implemented only in the fall of 1991. Macroeconomic stabilization was necessary as its budget deficit soared (Figure 5) and inflation reached 226 percent in 1992 (EBRD, 2011a). In order to increase budget revenue, a new system of taxes and duties was implemented. In a monetarist manner, the country later in 1992 focused on inflation, which was, according to EBRD (2011a), reduced by 10 times from 226 to 22.6 percent in just two years. At the end of 1991 prices also started to liberalize. In 1992 the country also started to liberalize external trade and implemented a flexible exchange rate. International trade was dominated by the private sector (80 percent of imports and 50 percent of exports). The current account deficit continued to be a problematic issue: export declined due to poor quality, CMEA (the Council for Mutual Economic Assistance) collapsed, and domestic production generally declined while imports increased, which was the standard set of poor exports results across transition countries (e.g. Gomulka, 2000).

As far as restructuring goes, in 1992 a banking law was approved that started the restructuring of the banking system. Privatization also began in 1992 with small-scale privatization, and in 1995 privatization of large companies followed. Foreign capital was also invited to participate. According to Memaj and Koçi (2001), the results of privatization were mixed.

Figure 4: Real changes in GDP, industrial gross output and agricultural gross output in Albania, 1989-2010



Source: EBRD, 2011a.

Meanwhile, according to Muco (1997, p. 21) political tensions gave rise to a lack of clear vision and inefficiency, similar to many other transition countries, which

contributed to poor economic results (Figure 4). GDP declined by 28 percent in 1991, while industrial production shrank by 42 percent. The unemployment rate peaked in 1992 at 24.4 percent (Figure 7). Business environment also remained one of the important challenges (see Ministry of Economy, Trade and Energy, 2009).

An initial period of transformational recession was followed by a period of growth in the mid-1990s and the country received praise for following the Washington consensus (e.g. Williamson, 2004). But in 1997, Ponzi schemes dragged the country 'within weeks into anarchy, widespread violence, plundering, and food shortages' (Bezemer, 2001)⁴. According to the Bezemer (2001, p. 4), the depth of the crisis was a result of Ponzi games, other financial aberrations fostered by restrictive monetary policy, poor regulation, significant capital inflow and a weak government.

The crisis spurred additional reforms to strengthen the weak elements that the 1997 crisis revealed. Consequently, Albania made significant progress in the first decade of transition and growth has since continued (Table 3). The 1999 Transition Report (EBRD, 1999) sums the results of the first decade of transition in Albania by stressing the progress in legislation reforms (especially tax and secured transactions), investment climate, and the invitation of strategic partners and foreign capital in large-scale privatization, all of which enhanced the credibility of the program. Nonetheless, the EBRD (1999, p. 126) points out the problems of ineffective law enforcement, the black market, and weaknesses in the financial system, particularly the lack of credit available to enterprises.

⁴ There was great trust in the economy in money pyramids, even on the side of the state. This further increased the confidence of the people. Some even sold their appartments to join the pyramids with a greater stake. The collapse hugely impacted consumption in the economy, leading to a crisis.

¹⁵

Table 3: Reform process summary

	Liberalization, stabilization,	Enterprises, infrastructure, finance and				
	privatization	social reforms				
1991	Small-scale privatization begins					
1992	Full current account convertibility	Two-tiered banking system established				
	introduced Exchange rate unified					
	All quantitative controls on foreign trade					
	removed					
	Most prices liberalized					
1993	Restitution law for non-agricultural land	First foreign-owned bank opened				
	adopted	Enterprise restructuring agency established				
	Privatization of housing begins					
	Privatization agency established					
1994	Modernization of tax administration begins					
	Treasury bills market initiated					
400-	Most small-scale privatization completed					
1995	Voucher privatization begins	Competition law enacted				
100 (Land titles introduced	Bankruptcy law enacted				
1996	Central bank independence law adopted	Securities and exchange commission established				
	VAT introduced	Stock exchange established				
		First paramid scheme collapsed				
1007	0.11	First pyramid scheme collapsed				
1997	Urisis	Law on transparency adopted				
	VAI increased	Pyramid schemes placed under international				
1000	These many also as and with IME	administration				
1998	Comprehensive tex reform edented	State-owned Kural Commercial Bank closed				
1000	Main nin floor of reference former Kanana	Garital advances notice mixed to 12 generat				
1999	Major minux of refugees from Kosovo	Credit collings for private banks				
2000	Accession to WTO	Secured transitions law enacted				
2000	Accession to w 10	National Commercial Bank sold to foreign				
		investor				
		Mobile telephone operator sold to foreign				
		investor				
2001		Second mobile license awarded to foreign				
1		investor				
2002	Three-year plan adopted with the IMF	Deposit insurance law enacted				
	, <u>i</u> i	Bankruptcy law enacted				

Table 3: Reform process summary- continued

	5 1 5	
2003	Negotiations with the EU on a	
	Stabilization and Association Agreement	
	begin (SAA)	
2004	New competition law and establishment of	Increased credit to private sector
	independent Competition Authority	Infrastructural investment
	Legal reforms (Serious Crimes Court)	
2005	Implementation of law on restoration and	Infrastructural investment (roads, railway, port
	compensation of property	of Durres, electricity)
2006	Negotiations with the EU on a	Lower subsidies for electricity
	Stabilization and Association Agreement	New law on banking sector aimed at stronger
	finished	supervision
	New government begins fight with	Strong credit growth (backed with collateral)
	corruption and inefficiency with a series of	
	laws (Millennium Challenge Account)	
	Easing bureaucratic procedures for starting	
	companies	
	Social transfers scheme changed Law on	
	concessions and private procurement	
2007	Albania part of CEFTA	Electricity problems led to break up of KESH
	Sale of Albtelecom approved in parliament	Airport development
		Strengthened bank supervision
2008	New national business registration center	Electricity infrastructure investment
	(cutting time needed to register business)	Privatization - sale of ARMO (oil refinery)
	Legislation for protection further amended	
2009	SAA ratified by all member states of the	Further progress in power sector, participation
	EU	of private sector
	One stop shop for licenses and permits	Beginning of opening up the power market and
	opened to improve business environment	introducing competition
	Formal application for EU membership	Banking sector fully privatized
		Financial crisis did impact the financial sector,
		but less due to low international integration of
2010		the sector
2010		Privatization of some sectors delayed (insurance
		INSIG, power KESH)
		Deposit growth of 16 percent, also growth of
2011	Durante translation 1, 1, 1	Erecut, but only 8.0 percent
2011	Progress towards EU naited	Focus on improving transport intrastructure
	small state expend firms to privatize)	improved management of water and
	sinal state-owned limits to privatize,	Wastewater
	(insurance company) and Alberty-1	Figure - electrical plants concessions
	(insurance company) and Aldpetrol	Telegommunications
I		Telecommunications

Source: EBRD Transition Report, 1999, pp. 126-127 for years 1991-1999, for period from 2000 on the source were EBRD Transition Reports 2000-2011, Country report on Albania.

The second decade of transition was marked by the continuance of the reforms, especially in the field of integrating in the European cooperation processes with the Stabilization and Association Agreement with the EU (2009 ratified), CEFTA (2007) and applying for EU membership in 2009. The country has also been tackling several important challenges, especially in improving its business environment and the rule of law, dealing with infrastructural weaknesses, especially transport and electricity supply, and has also privatized some of the major companies (Albtelecom, KESH, partially also INSIG) (Table 3).

2.3 Albania today

Two decades of transition brought significant economic progress and systemic reforms during transition and have also provided a relatively solid foundation for growth (Figure 4). The country moved from a low-income to middle-income country with a per capita GDP of around 4,000 euros. Macroeconomic development was marked with strong growth, close to six percent on average in the past decade, while inflation, interest rates and exchange rates maintained stable (World Bank, 2010a). During the past decade economic growth in Albania was driven by the development of construction, business services and transport. The country also benefited from credit expansion, which has been very wide given the previously low base (EBRD, 2005, p. 95). Also, according to the World Bank (2010b), the economic progress in the past decade resulted primarily from structural shifts, labor moved from sectors with lower productivity to sectors with higher productivity: jobs were primarily created in manufacturing (primarily construction) and services, while the role of agriculture decreased.

The economic progress in the past decade was supported strongly by stable macroeconomic policies and structural reforms (Table 3).

Strong economic growth was stimulated also by strong investment performance. The total gross investment increased from 24.6 percent in 2000 to 29.5 percent of GDP in 2009, while private investment rose from 18 to 21.4 percent in the same period. But the World Bank (2010b) adds that the investment structure was skewed towards construction, primarily until 2006. In this period non-construction investment was only 10 percent. Since 2006 the situation has improved, although construction investment is still high, non-construction investment has been increasing (World Bank, 2010b, p. 4).

Figure 5: General government balance, 1990-2010, in percent of GDP





A stable macroeconomic environment, primarily resulting from prudent fiscal and monetary policy, also assisted economic growth. Between 2002 and 2007, improvements in revenue administration along with strong economic performance contributed significantly to deficit reductions and the decline of public debt from 62 percent in 2003 to 53 percent of GDP in 2007 (Figure 5). But in 2008 fiscal expansion, caused by an increase in public infrastructure investments and increase in public sector wages and pensions, increased public spending from 29.2 to 33.7 percent of GDP. The government failed to adjust spending, and by 2009 the deficit rose to seven percent. Also, as a result of the government's contraction of two syndicated loans, public debt increased to almost 60 percent of GDP (World Bank,

19

2010a, p. 5, World Bank, 2010b, p. 6). At the moment there is great pressure exerted on the government by the opposition and international organizations to not exceed 60 percent of GDP, which means halting infrastructural investment.

Monetary policy was also important in assisting transition by keeping inflation low. Inflation was high in the first half of the 1990s and rose again during the period of financial distress in the second half of the 1990s, but monetary policy later managed to stabilize it and keep it low (Figure 6) and within the central bank's objective of two to four percent. Albania initially adopted monetary aggregate targeting, but in practice the central bank besides monetary aggregate M3 also took into account exchange rates, price changes in different markets, and demand conditions. Overall, the policy was successful. Additionally, the choice of a flexible exchange rate system also proved to be wise, as it was a good shock absorber (World Bank, 2010b, p. 6).



Figure 6: Consumer price index, percent, annual average, 1989-2010

Source: EBRD, 2011a.

Macroeconomic stability contributed to the development of the financial sector, which supported economic expansion. Also, structural reforms and privatization (especially the privatization of the largest bank, the Savings Bank) led to a quick

expansion of the sector. The ratio of credit to GDP grew from 4.7 percent in 2001 to 37 percent in 2009 (World Bank, 2010b, p. 6).

In 2010 the Albanian economy continued to grow, although the pace of growth slightly deteriorated due to worsened external conditions. The estimated growth for 2010 was 3.8 percent, which is still the highest growth in the region for the second consecutive year. It is important to note that export was the strongest contributor to growth, especially the energy sector exports. Also, Albania managed to attract significant FDI inflow, around 800 million euros. The current account deficit, which was a problem throughout transition, was lower, estimated at about 12 percent of GDP in 2010 (EBRD, 2011b and 2011d). Even the financial crisis left for the moment only a minor mark on the economy. Economic growth remained positive throughout the period with a minimum of 1.9 percent forecasted for 2011, while industrial growth in the first quarter of 2011 was over 25 percent (Bole, 2011). The impact of the crisis will be discussed in detail later on.

Table 4: Macroeconomic data and projections for Albania, 2008-2011*

	1 5 5	-			
	2008	2009	2010	2011	2012
				projected	projected
GDP growth	7.5	3.6	3.8	1.9	3.5
Inflation (end year)	2.2	3.7	3.5	3.7	3.5
Government balance/GDP	-5.5	-7.0	-4.2	-3.7	na
Current account	-15.6	-15.2	-11.8	-10.9	na
balance/GDP					
Net FDI (in million US\$)	874	924	1098	914	na
External debt/GDP	32.6	41.1	36.6	na	na
Gross reserves/GDP	17.5	19.8	22.7	na	na
Credit to private sector/GDP	35.2	36.7	38.0	na	na

* For 2011, growth projection by the IMF is 2.5 percent for real growth and 3.9 percent for inflation. EBRD does not publish projections for 2012, consequently for 2012 IMF projections are given.

Source: EBRD, 2011b, EBRD, 2011c; EBRD, 2012; IMF, 2011a.

Of course there are numerous challenges ahead. Primarily, the process of transformation has yet to be finished, especially in the field of liberalization, structural reform, business environment and infrastructure. Also, economic growth, which today largely depends on the primary sector (agriculture, natural resources use), the construction industry, and cost competitive low value-added industries (e.g. textiles and leather industry) will have to be further restructured.

Figure 7: Real changes in the labor force and employment and the unemployment rate in Albania, in percent, 1989-2010



Source: EBRD, 2011a.

One of the important challenges is also the labor market. Despite successful growth, the unemployment rate remained high at around 13 percent (Figure 7) and did not follow the economic cycle. Employment was relatively stable; only in 2008 was there a slight improvement (World Bank, 2010a, p. 5). This indicates that growth was caused primarily by restructuring and consequent productivity growth. Of course employment growth also did not follow the pattern of growth. Job creation was poor and present primarily in industry, transport, telecommunication, trade, hotels and restaurants (IMF, 2008, p. 9).

22

Surprisingly, despite economic progress and population growth the labor force has been declining gradually since 1999 (with the exception of 2006 with 0.0 percent growth). Also, unemployment did not increase. An important cause of decline in the labor force and stable unemployment is emigration (World Bank, 2010a, p. 5), which is a great loss for future growth. Namely, according to Markova (n.a) in 2005 emigrants represented 26 percent of the working age population (15-64 years) and over one-third of the work force. Overall, according to the National strategy for development and integration (IMF, 2008, p. 9), around 20 percent of the population has been living abroad for more than 10 years (both legal and illegal migration). The government recognizes the size of the migration problem, with both it downsides and positive consequences. It primarily wants to fight brain drain by providing young, skilled personnel (also those educated abroad) good employment opportunities and improve the image and treatment of its emigrant workers abroad by providing suitable institutional support. Remittances are an important positive side effect of migration, and the government aims at directing them into productive investment (IMF, 2008, p. 32).

Closely related to the labor market and migration is also poverty. According to the World Bank (World Bank, 2010b, p. 6), the absolute poverty rate fell from 25.4 percent in 2002 to 12.4 percent in 2008, meaning that instead of over half a million people, now about 275,000 live in absolute poverty. There are significant differences between rural and urban poverty. Rural poverty declined by close to 50 percent, primarily because of remittances inflow in combination with employment shifts from agriculture to other sectors. According to the 'Albania: Poverty reduction strategy paper' (by IMF, 2008, p. 15), poverty is desired to fall below 10 percent by 2013 also due to ambitious developmental and social policy strategies. The most important policy tools are (1) changes in social protection programs (aimed at better coverage of the needs of the poor), (2) reforms in the pension

system, public health system (a set of services would be provided for free), (3) equal opportunities programs (designed to lower the female poverty rate by providing equal opportunities), and (4) reforms in the education system, which is expected to be more responsive to the needs of the economy and thereby increase employment prospects. Importantly, economic progress is expected to additionally reduce poverty, which is primarily due to public stimulation of agricultural and rural development, regional development programs, and spatial planning.

Several other developments are crucial for Albania's economic future, including the outcomes of privatization, foreign direct investment, new entry and business environment quality.

3 The business sector

3.1 Privatization

Privatization in Albania began in 1992 with land privatization, continued with the privatization of housing and small and medium-sized enterprises. Large companies were the last to follow. The process today is still not completely finished, but the majority of enterprises have been privatized, thus making it possible to judge the efficiency of privatization.

Similarly as other transition economies, the main goals of the privatization process were a reduction of state ownership inefficiency and an increase in output and welfare. At the same time, privatization was expected to be fast, socially just and transparent (Memaj and Dika, 2005, p. 1). Also, small–scale privatization was in progress, while in agriculture a significant amount of land (70-75 percent of arable land) was freely distributed in 1992 (Muco, 1997, pp. 19-20). Land privatization allowed the private sector in GDP to increase fast because of the relative importance of agriculture in GDP (24.7 percent in 1994, EBRD, 2011a). According to the privatization law of 1991, the virtual free distribution referred not only to agricultural land to members of state farms and cooperatives, but it also enabled rapid privatization of small businesses such as retail shops. As a result of these two processes, the private sector accounted for 50 percent of GDP (Sallaku, 2001) in 1994.

In 1993 housing privatization began and a year later it was basically finished. In November 1993, 97 percent of houses in urban areas were privatized, and 230,000 dwellings transferred to their residents (Hashi and Xhillari, 1999 in Sallaku, 2001).

In 1995 large enterprises (over 300 employees and a value over 500,000 US\$) also started the process of ownership transformation, with mass voucher privatization being the most important method of transformation. Vouchers could be used either in the bidding process, direct sales of assets and companies, or could be placed in investment funds (for details, see Memaj and Koci, 2001 and Malaj and Memaj, 2003). Mass voucher privatization¹ was used in Albania, similar to many other countries, due to its perceived social justice. In 1995 and 1996 in a total of five rounds of mass voucher privatization, 97 companies were privatized in the mechanical, chemical, construction and food industries (Sallaku, 2001).

In 1998 the government launched a new privatization strategy. The new strategy was aimed at increasing company efficiency and developing financial markets. It

⁵ Albania also used privatization leks, a term referred to compensation bonds to expolitically prosecuted individuals. They were treated as vouchers and could be used in privatization (Memaj and Koci, 2001).

²⁵

was based on first corporatizing remaining companies and then offering the shares in auctions to both domestic and foreign buyers. Therefore, the aim was also to attract foreign capital, especially in the case of larger companies (Albtelecom, Kesh, Albpetrol, Albanian Mobile Communications). The expected benefit was not only fresh capital, but also a restructuring of ownership (Sallaku, 2001).

The case of Kürüm International: Successful privatization by a foreign owner

Kürüm International is a steel production hub of Kürüm Holding and a part of the Kürüm Group, a privately owned company from Turkey, which operates in the field of construction steel production (companies in Albania and Turkey), construction (Turkey), production of industrial gases (Albania), shipyard (Albania) and the insurance business (Turkey, Kosovo). It was acquired in 1998 via Albania's privatization program. After privatization the company had to go through painful reorganization, as Chinese technology was obsolete and the whole facility was over-dimensioned. So far, Kürüm Group has invested 170 mil EUR into modernizing the Albanian facility. Today Kürüm International accounts for 40 percent of production of the entire Kürüm Group steel division.

The main products of Kürüm International are hot-rolled steel re-bar, billet and lime (fragmented-hydrated), which mostly serve as supply material to the construction business. The company is purely B2B oriented and very narrowly specialized to serve construction companies with no diversification of their products portfolio. Production in 2009 amounted to 308,141 tons of steel re-bar, 220,541 tons of billet, and 25,950 tons of lime. Although Kürüm International is exporting their products to customers in Serbia, Kosovo, and Montenegro, 60 percent of its sales are in Albania, and they cover 75 percent of all market demand in Albania. The company had 135 mil EUR sales in 2007 and is, despite the crisis, realizing a 6 percent year-to-year growth of sales.

Raw materials used in production are mostly (80 percent of them) imported from the Black Sea area (Turkey, Bulgaria). Albanian iron ore is not clean enough due to obsolete technology. Production know-how is imported from Turkey from where the majority of engineers and expert staff come from (60 middle and top management employees). The rest of the workers (800 employees) are Albanian. Although the company has too many workers given its current production, which is impacting productivity, it is in the mid-term, due to negative publicity caused by lay-offs, more rational to keep the workers.

Production of the company's products is very energy intensive. The company is one of the biggest consumers of electricity in the country (they use 360GWH per year), and is therefore very dependent on the public supply of electricity. To minimize the risk of electricity blackouts, Kürüm International is planning to invest in a 120mega watt hydro plant in 2012 and start buying electricity on the market. The company has ISO 9001:2008, ISO 14001:2004, and OHSAS 18001 certificates.

Competitors of Kürüm International are international producers that are exporting to markets of Kürüm International. There is a tough battle for the market since a large share of sales is directly connected with public tenders for construction in the Balkan region. The

success of Kürüm International is attributed to focused marketing – following the demand from construction companies. The company is focusing their exports to Kosovo where there is the largest expansion of public investments for reconstruction infrastructure and real estate. Otherwise, Kürüm International is facing problems with over-supply of real estate in the Balkan area (decline of demand), political instability, which could possibly influence sales – "wrong party follower," and is not getting licenses/permits to operate. In order to be successful, Kürüm International has to be able to quickly adapt to changes in the business environment. This is mainly realized through the Kürüm Group's vision to further diversify their vertically integrated portfolio of operations where they also see two main opportunities for future growth in Albania – the energy sector and mining.

The overall opinion is that the privatization process in Albania in many aspects failed to bring the desired results: efficiency and restructuring. Estrin et al. (1998) estimated that privatization in Albania began in worse initial conditions and with delay. The process is still not completely finished (EBRD, 2010). Nonetheless, studies of privatization efficiency have been prepared. Memaj and Dika (2005) estimate that the voucher privatization process was relatively unsuccessful, and that the mix of methods and the choice of companies were in many cases inappropriate. Memaj and Koci (2001) find that voucher privatization initially caused problems due to the inefficiency of highly dispersed ownership. Thus, a gradual concentration was necessary. An additional problem was also the lack of desire of new shareholders to enter, which caused problems of restructuring and investment. Memaj and Koci (2001) also stress both the lack of and high cost of external finance. Consequently, the majority of companies desire foreign funds to enter. Memaj and Dika (2005) add that the purpose of privatization to increase efficiency was not achieved as expected, and that the causes of inefficiency can be attributed to 'negative effects of government policies to create a favorable environment for progress, difficult situation for financing the technological restructuring and lack of assistance for managerial restructuring of these companies, unfavorable environment for ownership concentration and for foreign investors' participation in mass voucher privatization and during the post privatization period, inefficient role of supporting actors in the process and playing "without rules of the game" for

different actors of the demand side etc.' Importantly, the state was not expected to provide direct financial assistance and support, nor to provide required training needed to successfully run the companies (knowledge of business legal rules, marketing, finance, etc.) (Memaj and Koci, 2001). The authors were very critical: 'The state has tried only to "get rid" of them.' (Memaj and Koci, 2001, p. 11).

Memaj and Dika (2005) and Memaj and Koci (2001), having studied the inefficiencies linked with the privatization, prepared a series of recommendations to improve company performance. Many of them are linked to the general business environment. Namely, the authors recommend: (1) the creation of a competitive economic environment, (2) the creation of demand through public procurement, (3) the development of specialized consulting agencies to help the economy (e.g. for foreign investors), (4) the stimulation of growth of the Tirana Stock Exchange in order to help the process of ownership concentration and accessibility of external finance, (5) the development of human resources programs by the government to help companies manage their labor needs, (6) the creation of a suitable regulatory environment.

Economic development in transition economies, however, depends not only on existing companies and their privatization, but also on newly established companies.

3.2 New entry

Socialist economic environments were dominated mostly by large conglomerate enterprises. In some countries the small private sector was allowed, but in a very limited fashion, while medium-sized firms were virtually non-existent, which is referred to as the 'socialist black hole' (Petrin, 1989). According to Jeffries (1990, p.

80) there was no legal private sector in industry in Albania. While a limited amount of private property was allowed in agriculture (200 square meters), private output had to be sold to co-operatives at state determined prices (Jeffries, 1990, p. 82).

The role of the SME sector is very important in Albania. According to Gruda and Milo (2010, p. 8), 98 percent of all active enterprises are micro in scale (not including the agricultural sector); employing less than five (average employment is 1.5), while about one percent of enterprises are medium-sized companies with an average employment of 42. At the end of 2008 there were 107 thousand companies. The SMEs contribute to 82 percent of turnover, 52 percent of GDP and 57 percent of employment (Gruda and Milo, 2010). The small and medium-sized companies predominantly engage in trade activity (57 percent), followed by agriculture (18 percent) and construction (15 percent) (Gruda and Milo, 2010). Slightly older data (OECD, 2003) also show high concentration of SMEs in Tirana (40 percent or more of all SMEs).

The development of the small and medium sector depends on a number of factors: macroeconomic and microeconomic environment, entrepreneurial and business skills, social and economic conditions for entrepreneurship, financial assistance to small and medium-sized firms, non-financial assistance to small and medium-sized firms and market access (Gruda and Milo, 2010, p. 6). If SMEs in Albania are motivated primarily by obtaining the (minimum) means of survival, their existing nature could hide several potential threats. Prašnikar et al. (2010) analyze the entrepreneurial sector in Kosovo. The authors examine why the SMEs in Kosovo failed to grow. The answer is that there is s significant difference between selfemployment and actual risk-taking entrepreneurship. Self-employees, due to their limited resources, often engage in very similar simple activities (trade, kiosk, etc.), earning barely enough to survive. True entrepreneurs will more often recognize the difficulty of obtaining funds and will more actively recognize problems in the business environment. Both groups are a potential pool of successful SMEs. It is extremely important, though, for the government to open horizons by providing more opportunities for employment in general, especially in the more propulsive sectors. This goal can be achieved also by obtaining more foreign direct investment.

The importance of the environment on the development of SMEs is widely recognized (Ministry of Economy, Trade and Energy, 2011, OECD, 2003, Bahiti and Shahini, 2010), primarily suitable SME policy, legal and regulatory environment, tax policy and supportive environment (suitable financial instruments, advisory services and business incubators), education and training for entrepreneurship, availability of cheap and fast start-ups, improving on-line access, strengthening the technological capacity of small enterprises, and developing stronger, more effective representation of small businesses' interests. The government made several legislative improvements, and established the Agency for SME Development, and is also focusing on SMEs with regards to technological developments (e.g. Ministry of Economy, Trade and Energy, 2011) and the general business climate.

According to the Ministry of Economy, Trade and Energy (2009), the progress made by the country resulted in many positive growth effects, especially increased investment efficiency, the advantages of scale economy, increased investments, the adaptation to new technologies, and the creation of new firms and industries. The document also acknowledges the importance of stable economic growth of the SME sector, primarily an increase in the number of new enterprises, improvement in productivity and competitiveness, and expansion of firms with international potential. It is also stressed that in the view of future EU integration, the competitiveness of the SME sector, which depends on innovativeness, will be important.

The case of GPG Company: An innovative company in cost competition

Golden Pen Generation – GPG was founded in 2007 by an Albanian entrepreneur that took over the knowledge and technology for producing plastic pens of the formerly bankrupt company. GPG is a purely production-oriented company that is exporting 100 percent of its production through a Slovenian partner company that operates as an outsourced sales department. The Slovenian sales partner is also the original designer of GPG's main product. GPG is part of a larger group called Nuova Plastica, which started as a business expansion of GPG producing antitheft security plastic products for clothes and plastic elements for the car industry.

GPG is an exporting company that generates revenues of around 1 mil EUR. GPG buys all of its major plastic supplies (PCPMAA) from Slovenia (Šenki, d.o.o. – www.senki.si). Its main competitive advantage is low price due to low labor costs and huge flexibility in production since 52 out of 60 employees are so-called "rented employees," that is they are employed only when an order is placed and are doing the assembly of pens which is the only part of production which is not automated. All of the GPG engineering staff is trained in Slovenia.

Such a business model, although successful for now, poses a few challenges due to Albania's unique business environment and due to common challenges of production companies around the world. One of them is taxation problems of exporters in Albania due to bureaucratic and corruption issues. Albania has a VAT system, meaning input VAT is paid on all goods purchased, and since exporters do not charge output VAT, this means that exporters claim input tax from the government (are net receivers of VAT). However, net receivers report problems from collecting VAT from the government, which in many cases results in write-offs of input VAT and thus creates substantial influence on financial outcome.

The biggest production cost of GPG is the plastic granule for molding the plastic parts, which represents around 30 percent of the GPG pens price. Another 35 percent of the price is other production costs including labor. Since labor costs are on close to its minimum, the main focus of GPG is to reduce the cost of the granule (main material input) and to further optimize the production process to increase quality of its products, which is already on a very high level with faulty products of less than 0.003 percent. Therefore GPG has established an R&D department, which consists of two of its 8 engineers employed. Their R&D department is cooperating with Italian academic experts on plastic. GPG's main methods are trial and error and learning by doing. In this way they have been able to gradually increase the quality of their pens and introduce new input materials. With successful research they have found out that recycled car headlights are equal substitutes to plastic granule. GPG is thus able to reduce the cost of the main input material and at the same time include an environmentally friendly component into their production by using recycled materials. This will result in lower dependence on their main supplier and lower dependence on input cost fluctuations as plastic granule is a direct derivative from oil. Although the use of recycled materials has big potential, there are bureaucratic limitations due to regulation - mainly ban on "garbage" imports in Albania, and at the same time Albania does not collect enough waste materials separately so it could be efficiently used by GPG. However, they are discussing options on importing recycled headlights from an Italian supplier.

Although GPG is a small company, it is export oriented, focused on technology improvements, and aware of changing business models concerning the environment by using recycled materials.

3.3 Business environment

Albania's business environment remains one of its key challenges (EBRD, 2011b). Currently, the country ranks 82nd among 183 economies on the 2011 Doing Business scale, which is low in comparison to the rest of the transition economies if judged by Doing Business. In 2011 according to Doing Business (2011a), Albania was behind the CEE economies and also several former Soviet economies (Table 5), which are often criticized because of their business environments.

Economy	Rank	Economy	Rank	Economy	Rank
Singapore	1	Macedonia	38	Poland	70
Hong Kong	2	Slovak R.	41	Mongolia	73
New Zealand	3	Slovenia	42	Albania	82
United Kingdom	4	Kyrgyz R.	44	Croatia	84
United States	5	Hungary	46	Serbia	89
Denmark	6	Armenia	48	Moldova	90
Canada	7	Bulgaria	51	Bosnia&Herceg.	110
Norway	8	Azerbaijan	54	Kosovo	119
Ireland	9	Romania	56	Russian F.	123
Georgia	12	Kazakhstan	59	Tajikistan	139
Estonia	17	Czech R.	63	Ukraine	145
Lithuania	23	Montenegro	66	Uzbekistan	150
Latvia	24	Belarus	68		

Table 5: Ease of doing business 2011: Top 10 and transition countries

Source: Doing Business, 2011a.

The most problematic procedures of doing business in Albania are: dealing with construction permits, paying taxes and closing a business (Table 6). Property registration, contract enforcement and trade are also quite problematic areas.

32
Topic Rankings	DB 2011 Rank	DB 2010 Rank	Change in Rank
Starting a Business	45	44	-1
Dealing with Construction Permits	170	172	+2
Registering Property	72	69	-3
Obtaining Credit	15	14	-1
Protecting Investors	15	15	No change
Paying Taxes	149	152	+3
Trading Across Borders	75	70	-5
Enforcing Contracts	89	91	+2
Closing a Business	183	183	No change

Table 6: Doing businesses in Albania

Source: Doing Business, 2011b.

Corruption is often mentioned as a problem in Albania. According to the 2010 Corruption Perception Index (CPI), Albania ranked 87th among 178 economies with a score of 3.3 on a scale from 10 (highly clean) to 1 (highly corrupt). Denmark, New Zealand and Singapore are ranked at the top with an index 9.3, followed by Finland and Sweden at 9.2. At the end are Somalia (1.1), Myanmar and Afghanistan (1.4) and Iraq (1.5) (Transparency International, 2011).

Despite low rankings the country has improved its business environment since the beginning of transition. Corruption has also been a focus of the government, especially in the past few years. According to EBRDs Transition Reports (see especially issues 2005-2010), Albania has made significant progress in the business environment, primarily fighting corruption and organized crime, reviewing administrative procedures and improving the rule of law in order to increase efficiency and transparency and strengthen the judiciary system in order to provide a better framework for investors, especially foreign investors.

3.4 Trade and FDI

In the past Albania followed the import substitution model of development and tended to reach self-reliance. With transition, Albania started to open up and export promotion became a central element of economic growth and development (World Bank 2010a, World Bank, 2010b). In 2000 Albania became a member of the WTO (WTO, 2012).

	2005	2006	2007	2008	2009
Manufactures exports (million US\$)	525	585	754	948	763
Structure (in perc	ent)				
Iron and steel	7.4	3.9	4.2	12.1	5.6
Chemicals	0.6	0.5	0.5	0.6	0.9
Pharmaceuticals	0.2	0.0	0.1	0.1	0.1
Machinery and transport equipment	5.1	5.3	5.8	5.7	6.8
Office and telecom equipment	0.8	1.2	1.2	0.8	1.3
Electronic data processing and office equipment	0.4	0.5	0.7	0.5	0.5
Telecommunications equipment	0.4	0.7	0.5	0.3	0.7
Integrated circuits and electronic components	0.0	0.0	0.0	0.0	0.0
Automotive products	0.4	0.2	0.3	0.2	0.4
Textiles	0.4	0.5	0.7	0.8	0.9
Clothing	37.7	38.5	38.3	37.0	38.1

Table 7: Total merchandise exports in millions of US\$ and structure of merchandise exports, 2005-2009

Source: WTO Database, 2011.

The country is a very small exporter in global terms, representing only 0.01 percent of total world exports. In 2009 the majority of exports (70.1 percent) was manufactured goods, 21.1 percent was fuels and mining products, and the rest was agricultural exports. The country's reorientation towards exports is quite evident. Namely, from 2000-2009 the average yearly growth of merchandise exports was 17 percent. Main exporting products are clothing, iron and steel industry products and machinery and transport equipment (Table 7). The biggest merchandise export destinations are the EU (84.7 percent), China (4.8 percent), and Macedonia (2.8 percent) (data 2009). In terms of services, the main service to be exported is travel (77.8 percent), followed by transport (10.7 percent) (data for 2009). The average growth of service exports was 21 percent between 2000 and 2009.

FDI is also becoming increasingly significant for future development in Albania. The data shows that the yearly inflow of FDI to Albania reached almost 8 percent of GDP (2009) and 9 percent in 2010 (Table 8). But the inflow is lower than desired (Ministry of Economy, Trade and Energy, 2009), primarily because of the poor institutional environment, rule of law and corruption (EBRD, 2007-2010).

	1		5		0/ of Cross	0/ of total	
					70 OI GIOSS	70 OI tOtal	
					Fixed	trade	
	Total FDI	Per capita	% of total		Capital	(goods &	% of trade
	*	FDI *	world FDI	% of GDP	Formation	services)	of goods
2000	144.3	47.0	0	4	12.5	20.3	55.2
2001	206.4	67.1	0	5.1	13.2	24.6	67.7
2002	135.0	43.7	0	3	8	14.7	40.9
2003	178.0	57.3	0	3.1	7.7	15.2	39.7
2004	345.7	110.6	0	4.7	12.7	21.5	57.1
2005	264.3	84.1	0	3.2	8.8	14.5	40.2
2006	325.3	103.0	0	3.6	9.3	14.1	40.8
2007	656.0	207.0	0	6.1	15.8	21.7	60.9
2008	988.3	310.6	0.1	7.6	20.7	25.8	72.9
2009	979.4	306.8	0.1	8.1	24.5	28	89.8
2010	1096.9	342.3	0.1	9.2		26.3	70.8

Table 8: Trends in FDI in Albania: FDI inflow

* US dollars at current prices and current exchange rates in millions Source: UNCTAD Database, 2012.

3.5 Sectoral developments

The Albanian economy has changed extensively since the beginning of transition. While during socialism the focus was on industrial development, transitional restructuring shifted the economic focus away from manufacturing. By 2008 the share of industry fell to only 10.4 percent. Agriculture was traditionally strong in Albania, but its share also declined from around 35 percent in 1989 to 21.5 percent in 2008. The share of services increased dramatically to over two-thirds of GDP (Figure 8).



Figure 8: Structure of Albanian GDP, in percent, 1989-2008

Source: EBRD, 2011a.

Given that the transformation of the Albanian economy is not yet over, the composition of GDP will continue to change. But a detailed look at the sectoral structure reveals that the key sectors in Albania are agriculture, the textile and leather industry, wood and related industries, electricity, gas and water supply, construction, trade, real estate and business services, transport and financial sector (Table 9).

Sectors		Growth
		rate
Agriculture, hunting and forestry		
Agriculture, hunting and forestry	17.4	1.1
Fishing	0.2	0.0
Industry		
Mining and quarrying of energy producing materials	0.8	0.3
Mining and quarrying of except energy producing materials	0.4	0.0
Manufacture of products based on cereals	0.6	0.2
Other manufacture of food products	0.8	0.1
Manufacture of textile and leather products	1.5	0.1
Manufacture of wood, paper, furniture; publishing and printing	1.1	0.1
Manufacture of coke, refined petroleum products and nuclear fuel	0.1	-0.1
Manufacture of chemicals, chemical products, rubber and plastic		
products	0.3	0.0
Manufacture of other non-metallic mineral products	1.0	0.0
Manufacture of basic metals and fabricated metal products	1.0	0.2
Manufacture of machinery and equipment	0.2	0.0
Electricity, gas supply, water	3.0	1.3
Construction	9.1	-2.7
Services		
Trade	16.3	1.4
Hotel and restaurants	3.0	0.4
Transport	5.2	0.5
Post and communication	3.0	-0.3
Financial activities	4.1	-0.3
Real estate, renting and business activities	8.5	1.0
Public administration and defense, compulsory social security	4.0	0.1
Education	3.3	0.0
Health	1.8	0.0
Other community, social and personal service activities	3.1	0.4
Total VADD	89.8	3.9
FISIM	3.6	-0.2
GDP at basic prices	86.2	4.2
Taxes	14.0	0.1
Subsidies	0.1	0.0
GDP at market prices	100.0	43

Table 9: The structure of Albanian GDP in percent and growth rate of sectors in 2010

VADD: Gross value added, FISIM: value of financial intermediation services indirectly measured

Source: Instat, 2011.

3.5.1 Agriculture

Prior to transition, Albanian traditional mainstay agriculture contributed about a third to the country's net material product and employed more than one-half of the population. Farm products represented a major share of exports; 25 percent in 1990 (Zickel and Iwaskiw, 1992, p. 133). Data (Figure 9) reveals that cereals, primarily wheat, sugar beets and milk represented the most important products in Albania. The data is for 1988, but the structure did not significantly change during 1979-1988 (see Zickel and Iwaskiw, 1992).

Figure 9: Primary agricultural output in Albania in 1988, in thousands of tonnes and percent of total



Source: Zickel and Iwaskiw, 1992.

Today, agriculture and the food industry combined still represent up to 25 percent of GDP. The sectors employ about 60 percent of all labor. Despite the size of these sectors, production does not meet demand in many areas. Agriculture in Albania is on average inefficient and uncompetitive, local terrain is mountainous. As the case study of EWH, Gmbh shows, there is room to develop. The government is also working actively on promoting the wine industry, and olive and fruit production (Ministry of Economy, Trade and Energy, 2009).

The case of EWH, Gmbh: Building a leading brand in the food industry based on quality and worker relations

EHW Company is a family operated, meat processing company and retailer of meat products, established in 1992. The company ownership is split equally among an Albanian partner, who is also the executive manager, and a passive Italian owner (equity investor). EWH has a roughly 45 percent market share and realizes around 20 mil EUR revenue. EWH exports a small share of their products to the Egyptian Christian community, otherwise they operate purely in Albanian market.

EWH Company imports all of its meat inputs, primarily from Brazil (Sadia being the biggest supplier). EWH decided for input import due to cost advantages and quality (meat is controlled). Even though around 60 percent of the Albanian population is Muslim, most have no problem eating pork meat, thus around 90 percent of the imported meat is pork. Albanian meat is only bought to be sold as fresh meat locally, but in small quantities and mostly as beef.

Production is centralized in their facility near Tirana. EWH produces 70 types of different products, primarily sausages (40 percent of production). The company is dedicated to using higher quality meat with a lower fat content (10 percent most) in producing products based on its own recipes, which are constantly being improved in line with current trends and available information.

The production lines were imported from Germany and Italy and were in part adjusted to the specific needs of the company. Staff training and maintenance services contribute to higher quality production. Packaging material is imported from Italy. The company also has internal quality control labs for: 1) chemical and physical analysis, 2) microbiological analysis. Internal control and standard for meat quality exceeds those of state regulation.

Distribution is organized through 14 of EWH's retail stores, representing 35 percent of total sales. The rest is wholesale to EWH exclusive stores (some form of franchise), which sell only EWH meat products, have EWH company logos in their stores, but are not regulated on other goods sold in the store. In company-owned stores EWH also sells local fresh meat.

The company employs 200 people from the vicinity, 80 of which are in production. Although there is no collective agreement that offers great flexibility to the employer, the company has a very low fluctuation rate, also due to higher wages than average in the industry. Workers have left the company voluntarily, mostly due to emigration and not better job offers from domestic competition. Workers have to be notified two months in advance if they are to be laid-off, while they do not have to notify the company if they want to leave. EWH also organizes a bus for taking workers to work and home. Labor costs represent roughly 6 percent of total costs.

EWH, with their high quality standards and well-organized production and distribution process, has been able to become a market leader and a price setter. Their price positioning is slightly higher than their competition due to better quality of their products, and high consumer awareness of their brand. There are approximately 50 smaller local meat producers/sellers and 2-3 bigger ones that compete with EWH in the processed meat market. However, they will be faced with changes in the industry if Albania enters the EU. They will have to stop importing from their current source because of EU regulation.

Xherdo Ltd. is also a fine example of a firm that represents the direction for possible development.

The case of Xherdo Ltd.: Investment and innovation facilitating transition from family business to foreign markets

Xherdo Ltd. is a medicinal and aromatic plants and essential oil processor and distributer. The company started as a family business in 1991 and incorporated in 1995 as Xherdo Ltd. Although the first distillation units were primitive, with time the technology was upgraded which led to an increase in production volume and export value as well. The company started exporting in 1995. Their most important markets are currently Austria, Germany, Italy, and France.

Although Xherdo started as a family business it incrementally developed into an export-oriented company. It realized the need to invest in order to strengthen its position in the market, as it is currently a relatively small player in its export markets with around 2 million EUR of revenues in 2010. The company made its first significant investment in 2007 serving the company's future goals and marking a new investment cycle. A new facility of 11,500 m2 was established in Maminas and designated for the processing of dried herbs and spices and the production of essential oils. In 2008 it invested in stainless steel distillation machineries. In 2009 completely new processing lines of dried herbs were designed in Germany and Austria, which were then installed at the processing factory by German specialists. In total, the company invested about 1.3 million EUR. Expected yearly revenues with all equipment fully functional are around 4-5 million EUR.

While 45 percent of revenues come from higher value added essential oils (9 kinds of oil) production, Xherdo still generates 55 percent of revenues from medicinal plants – herbs which are just cleaned, ground, processed, and packed in big bags. Xherdo also produces some products under their own brand (fruits – blueberry and oregano). Xherdo currently employs 12 people permanently and 45 seasonal workers for sorting and harvesting (blueberries and other plants in the region). Xherdo has a supply network operating all over the country. There is still space for improving efficiency through automation in this part of the process.

Xherdo won several award for their operations (International Star Award for Quality in Geneva in 2009 and International Quality Summit Award in the Platinum Category in New York in 2010), and became a member of international organizations like IFEAT (International Federation of Essences and Aroma Chemicals Trade), IFOAM (International Federation of Organic Agriculture Movement), and AMAPSEEC (Association of Medicinal and Aromatic Plants of South Eastern European Countries).

Xherdo's current position in the market offers a lot of challenges and threats. As their current operations are well established and generate decent returns, they are challenged externally by their direct competitors in Albania (eight competitor companies of which three are very similar in nature of business as Xherdo), and internally by their own not optimally efficient processes. Xherdo is currently not producing at optimal capacity and this is the focus of its short-term strategic plan.

On the other hand, Xherdo's long-term goals are ambitious. The company is thinking of using aromatic hydrolytes in the production of organic shampoos, creams, etc., and using waste of extracts and dried herbs as compost/organic matter for park plants. Such thinking may be very good in terms of diversifying portfolio of products, but may also slow down the plan of achieving economies of scale in Xherdo's current operations, which is still not optimal.

3.5.2 Manufacturing

The manufacturing sector increased significantly in the Albanian economy during socialism. The data for 1988 (Zickel and Iwaskiw, 1992) reveals that the focus was even then on the food industry, which contributed almost 25 percent to total manufacturing output, followed by the light industry (16.2 percent) and engineering industry (14.5 percent). The production of ore was very important with copper itself contributing 8.8 percent of total manufacturing production and other ore-related industries or ore production another 8 percent (iron and metallurgy, glass and ceramics, coal, chromite) combined.

Today, as Table 9 revealed, Albanian manufacturing in the non-food sector is concentrated primarily in textiles, wood, the chemical industry, the leather and shoe industry, metallurgy, and machinery. Production facilities are concentrated primarily in Tirana, Fieri, Durresi and Elbasani.

The shoe and leather industry is one of the most prosperous branches in Albania at the moment. Approximately 100 companies produce well over one million pairs of shoes every month. The majority of these shoes are exported. Albania is the second largest exporter of shoes to Italy, which in turn is the largest exporter of shoes worldwide. Over the last few years, Albanian exports in the shoe and leather sector have increased by 20-30 percent per year. Albania is very attractive due to its cost competitiveness; the labor costs are only one-tenth of the costs in Italy and onefifth of the costs in Greece. Also, Albania offers quick transport to Italy and Greece, which both serve as a base for further distribution. The abundance of unskilled and skilled labor (leather and textile engineers) makes the sector more attractive to foreign companies (Ministry of Economy, Trade and Energy, 2009).

The textiles industry is, at the moment, the next highly dynamic sector. Its development is based on the same competitive (cost) advantages. Production in this industry is primarily exported and is based on commission contracts (Ministry of Economy, Trade and Energy, 2009). The case of Stella Company reveals the disadvantages of such an approach.

The case of Stella Company: Investment in intangible capital needed to increase value added in the textiles industry

Stella is a textile company producing underwear under their own brand. It was founded in 1993 and is owned from 1996 onward by its CEO, an Albanian businessman. Stella is mainly dependent on its Italian partners as is the case in many Albanian SMEs. All of Stella's 3 million revenues are made in the Italian market, around 20 percent of which is their own sales and around 80 percent is through eight Italian partners. Such partnership arrangements allowed Stella steady, but moderate growth. Its Italian partners also supplied Stella with knowledge about new technologies and adapting cuts and styles to the needs of the international market. Today Stella produces around 120 different styles and cuts for all ages.

Materials are mainly imported from Turkey and Pakistan. The factory has two production plants and employs 300 workers, paid better than the average in the industry (around 200 EUR net and 250 EUR gross wage). The majority of workers are employed in production (sewing). Although they have managed to shift from the mass production of a few products to smaller batches of many diversified products with learning by doing, and they have received knowledge from Italian partners, Stella still lacks the knowledge and equipment to shift production from mainly "refinement operations" to higher value added complete production operations. Stella also does not achieve quality and control standards of high branded European producers and, therefore, cannot pitch the outsourcing refinement business of these companies.

In order to expand operations and sales, Stella will have to make substantial investments in equipment and education to further diversify into the higher value added lingerie market. Currently their brand is positioned in the lower class of their segment. Although Stella has some domestic competition with similar business models, the competition has problems with regulation for exporting to Italy. Stella is mainly threatened by low cost producers from China, Pakistan, and Turkey. In order to further develop and enter new markets with different business models (sales on their own), Stella urgently needs to invest into knowledge and better equipment.

The wood industry is also very important in Albania. According to Policy options for wood energy (2009), forests cover 36 percent of Albania. This sector is one of the fastest growing in Albania and represents a growing share of exports; wood and

paper products represented 3.3 percent of exports in 2009 (Instat, 2010). Given that the exports in this industry are primarily material and intermediate products and that the furniture industry is growing with interest from foreign investors, there is potential for the sector to move into higher value-added production. Crucial investments in tangible and intangible capital are needed in order to make such companies more competitive (the case of Dafinor, Ltd.).

The case of Dafinor Ltd.: Low technological solutions and export of cheap raw materials

Dafinor is a wood panel production company. The company started wood production in 1990 as a government-owned company and was privatized by the voucher system in 1993. The company is currently owned by two Albanian brothers and an Italian partner.

Dafinor's production processes of tradable goods in wood industry products and production processes are defined by low technological solutions and the export of semi products or cheap exports of raw materials. Dafinor's technology, quality control, and supervision come from Italy, which is also their main export market (90 percent of their export). Although Dafinor is trying to expand its operations into a higher value added market of finished products, 80 percent of their production is represented by wooden panels and 20 percent by final products, which range from interior design to the manufacturing of a wide range of furniture from massive wood and other wooden products. Therefore only a minority of their products is distributed via B2C market. The biggest advantage of Dafinor is cheap production as raw materials are supplied locally (wood) and the labor force is cheap in respect to the EU average in the industry (although Dafinor pays its workers 20-30 percent higher than the Albanian average). Dafinor, interestingly, employs female workers in the production process, whereas male workers have higher-level positions.

Mining is one of the strategic sectors in Albania. Natural resources such as chromium, copper, nickel and coal are abundant throughout the country. In terms of chrome, the country was, prior to transition, the world's third largest exporter. Today the majority of production is exported only partially processed despite the fact that finalization reaps more economic benefit. Albania also has oil reserves that are estimated to be 30 million tonnes of extractable reserves. The sector is being

actively developed, with the help of the government, and foreign capital is welcomed.

In general, the Albanian government is investing a great deal of effort to help develop the industrial sector. The policies in the industrial sector are closely intertwined with the rest of the development policies, especially those dealing with SME development, export promotion, and FDI (Ministry of Economy, Trade and Energy, 2009).

3.5.3 Services

Services were traditionally neglected in socialism due to a focus on material production, especially manufacturing. Also, private property was virtually absent and retail trade was a state-owned operation with fixed prices and rationing. Thus, the black market was very lively. With the start of transition, controls on domestic trade were lifted, allowing private initiative. Thus, the service sector boomed, especially retail shops and restaurants (Zickel and Iwaskiw, 1992, p. 56).

Today, services contribute slightly over 50 percent of total GDP (data for 2010, see Table 9). The most important is trade representing 16 percent of GDP, followed by real estate with 8.5 percent, transport with slightly more than five percent, and financial services with 4.1 percent of GDP.

4 Albania: financial crisis and future challenges

The global economy has been earmarked by the 2007 burst in the financial sector, leading to a general economic distress with a high toll on many developing economies, including transition economies (e.g. Baltic countries, Slovenia, see Eurostat, 2011). Albania has so far been relatively successful, comparatively, in weathering the crisis.

	Albania	EU27	Slovenia	Montenegro	Croatia	Macedonia
2001	7.2	2.0	2.9	1.1	3.7	-4.5
2002	3.4	1.2	3.8	1.9	4.9	0.9
2003	6.0	1.3	2.9	2.4	5.4	2.8
2004	5.9	2.5	4.4	4.4	4.1	4.6
2005	5.7	2.0	4.0	14.7	4.3	4.4
2006	5.5	3.4	5.8	8.6	4.9	5.0
2007	5.9	3.1	6.9	10.6	5.1	6.1
2008	7.7	0.5	3.6	6.9	2.2	5.0
2009	3.3	-4.3	-8.0	-5.7	-6.0	-0.9
2010	3.0	1.9	1.4	2.5	-1.2	1.8
2011	1.9	1.6	1.1	2.7	0.6	3.0
2012	3.5	0.6	1.0	2.2	0.8	2.5

Table 10: Economic growth in selected economies, 2001-2012*

*Data for 2012 are forecasts.

Source: EBRD, 2011c, Eurostat, 2011.

One of the main reasons behind Albania's success is its comparative closedness and underdeveloped financial system (EBRD, 2011c). In 2007, Albania's growth was strong at 5.9 percent and was, according to EBRD (2011d), also becoming broader based, which is a condition for its sustainability. In 2008 growth was even stronger at over 7 percent, while in 2009 and 2010 it did decelerate, but dropped to a still strong 3 percent. This is the result of a combination of reduced industrial output, a drop in capital inflows, falling remittances, and slower credit growth on the negative side and increased government spending aimed at reducing this impact. The main contributions to growth come from new mining projects, construction and strong performance in tourism and exports (EBRD, 2011d).

Comparatively, Albania is doing quite well (Table 10). Despite the crisis, economic growth remained positive, unlike in the EU27 and other selected Balkan economies, and was in 2010 the highest in the region for the second consecutive year. Bole (2011) also shows that Albania also experienced relatively stable growth. Only in the fourth quarter of 2009 did growth fall slightly below zero percent. In the countries of South-Eastern Europe growth rates were, on average, negative from the last quarter of 2008 through the second quarter of 2010, which is a significantly longer period. Similarly, the successful trend in Albania can also be observed in industrial production (Bole, 2011).

The fiscal situation should be closely monitored due to the deterioration in 2008 and consequent rise in deficit and debt. The World Bank and the IMF (in World Bank, 2010a) forecast a negative overall deficit of 4.5 percent of GDP for 2011, remaining over 3.5 percent in the red also through 2012-2013. The primary budget is also expected to be negative at around one percent of GDP or less (1.1 in 2011, 0.7 and 0.8 in 2012-2013). A comparative analysis of the fiscal situation confirms that Albania has a high deficit and debt (Table 11), which is a problem recognized also by international organizations. The country's relatively large public debt and lack of exports, competitiveness, and consequent current account deficit pose a threat to future development according to EBRD (2011c). Similarly, the World Bank (2010a) claims the country must focus on retaining macroeconomic stability primarily through fiscal policy rule.

	Budget deficit	Public debt
Albania	-3.73	59.36
Bosnia and Herzegovina	-3.47	39.59
Bulgaria	-2.51	17.78
Croatia	-5.74	47.53
Macedonia	-2.51	26.30
Montenegro	-3.38	43.07
Romania	-4.41	34.36
Serbia	-3.77	44.07
Slovenia	-5.60	43.64

Table 11: Budget deficit, public debt and foreign financial investment as a percent of GDP, estimates for 2011

Source: Bole, 2011.

The banking sector for now remains relatively stable. According to the World Bank (2010a), the Albanian banking sector was one of the few in the region that had a large positive net foreign asset position, primarily because of prudent management and limited credit lines of foreign banks, which meant low external private debt. The central bank's efficient supervision spurred by the financial distress caused by pyramid schemes also contributed to the country's relatively stable position. Namely, although non-performing loans increased, the situation is, according to the World Bank (2010a), not critical and claims that the financial system is able to withstand even more severe stress testing. However, a comparative analysis of the region (Table 12) reveals a few caveats, those primarily being the change in capital.

47

	Coverage of non- performing loans by capital write-off (%)	Change**	Non- performing loans*	Change in non- performing loans	Capital	Change in capital
Albania	56.8	14.0	14.4	7.8	9.1	-4.9
BiH	40.2	2.3	11.7	8.6	18.0	15.7
Bulgaria	71.2	-37.8	13.5	11.0	10.6	48.4
Croatia	39.3	-9.4	11.5	6.6	14.2	23.6
Macedonia	103.3	-14.8	9.1	2.4	11.0	25.8
Montenegro	30.7	-42.9	21.0	17.8	10.5	53.4
Romania	58.6	-1.7	13.4	10.6	7.8	9.5
Serbia	141.5	-46.3	18.6	7.3	21.0	67.3
Slovenia	70.0	-9.3	5.1	3.3	8.8	0.4

Table 12: Financial system in the region, estimates for 2011

* Non-performing loans as a percent of total loans.

** Change refers to the previous year.

Source: Bole, 2011.

One of the concerns for Albania is also its trade balance (Table 13).

Table 13: External sector in Albania, in percent of GDP

	Actual				Estima	ate	Projec	tion	
	2005	2006	2007	2008	2009	2010	2011	2012	2013
Current account	-6.5	-5.9	-8.6	-14.7	-15.2	-11.4	-10.3	-9.2	-8.5
balance (incl.									
official transfers)									
Trade balance	-24.0	-24.3	-25.9	-28.0	-27.2	-22.2	-21.4	-20.6	-20.0
International	4.0	4.7	4.3	4.0	4.0	4.0	4.0	4.0	4.0
reserves (in months									
of imports)									
Remittances	13.7	14.1	13.5	11.2	11.1	11.0	10.9	11.0	11.0
FDI	3.3	3.6	6.0	7.0	8.2	5.3	5.3	5.3	5.2

Source: World Bank, 2010a.

The current account balance continues its negative trend, and since 2010 the current account deficit is around 10 percent. The trade deficit is much higher at around 20 percent. Thus far, financing the trade deficit was not a problem, as about

one-half was covered by remittances and one-quarter by FDI. Interestingly, other capital inflows of primarily unidentified sources (unrecorded remittances) funded about one-third according to the World Bank (2010a, p. 7), while the balance was financed by loans.

Exports have yet to become a major driver of growth. So far, since the end of the 1990s, exports have increased significantly from 10 percent to around one-third of GDP. However, due to the industry's high elasticity to EU growth rates (on average the elasticity is 4, but in construction elasticity is 8), further European problems could harm Albania (World Bank, 2010a). At the moment the country is in danger also because of the risk of spill-over effects from Greece in terms of lower investment, remittances, and financial system crunch.

In the future the main question is how to continue such a growth trajectory. The success, or lack thereof, will depend primarily on the dedication to further improve the business environment, promote exports (also to ease problems with the current account deficit), attract foreign capital, which might become easier conditional on success with European Union negotiations and pacts (SAA, CEFTA), and improve infrastructure to help exports, FDI, and tourism to promote exports (EBRD, 2011b, EBRD, 2008). According to the IMF (2011), the crisis helped reduce the external imbalance and the financial system did not break, although asset quality declined.

Country growth strategy (World Bank, 2010a and 2010b) stresses that becoming an upper middle-income country is challenging. Up until now, the productivity rise resulted from inter-sectoral shifts of labor, but in the future this process will become largely dependent on intra-industry changes. Momentarily, Albania's competitiveness in world markets is primarily due to its lower costs. In the future

the quality of products will become increasingly important. Therefore, investment in innovation, skills, and human capital in general is important. Increasing globalization will also make the country more vulnerable in terms of exposure both to competition as well as external shocks like the current crisis. Thus, it is becoming increasingly important for the government to provide a solid foundation on the basis of which the private sector (domestic and foreign) can develop.

References

Bahiti, R., & Shahini, B. (2010). Strengthening Innovation and Technology Policies for SME Development in Albania. *Economic Alternatives, 1,* 119-136.

Bezemer, D. J. (2001). Post-Socialist Financial Fragility: The Case of Albania. Cambridge Journal of Economics, 25(1), 1-23.

Bole, V. (2011). Gospodarstvo v 2012. Dostop do posojilnih skladov ali propad okrevanja? Presentation at the Poslovna konferenca Portorož, 17th-18th November, 2011. Retreived January 20th from http://beta.finance-on.net/files/2011-11-21/4_Bole_predavanjeportorozt11%20def.pdf.

Doing Business. (2011a). *Economy rankings*. Retrieved August 10th from http://www.doingbusiness.org/rankings.

Doing Business. (2011b). *Ease of doing business in Albania*. Retrieved August 10th from http://www.doingbusiness.org/data/exploreeconomies/Albania.

EBRD. (1999). *Transition report. Ten years of transition*. London: European Bank for Reconstruction and development. Retrieved August 10th from http://www.ebrd.com/downloads/research/transition/tr99.pdf.

EBRD. (2000). *Transition report. Employment, skills and transition*. London: European Bank for Reconstruction and development. Retrieved August 10th from http://www.ebrd.com/downloads/research/transition/tr09.pdf.

EBRD. (2001). *Transition report. Energy in transition*. London: European Bank for Reconstruction and development. Retrieved August 10th from http://www.ebrd.com/downloads/research/transition/tr01.pdf.

EBRD. (2002). *Transition report. Agriculture and rural transition*. London: European Bank for Reconstruction and development. Retrieved August 10th from http://www.ebrd.com/downloads/research/transition/tr02.pdf.

EBRD. (2003). *Transition report. Integration and regional cooperation*. London: European Bank for Reconstruction and development. Retrieved August 10th from http://www.ebrd.com/downloads/research/transition/tr03.pdf.

EBRD. (2004). *Transition report. Infrastructure*. London: European Bank for Reconstruction and development. Retrieved August 10th from http://www.ebrd.com/downloads/research/transition/tr04.pdf.

EBRD. (2005). *Transition report. Business in transition.* London: European Bank for Reconstruction and development. Retrieved August 10th from http://www.ebrd.com/downloads/research/transition/tr05.pdf.

EBRD. (2006). *Transition report. Finance in transition*. London: European Bank for Reconstruction and development. Retrieved August 10th from http://www.ebrd.com/downloads/research/transition/tr06.pdf.

EBRD. (2007). *Transition report. People in transition*. London: European Bank for Reconstruction and development. Retrieved August 10th from http://www.ebrd.com/downloads/research/transition/tr07.pdf.

EBRD. (2008). *Transition report. Growth in transition.* London: European Bank for Reconstruction and development. Retrieved August 10th from http://www.ebrd.com/downloads/research/transition/tr08.pdf.

EBRD. (2009). *Transition report. Transition in crisis?* London: European Bank for Reconstruction and development. Retrieved August 10th from http://www.ebrd.com/downloads/research/transition/tr09.pdf.

EBRD. (2010). *Transition report. Recovery and reform.* London: European Bank for Reconstruction and development. Retrieved August 10th from http://www.ebrd.com/downloads/research/transition/tr10.pdf.

EBRD. (2011a). *Macroeconomic indicators*. European Bank for Reconstruction and Development. Retrieved August 10th from http://www.ebrd.com/downloads/research/economics/macrodata/sei.xls.

EBRD. (2011b). *Albania Strategy*. European Bank for Reconstruction and Development. Retrieved August 10th from http://www.ebrd.com/pages/country/albania/strategy.shtml.

EBRD. (2011c). Albania country assessment. European Bank for Reconstruction and Development. Retrieved August 10th from http://www.ebrd.com/pages/research/publications/flagships/transition/albania.shtml,.

EBRD. (2011d). *Transition report. Crisis in Transition: The People's Perspective*. London: European Bank for Reconstruction and development. Retrieved August 3rd 2011 http://www.ebrd.com/downloads/research/transition/tr11.pdf.

EBRD. (2012). *Economic forecasts and other regional economic prospects data*. London: European Bank for Reconstruction and development. Retrieved July 3rd 2012 from http://www.ebrd.com/pages/research/economics/data/macro.shtml.

Estrin, S. (1998). Privatization and restructuring in Central and Eastern Europe. In Boone, P. and Gomulka, St. and Layard, R., (eds.) *Emerging from Communism: lessons from Russia, China and Eastern Europe* (pp. 73-98). Cambridge, Massachusetts: MIT Press.

Eurostat. (2011). *Database*. Retrieved August 3rd 2011 from http://epp.eurostat.ec.europa.eu/portal/page/portal/eurostat/home/.

Gomulka, S. (2000). *Macroeconomic policies and achievements in transition economies, 1989-1999.* CEPDP, No. 475. London, UK: Centre for Economic Performance, London School of Economics and Political Science.

Gruda, S., & Milo, L. (2010). *SMEs development and competition policy in Albania*. PECOB (Portal on Central and Eastern Europe) Paper Series No.6/December 2010.

Hashi, I., & Xhillari, L. (1999). Privatisation and transition in Albania. Post-Communist Economies, 11(1), 99-125.

IMF. (2008). Albania: Poverty Reduction strategy paper-National Strategy for development and integration. IMF country report NO.08/269. Retrieved August 3rd 2011 from http://www.imf.org/external/pubs/ft/scr/2008/cr08269.pdf.

IMF. (2011). Albania and the IMF. International Monetary Fund. Retrieved August 3rd 2011 from http://www.imf.org/external/country/ALB/index.htm.

Instat. (2010). *Albania in figures 2010*. The statistical service in Republic of Albania. Retrieved July 29th 2011 from http://www.instat.gov.al/graphics/doc/downloads/publikime/2010/Shqiperia_ne_shifra %202010.pdf.

Instat. (2011). *Database*. The Statistical Service in Republic of Albania. Retrieved through July to November 2011 from http://www.instat.gov.al/.

Jeffries, I. (1990). A guide to the socialist economies. New York: Routledge.

Malaj, A., & Memaj, F. (2003). Ownership Structure and Corporate Governance in the Albanian Postprivatization. Second Conference of ASECU, Belgrade October 6-9, 2003.

Markova, E. (n.a.). Economic and Social Effects of Migration on Sending Countries: The cases of Albania and Bulgaria. Retrieved November 10th 2011 from http://www.oecd.org/dataoecd/34/4/38528396.pdf.

Memaj, F., & Dika, I. (2005). *Privatization in Albania*. University of Frankfurt: Peter Lang Verlag, Europäischer Verlag der Wissenschaften.

Memaj, F., & Koci, N. (2001). *Post privatization in Albania: Success or Failure*. International conference "Integration and transformation in Albania, Hungary and Macedonia" Bamberg, Germany, 5-7 July 2001. Retrieved November 10th 2011 from http://www.seedcenter.gr/projects/MNE/1stconfer/1stconf_papers/Mema.pdf.

Ministry of Economy, Trade and Energy. (2009). Raport mbi ekonomine e Shqiperise (Report on Albanian Economy). (2009). Republic of Albania, Ministry of Economy, Trade and Energy, Tirana.

Ministry of Economy, Trade and Energy. (2011). *Strategic programme for the development of innovation and technology of SMEs for the period 2011-2016*. Republic of Albania, Ministry of Economy, Trade and Energy, Tirana. Retrieved January 28th 2012 from http://www.mete.gov.al/doc/20110218092915_english_draft_programi_strategjik_dt._10.0 1.2011.pdf.

Muço, M. (1997). *Economic transition in Albania: political constraints and mentality barriers*. Tirana: University of Tirana. Retrieved July 27th 2011 from http://www.nato.int/acad/fellow/95-97/muco.pdf.

National Agency of Mineral Resources. (2010). *Invest in Albanian natural resources*. National Agency of Mineral Resources. Retrieved November 10th 2011 from http://www.akbn.gov.al/editor_files/file/content/Minierat.pdf.

OECD. (2003). Albania: Enterprise policy performance assessment. Prepared by Organization for Economic Cooperation and Development and European Bank for Reconstruction and Development. Retrieved November 10th 2011 from http://www.oecd.org/dataoecd/37/21/30632915.pdf.

Petrin, T. (1989). Restructuring of the Yugoslav Economy through Entry of New Firms. Amherst: University of Massachusetts at Amherst.

Policy options for wood industry. Country report: Albania. (2009). Workshop on Policy options for wood energy. Dubrovnik, Croatia, 17-20 November 2009. Retrieved November 17th 2011 from

http://www.fao.org/fileadmin/user_upload/Europe/documents/Events_2009/WoodEnergy/Albania_Report_en.pdf.

Prašnikar, J., Koman, M., & Qorraj, G. (2010). The enterpreneurship in developing countries: a case of Kosovo. *Transform. bus. econ. (Spausd.), 9(3),* 129-146.

Sallaku, S. (2001). *Mass privatization in Albania*. Master Thesis. Stok on Trent, UK: Staffordshire Univesity.

Transparency International. (2011). Corruption perceptions index 2010 results. RetrievedDecember10th2001fromhttp://www.transparency.org/policy_research/surveys_indices/cpi/2010/results.

UNCTAD. (2012). *Database*. United Nations Conference on Trade and Development. Retrieved November 10th 2011 from http://unctadstat.unctad.org/ReportFolders/reportFolders.aspx.

Williamson, J. (2004). *The Washington Consensus as Policy Prescription for Development*. A lecture in the series "Practitioners of Development" delivered at the World Bank on January 13, 2004.

World Atlas. (2011). Albania. Retrieved October 24th 2011 from http://www.worldatlas.com/webimage/countrys/europe/al.htm.

World Bank. (2010a). Country Partnership Strategy for Albania for the period FY11-FY14. Retrieved December 10th 2001 from http://www-wds.worldbank.org/external/default/WDSContentServer/WDSP/IB/2010/06/25/000334 955_20100625012857/Rendered/PDF/541880CAS0P118101Official0Use0Only1.pdf.

World Bank. (2010b). Albania: The New Growth Agenda. A Country Economic Memorandum. Retrieved December 10th 2001 from http://www.wds.worldbank.org/external/default/WDSContentServer/WDSP/IB/2010/11/21/000333 037_20101121235513/Rendered/PDF/535990ESW0gray1e0only1910BOX353785B.pdf.

World Bank. (2011). Gross national income per capita 2010, Atlas method and PPP.RetrievedDecember10th2011fromhttp://siteresources.worldbank.org/DATASTATISTICS/Resources/GNIPC.pdf.

World Bank. (2012). Albania at a glance. Retrieved April 5th 2012 from http://devdata.worldbank.org/AAG/alb_aag.pdf.

WTO Database. (2011). World Trade Organization. Retrieved December 10th 2011 from http://stat.wto.org/CountryProfile/WSDBCountryPFView.aspx?Language=E&Country=AL.

WTO Database. (2012). Members and observers. Retrieved April 2nd 2012 from http://www.wto.org/english/thewto_e/whatis_e/tif_e/org6_e.htm. Zickel, R.E., & Iwaskiw, W.R. (1992). *Albania: a country study*. Minesotta: University of Minesotta.

INTANGIBLE CAPITAL IN ALBANIA: RESEARCH METHODOLOGY

1 Intangible capital

Today, economic value is often created by intangible capital, both in developing and developed countries. Therefore, the role of intangible capital and investment in intangibles is gaining attention in various fields of economic and business science. Research at both macroeconomic and microeconomic levels confirms that intangible capital increases value added, productivity and growth (Corrado et al., 2009, VanArk et al., 2002, Fukao et al., 2009, Miyagawa et al., 2010, VanArk et al., 2009 and other).

The study of intangible capital in Albania was based on the prevailing definition of intangible capital and own extensions of the definition conditional on the current situation in these developing economies and global trends. Many authors have similarly defined the concept of intangible capital (e.g. Buguise et al., 2000, Fernandez et al., 2000, etc.), differing mainly on the focus of respective analyses (accounting, IT, etc.). The line of research focusing on the role of intangible capital in the process of economic growth and its link with productivity relies mainly on the seminal work by Corrado et al. (2009), which divides intangible capital into

three types: computerized information, innovative property and economic competencies (Table 1).

Type of intangible asset	Further classification
Computerized information	Software
	Databases
Innovative property	R&D, including social sciences and humanities
	Mineral exploration and evaluation
	Copyright and license cost
	Development costs in financial industry
	New architectural and engineering designs
Economic competencies	Brand equity (advertising expenditure, market research)
	Firm specific human capital (continuing vocational training,
	apprentice training)
	Organizational structure (purchased, own account)

Table 1: Intangible asset classification

Source: Corrado et al., 2009.

This study of intangible capital focused on Albania extended the definition of intangible capital to include relational, informational capital and social capital. All three types represent both theoretical and methodological innovation in intangible capital literature.

On one hand, informational capital refers to a firm's knowledge about its products, production processes, customers, and resources. It also includes knowledge about competitors' products, production processes, customers, and resources. On the other hand, relational capital includes the stock of relationship with customers, suppliers, competitors, government agencies, and unions (Hunt, 2000). Both types of capital are extremely important in building and sustaining quality and stability in the production and distribution chain. For firms in developing countries, both types of capital are crucial in finding ways for these firms to internationalize (Paley, 2011).

Firm behavior is examined by the bargaining model, focusing on the relationships between workers, management and owners, that is interest groups, and the impact of their relative power on firm behavior and strategy. Social capital is especially important in the former transition economies since it provides information on the direction of privatization and its strategic consequences. It also reflects the characteristics of labor markets and the ways the country is building its human capital.

The study of intangible capital in Albania, therefore, focused on the following aspects of intangible capital: (1) informational and relationship capital, (2) information technology (IT), (3) branding and brand capital, (4) innovation, (5) interest groups in the firm (social capital), and (6) human capital characteristics and organizational characteristics. Underdeveloped financial markets in Albania also play a major role in financing different types of investments, including intangibles. Hence, special focus is given in our work to test the pecking order hypothesis and different methods of capital budgeting procedures.

In continuing, a methodological overview is provided. In the second section, the questionnaire is first described, followed by the description of the sample and survey in Albania.

2 Research design

2.1 Questionnaire structure

The questionnaire for the study of intangible capital in Albania comprised of 7 sections focusing on different aspects of intangible capital (informational and relationship capital with information technology, branding and brand capital,

innovation and R&D, social capital, HRM and organization, and finance), while the last section focused on general company data, primarily financial data. This section provides only a brief overview of each sub-section of the questionnaire. A detailed presentation will be provided separately in each of the chapters dealing with each specific topic.

Information and communication technology (4 questions). ICT was examined in three subsections. First, the size of ICT in 2009 was examined, given the expectation that firms with more ICT would be more productive. Then, the importance of ICT in a firm is examined via the hierarchical level of the IT manager within the firm. The higher the position of the IT manager, the more productive a firm should be. Last, the strategic importance of ICT in company documents and the implementation of IT plans are analyzed.

Relational and informational capital (10 questions) was measured using a three-part questionnaire focusing on customers, competition and suppliers. First, the export orientation of a firm is examined as a major part of informational capital. Second, we examine how the companies follow and cooperate with the customers and involve them in product development. Third, we examine a firm's business environment, primarily the intensity and consequences of competition, as this can either enhance or reduce innovation activity. Last, we focus on the suppliers and their origin, since it is expected that there is also a link between supplier and company performance in terms of innovation.

Research and development (10 questions). The questionnaire on R&D comprised of 10 questions focusing on four major elements of R&D: product and process innovation, sources of information, organization of R&D activity and competences and capabilities. The analysis of R&D was based on the premise that companies

differ significantly both in terms of their origin (developed, developing country) and target market (home based, exporter). Companies from developed countries or those selling to developed markets were expected to be more innovative. Regarding product innovation, we were interested in company comparative performance in terms of product innovation, types of product innovation, the share of revenue dedicated to R&D and the organization of R&D. Companies were asked to share information on process innovation. Next, both internal and external sources of innovation ideas and information relevant for innovation were examined. The last set of questions refers to competences: technological, marketing and complementary.

Branding and brand capital (5 questions). Intangible capital is largely dependent on the activities of the marketing sector. The questionnaire consists of five sets of questions to managers that cover decisions in the fields of brand development, brand value, brand investment, marketing innovation and future orientation. Brand development was measured through brand management activities and included questions on whether or not companies develop their own brands, if they develop corporate brands in addition to the separate brands for their products/services, and also, whether or not they have developed brand architecture. Brand management activities related to brand value were estimated by examining if companies have legally protected company brands (with patents, trademarks), finance activities to increase brand value, either corporate brands or product/services brands, and, finally, if they measure brand value, either corporate of sales for activities to increase the value of brands were measured (including external costs of advertising and marketing activities of advertising agencies and media).

Interest groups within companies (7 questions) are especially interesting from the transitional perspective, given that the behavior and strategies of firms can be linked to ownership structure and, consequently, linked to the privatization process in transition countries. In addition, the characteristics of labor markets also determine corporate governance. The sub-section could be roughly divided into three parts. First, the ownership structure is analyzed. Next follows an analysis of employment characteristics in firms, the nature of wages, and presence of worker unionization. Last, decision making and risk sharing are both analyzed

HRM and organization sub-section (5 questions) analyzed the three most important aspects: human capital and motivation, organizational climate and organizational structure. In particular, the questions focused on the organization of training within companies and the extent of such activities, the transfer of knowledge, the dynamics of dealing with key employees, and the performance measures of workers. Employee satisfaction and motivation were also examined, as was the flexibility of the organization.

Finance and investment (5 questions). The analysis stretched beyond intangible capital to capture the nature of finance and the resources devoted to both tangible and intangible investment. A firm's financial policies can help accelerate or undermine the accumulation of investments in general, including investments in intangibles. To examine the financial behavior of Albanian firms and the level of development of financial markets in Albania, research on intangibles should thus be supplemented with research on financial policies, with both aspects having a strong influence on investments in intangibles. The questionnaire focused on the sources of finance. Capital structure and the criteria used in financing decisions were examined.

In addition to this, the questionnaire also asked for specific data referring to the company, sales, employment, wages and other costs. The data was needed to analyze the impact of intangible capital on productivity.

In total, the questionnaire comprised of 46 questions, the majority of which were structured.

2.2 Questionnaire methodology

The questionnaire was carefully designed using three main types of questions: cascade type of questions, Likert scale questions and standard questions asking for a specific piece of information (expenditure, etc.).

The majority of questions were based on the cascade type of question following Miyagawa et al. (2010). This is a set of three simple 'yes/no' statements. Each statement within the trio was carefully designed so that each additional yes means that the company is in some aspect at a higher level of development (see Table 1 for an example). Such an approach to building survey questions enables the creation of a measurement scale from 1 to 4, which allows empirical testing. If the first answer is 'no', the company is awarded '1'. If the first answer is 'yes', it is awarded '2'. If the answer to the second sub-statement is 'no', the value remains at '2', if the answer is 'yes', it rises to '3'. If all sub-statements get affirmative answers, the total value amounts to 4, indicating the highest possible attainment in a specific field. In the example question for instance (Table 2), if the company answered 'yes' to all statements, this means that the company also introduced new global products,

indicating the company's high potential in introducing new products.¹ Such an answer was translated into a numeric value of 4.

Table 2: Example of cascade question from innovation: question 2

2	Introducing new products	NO	YES
	The company introduced a significant number of new products in our		
	relevant market in the past few years.		
	The majority of those products were not new only for the company,		
	but were also new to the market we work in.		
	We introduced also products that were a novelty in the global markets.		

Source: Intangibles questionnaire for Albania, 2011.

The use of this cascading technique was an important innovation. Questionnaire testing conducted during the year prior to the actual study revealed that companies often have insufficient data, especially when it comes to hard data. Also, the testing clearly showed that the smaller, more diversified or less advanced the company was, the harder it was to obtain a reliable (hard data type) answer. Personal interviews using the questionnaire revealed how important it is not to pose questions that are too specific, long, complicated or detailed, as the answers might be completely different than what was asked (or expected as an answer). Therefore, the simple yes/no cascades allow full capture of the problem while ensuring data quality and reliability.

Second, the questionnaire comprised also some Likert scale questions using a 1 to 5 scale or a 1 to 3 scale, depending on the focus of the question (Table 3).

¹ But it must be noted that all questions from R&D referred to a company's 'relevant market', that is the market in which that the company sold its major share. Consequently, the question could also refer to domestic or even regional markets. But nonetheless, it indicates the potential of a company to stretch beyond its current borders.

⁶²

Table 3: Example of a Likert scale question from innovation

7	Technological competences	1	2	3	4	5
	Research and development in the firm					
1	is advanced.					
	Number of available technological					
	capabilities inside the firm or through					
2	strategic partnership is quite large.					
	We are good at predicting technological					
3	trends.					

Source: Intangibles questionnaire for Albania, 2011.

Cascade type questions were supplemented also by standard questions asking for a specific piece of information (market shares, sales, expenditure, etc.) (Table 4). This type of question was used primarily in the last part of the questionnaire where specific company data was requested, which was needed for the study of the link with productivity.

Table 4: Example of a question with specific data from branding/marketing section

Please, estimate the percentage of sales (in terms of total sales) to a certain area in each of the following years!				
	2009	2008		
Albania				
other countries of former Yugoslavia				
EU-15				
countries of former Soviet Union				
rest of the world				

Source: Intangibles questionnaire for Albania, 2011.

The questionnaire we briefly presented is different from existing standardized questionnaires in at least three major aspects. First, it is focused more on developing countries and provides answers to understanding the development gap within and between countries.

Second, the questionnaire primarily uses the cascading technique based on simple 'yes/no' questions, which adds to the quality and reliability of data and does not limit the potential for statistical analysis. The questionnaire captures the entire intangible capital structure, but at the same time keeps the questions simple. We supplement the descriptive data also with some hard data.

Third, the questionnaire allows us to obtain data on which types of intangible capital companies have, how processes are conducted, what the results related to these processes are, that is, whether intangible capital is also being used in an appropriate manner. The questionnaire was carefully designed to also capture these aspects, which is again extremely important in analyzing developmental problems at the firm level as well as a comparative perspective.

Last, given many similarities with the standardized questionnaires in selected aspects of intangible capital, the methodology applied nonetheless allows many comparisons, which is extremely important for future comparative analysis with the developed countries.

2.3 Survey description

The survey was conducted on a sample of 40 Albanian firms in 2011. The sample of firms was not chosen randomly. Due to limited resources, the sample was rather carefully chosen using the snowball method so as to represent the most typical structure of Albanian firms by industry.

The sample consisted of 12 joint stock companies and 28 companies with limited liability. Ten companies (25 percent) were from the construction industry, 37.5 percent of the sample (15 companies) was from the manufacturing sector, while

37.5 percent of companies were from the tertiary sector: 15 percent of companies (6 companies) from trade and 22.5 percent of the sample (9 companies) are from service activities other than trade.

Twenty-seven companies were involved in business-to-business operations, while 13 companies were selling their products to final markets. The companies sold the majority of their products/services in domestic markets (over 80 percent on average), although the majority of the sample (28 companies) were involved in foreign trade to at least some extent.

The sample also well covered the size structure; in the year 2010 companies on average employed 148 people, the smallest company again employed 4 employees, and the largest employed 990. In 2010 the total number of employees in the sample was 5,901.

In order to become familiar with the questionnaire, a company first received it by mail. The questionnaire was then answered during an interview with one of the Albanian research team members (researchers from the Faculty of Economics, University of Tirana). The quality of input data could thereby be directly controlled. The questionnaire was answered by a company's CEO, financial or HR manager, or, in some cases, a combination of the two in cases where specific data (e.g. financial data) was requested.

One of the focal points of the research was the link between a specific type of intangible asset and company performance. To analyze this link, the relationship between productivity and intangible asset type was analyzed. For this purpose, the sample was divided further into two groups, the dividing line being the median firm. Any other division of the sample, if not noted differently in the chapter, was also conducted using the median.

It should be noted that this type of research required great efforts from both researchers and companies. In some cases it was noted that the questionnaire was long and demanding. Thus, the cooperation of both researchers and companies is greatly appreciated. As a result of these joint efforts, the forthcoming chapters present an in-depth picture of the Albanian economy, which until now has been unavailable.

3 Conclusion

The study of intangible capital in Albania is the first of its kind and provides valuable information about the Albanian economy based on firm-level data. The research team developed the survey methodology applied with the view of the specifics of developing markets. The survey comprised of subsections analyzing each component of intangible capital separately, adding two new components to the standard intangible capital definition: informational and relational capital, and social capital.

The results provide an interesting and consistent explanation of a rather domestically oriented economy, at the moment, striving to become a more exportoriented economy. The analysis begins with the study of informational and relationship capital, ending with HRM.

References

Ark, B., Inklaar, R., & McGuckin, R. (2002). *Changing Gear: productivity, ICT and services: Europe and the United States.* Groningen Growth and Development Centre. Research Memorandum GD-60.

Buguise P., Jacquemin, A., & Marchipont, J. F. (2000). *Competitiveness and the value of intangible assets*. Cheltenham: Edward Elgar.

Corrado, C., Hulten, C., & Sichel, D. (2009). Intangible Capital and U.S. Economic Growth. Review of Income and Wealth, 55, 661-685.

Fernandez, E., Montes J., & Vazquez, C. J. (2000). Typology and strategic analysis of intangible resources: A resource-based approach. *Technovation, 20 (2)*, 81-92.

Fukao, K., Miyagawa, T., Mukai, K., Shinoda, Y., & Tonogi, K. (2009). Intangible Investment in Japan: Measurement and Contribution to Economic Growth. *Review of Income and Wealth*, 55, 717–736.

Hunt, S. D. (2000). A general theory of competition. Thousand Oaks, CA: Sage.

Miyagawa, T., Lee, K., Kabe, S., Lee, J., Kim, H., Kim, Y., & Edamura, K. (2010). *Management Practices and Firm Performance in Japanese and Korean Firms - An Empirical Study Using Interview Surveys.* RIETI Discussion Paper Series 10-E-013.

Paley, T. (2011). The rise and fall of export-led growth. Levy Economic Institute Working Paper.

VanArk B., Hao, J.X., Corrado, C., & Hulten, C. (2009). Measuring intangible capital and its contribution to economic growth in Europe. European Investment Bank Papers 14-1.

RELATIONAL, INFORMATIONAL, AND ITC CAPITAL

1 Introduction

In today's global world the role of intangible capital is becoming more important in achieving competitive advantage and, as such, it becomes critical to understand how a firm's intangible capital is affecting firm performance.

Resource-advantage theory (Hunt, 2000) states that a firm's resources are leveraged to provide for competitive advantage. Griffin et al. (2010) argue that competitive advantage is founded mainly on intangible resources, especially on human, organizational, relational, and informational capital. Organizational capital covers a firm's policies and norms, whereas human capital covers business skills and the knowledge of a firm's employees. Informational capital constitutes a firm's knowledge about its products, production processes, customers, and resources on one side, and also a firm's knowledge about competitors' products, production processes, customers, and resources. Relational capital includes firms' stock of relationship with customers, suppliers, competitors, government agencies, and unions (Hunt, 2000). Also, the ability to effectively manage information within the firm while successfully gathering new information about the environment,
customers, and competitors have become critically important since it may provide a basis for gaining a competitive advantage similar to other invisible assets possessed by firms (Sampler, 1998). The ability to obtain information about markets and customers helps to ensure that firms are more attuned to changes in the environment and can result in a competitive advantage over slower, ill-informed competitors (Barney et al., 2001).

In this chapter we will focus on the role played by informational and relational capital, and the role of information and communication technology (ITC) in Albanian firms. More precisely, we will investigate their size and also the effect they have on productivity.

The chapter begins with a brief description of relational capital, informational capital, and capital connected with the investment in ITC, with an emphasis on the measurement of each. Next, we present the size of all types of capital in Albania and their effects on productivity. In the last section we present our conclusion.

2 Measuring Relational, Informational and ITC Capital

Our main hypothesis is that informational capital, relational capital, and ITC capital positively affect a firm's performance, which we measure by productivity. As shown by Griffith et al. (2010), higher levels of relational and informational capital positively influence marketing capabilities (abilities in selling/marketing, product development/research, and the art of distribution), which in turn positively affects firm performance. Regarding investment in information and communication technology, a number of studies show that on the macro level ITC is an important factor in productivity growth (van Ark et al., 2002; van Ark, 2004). However, micro

economic studies on the effect of information technology (IT) on firm performance have presented no conclusive evidence (Tippins and Ravipreet, 2003; Huang et al., 2009). According to a resource-based view, ITC per se may not generate a sustainable advantage because competing firms could adopt the same technology (Clemons and Row, 1991). The advantages of ITC can be protected, however, by embedding them in an organization through complementarity and cospecialization (Powell and Dent-Micallef, 1997). In this respect, knowledge represents an important intangible resource for the firm. Most of the mixed results about the effect of information technology on firm performance can be attributed to the fact that most studies examined IT as a stand-alone resource, neglecting the role of complementarity and cospecialization.

We measured relational and informational capital in firms using a three-part questionnaire. The structure of the questionnaire is as follows: In the first part there are questions about a firm's customers, the second part deals with a firm's competitors, and the third section has questions about a firm's suppliers. Each question (factor) consists of three sub-questions, to which managers were able to give either a YES or NO answer. We will first briefly describe each factor, then the sub-questions of each factor that are shown in Table 2 will be described in detail in the result section.

Let us start with factors dealing with informational and relational capital. The first factor deals with export orientation. This is the only factor that is not measured by YES or NO answers. We have rather measured the share of sales to different regions. These regions are shown in Table 1. Export orientation belongs to a firm's informational capital. Based on the literature on exports and productivity, firms that export more are more productive (Wagner, 2007), and also firms that export to more destinations are more productive (Anderson et al., 2008). The second factor,

monitoring customers, is part of a firm's informational capital since it measures how closely a firm monitors its customers and whether or not it engages customers in the development of new products. More precisely, we asked firms if they meet regularly to exchange views and observations about relationships with their customers, and, if they do, whether or not they meet with their customers to find out about their needs. Finally, we asked firms if they engage their customers in the process of new product development.

The next factor is concerned with the business environment in which a firm operates. Does competition enhance productivity? In economic literature both positive and negative effects from competition on innovation have been found (Cohen et al., 1989). As argued by Aghion et al. (2006), there is an inverse U relationship between competition and innovation. Increasing competition can either enhance or reduce innovation depending on the initial level of competition.

The last factor in relational and informational capital deals with a firm's suppliers, more precisely the origin of suppliers. There is an increasing amount of empirical literature that focuses on the affects of imported inputs on productivity. Kasahara and Rodrigue (2008), Halpern et al. (2009) show that imported intermediates improve a firm's productivity. Intermediate imports allow firms to adapt to technology from abroad (especially if they come from developed markets) and so benefit from foreign research and development. They also allow a firm to focus resources and specialize in activities where it has a particular strength (Anderson et al., 2008). We asked firms about the origin of their suppliers (local markets, Albania, or developed markets). If more than 50 percent of a firm's suppliers (based on the value of total material costs) were from developed markets, the firm should be more productive.

We have analyzed the effect of ITC capital on a firm's productivity by four groups of factors (questions). Each factor focuses on the particular determinant of ITC system in the firms. We will briefly describe each factor while the sub-questions of each factor are shown in Table 3 and will be described in detail in the result section.

The first factor analyzes the share of revenues that the firm invested in ITC in the year 2009. Firms that invest more in ITC should, in general, be more productive. The second factor investigates the hierarchical level of the IT manager in the firm. The higher the position of the IT manager in the firm, the more productive the firm should be. With the next factor we investigate if the firm has a strategic plan for IT development and how this plan is implemented. More active implementation of an IT plan should result in higher productivity. The last factor deals with the role of informatics in current activities, business reorganization, or in achieving a competitive advantage.

3 Results

Our analysis is built only on 40 questionnaires. We obtained a statistically significant difference between the group of more productive and less productive firms for only a few questions. In interpreting the results, we emphasized the direction of the difference.

Albania is still not an export-oriented country even though their exports have increased significantly from the end of 1990. At that time exports represented around 10 percent of GDP, while today this share is around 30 percent (World Bank, 2010). Most of Albania's exports end up in the EU. This is illustrated in Table 1 where we also show the mean value of the share of sales by different regions in the years 2008 and 2009 for the entire sample of firms and two sub samples i.e. more productive and less productive firms. The mean firm from Albania generates only slightly more than 10 presents of its revenues from exports (not much difference between the years 2008 and 2009). Hence, almost 90 percent of a firm's revenues are generated in the domestic market. This implies that the typical Albanian firm is insider oriented. In comparison with Slovenia and the Republic of Srpska, the mean firm from Albania exports substantially less. The mean Slovenian manufacturing firm and the mean firm from the Republic of Srpska exported around two-thirds and one-third of firm revenues in the year 2009, respectively (Koman et al., 2010, Prašnikar et al., 2012). Table 1 also shows that the more productive firms in Albania export less compared to the less productive firms. Hence, market niches exist in domestic markets and allow firms, due to no market saturation or monopoly position, to acquire rents that result in higher productivity. In the year 2008 this difference is also statistically significant. On average, less productive firms exported around 18 percent of their revenues in the year 2008, while more productive firms exported only three percent of their revenues. In the year 2009, the less productive firms exported approximately the same share of revenues as in the year 2008, while more productive firms exported more (eight percent). The same qualitative result holds for firms from the Republic of Srpska, while the result is just the opposite for Slovenian firms. It seems that in less developed countries export is mainly due to distressed export (loan deals), where valued added and, thus, productivity are small (see as an example the case of Dafinor Ltd in Redek et al., 2012, in this book). While in the year 2008 both more and less productive firms in Albania created most of their export revenues from the EU-15 countries, this was not the case for 2009. For less productive firms, the greatest export in the year 2009 was still generated by the EU-15 countries (13 percent of total revenues), but for more productive firms the largest export markets were countries of the former Yugoslavia, with around 6 percent of the share of total revenues. The more productive firms exported only 2 percent of their

revenues to markets of the EU-15 countries in the year 2009, which is statistically significantly less compared to less productive firms (13 percent). This again suggests that exports to developed markets created by less productive firms are mainly the result of "loan" deals. For example, in Slovenia more productive manufacturing firms export more, on average, to the EU-15 markets than less productive firms i.e. 42 percent versus 37 percent (Koman et al., 2010).

	Al	ll firms	More productive firms		Less productive firms			
	n	Mean	n	Mean	n	Mean	T-test	P-value
2008								
Albania	40	0.897	20	0.970	20	0.825	1.937	0.060*
Other countries of former Yugoslavia	40	0.027	20	0.013	20	0.041	0.990	0.328
EU-15	40	0.064	20	0.017	20	0.110	1.507	0.140
Countries of former Soviet Union	40	0.0004	20	0.0008	20	0	1. 371	0.178
Rest of the world	40	0.012	20	0	20	0.025	1.600	0.118
2009								
Albania	40	0.863	20	0.918	20	0.807	1.252	0.218
Other countries of former Yugoslavia	40	0.054	20	0.064	20	0.04	0.362	0.719
EU-15	40	0.076	20	0.018	20	0.134	1.693	0.099*
Countries of former Soviet Union	40	0.0003	20	0.0005	20	0	1.446	0.156
Rest of the world	40	0.008	20	0.00005	20	0.015	1.185	0.243

Table 1: Mean values of shares of sales by different regions in year 2008 and 2009 between more and less productive firms

* Significant at 10 percent

In Table 2 we show the results of the three remaining factors (monitoring customers, business environment and origin of suppliers) that measure relational and informational capital in Albanian firms. More precisely, we show the share in

percent of positive answers for each sub-question for the factor in the more and less productive groups of firms.

The factor monitoring customers consist of three sub-questions that measure how closely firms monitor and engage their customers in business decisions. Seventythree percent of firms in the entire sample regularly exchanged views and observations on customers, and seventy-eight percent of firms planned changes in their supplies. Forty-three percent of firms also engaged their buyers in new product development. The value of this factor is smaller for the more productive group as compared to the less productive group, since the share in positive answers to each sub-question is higher in the less productive group compared to the more productive group even though the differences are not statistically significant. We observed the same in the Republic of Srpska. A larger share of the less productive firms (85 percent versus 70 percent) meets with their customers to find out about their customers' needs. Also, a greater proportion of less productive firms engage their customers in new product development (50 percent versus 35 percent). The most plausible explanation for this is that among less productive firms most of them produce products that are used by upstream firms in their production processes. For them a closer relationship with their customers and customer evolvement in product development is nowadays a quite common practice. This explanation is consistent with previous observations on export orientations (distressed export), since a less productive firm in Albania, on average, exports significantly more to the EU-15 countries than the average more productive firm, and with case studies presented in this book (see the case of Dafinor Ltd and the case of Stella Company in Redek et al., 2012, in this book).

With respect to sub-questions that measure business environment, there are also statistically significant differences between less and more productive firms in

Albania. Seventy percent of more productive firms stated that the activities of their competitors have an impact on their business, compared to 75 percent of less productive firms. This could imply that less productive firms are operating in a more competitive environment. As shown in the case of Stella Company (Redek et al., 2012, in this book), the company is facing strong competition from firms in China and Pakistan. However, when we asked firms how they respond to their competitors' strategic moves, 60 percent of more productive firms stated that they respond aggressively, while only 50 percent of less productive firms stated this response. Since more productive firms operate mainly in domestic market and some of them probably have a monopoly position in their respective market niche, their aggressiveness can be the result of preventing their (potential) competitors to (enter) expand their market share. The results in table 2 also show that 65 percent of less productive firms operate in an environment where at least one company in their core business had more than 20 percent of the market share; compared to only 35 percent in more productive firms (the difference between groups is statistically significant).

With the last three sub-questions we measured the origin of suppliers. In 45 percent of firms in the entire sample more than 50 percent of suppliers (based on the value of total material costs) are from developed markets. This is basically the same as in the Republic Srpska (42 percent) and substantially less than in Slovenian manufacturing firms, where this number was 73 percent. Table 2 also shows that in all three sub-questions, which measure the origin of suppliers, the differences between more and less productive firms are not statistically significant. Although the percentage of suppliers (based on the value of total material costs) from countries other than Albania is larger in more productive firms (80 percent versus 75 percent), both groups of firms have the same percentage of suppliers from developed countries (45 percent).

Table 2: Share in percent of positive answers for each sub-question in the more and less productive	
groups of firms for relational and informational capital	

			ľ	More	L	ess	Chi-	
			pro	ductive	prod	uctive	squar	Р-
	All	firms	f	irms	fi	rms	e	value
		Share		Share		Share	-	
	n	(%)	n	(%)	n	(%)		
Monitoring customers (Custor	mers)						L	
People from different								
functional areas of our								
company meet regularly in								
order to exchange views and								
observations about what's								
going on with our customers.	40	73	20	70	20	75	0.125	0.723
We regularly meet with our								
customers in order to find out								
about their needs.	40	78	20	70	20	85	1.290	0.256
Consumer representatives of								
our products are engaged in the								
process of developing new								
products.	40	43	20	35	20	50	0.927	0.337
Business Environment (Comp	etitors)						
The activities of our major								
competitors have an impact on								
our business.	40	73	20	70	20	75	0.125	0.723
Our company aggressively								
responds to the strategic moves								
of our main competitors.	40	55	20	60	20	50	0.404	0.525
At least one company in our								
core business has a more than								
20% market share.	40	50	20	35	20	65	3.600	0.058*
Origin of suppliers (Suppliers)								
Most of our suppliers are not								
local.	40	88	20	90	20	85	0.229	0.633
More than 50% of suppliers								
(based on the value of total								
material costs) are not from								
Albania.	40	78	20	80	20	75	0.143	0.705
More than 50% of suppliers								
(based on the value of total								
material costs) are from								
developed markets.	40	45	20	45	20	45	0	1

* Significant at 10 percent.

Results in Table 3 indicate the relative position of informatics within the firms tested by different qualitative measures. In our sample only 88 percent of firms invest at least one percent of their revenues in IT. Also, 35 percent of firms in our sample invest at least three percent of their revenues in IT. The above numbers are substantially low if compared to more developed countries like Slovenia (Domadenik et al., 2010). They are, however, higher than in the Republic of Srpska where only 15 percent of firms invested at least 3 percent of their revenues in IT. Both groups of firms (more productive and less productive) invest relatively the same amount of revenues in IT. Thirty-five percent of firms in both groups have a strategic plan for IT. Although this plan is updated every two years in more than 70 percent of firms that have it, the plan and its revisions exist mainly on paper, as only 10 percent of more productive firms (which is 30 percent of firms that have a plan) and an even fewer percentage of less productive firms (5 percent of all less productive firms or 14 percent of firms that have the plan) implement them. This suggests that IT should not be a source of competitive advantage. However, when we asked managers if IT is a source of competitive advantage, almost 50 percent of them said that it is. Hence, managers believe that IT can be a source of competitive advantage, but they are not practicing it in their firms. One possible explanation for this can be the lack of funds that are required to set IT as a source of competitive advantage.

	All firms		More productive firms		Droc	ess luctive	Chi-	
					fi	rms	square	P-value
		Share		Share		Share		
	n	(%)	n	(%)	n	(%)		
Investment in IT								
At least 1% of revenue.	40	88	- 20	90	20	85	0.229	0.633
At least 2% of revenue.	40	48	20	45	20	50	0.100	0.752
At least 3% of revenue.	40	35	20	35	20	35	0	1.000
Position of the IT manager								
IT manager is within the company hierarchical structure ranked higher								
than on the 4th hierarchical level.	40	45	20	45	20	45	0	1.000
IT manager is within the company hierarchical structure ranked higher								
than on the 3rd hierarchical level.	40	33	- 20	30	20	35	0.114	0.736
IT manager is a member of the board of directors (highest management								
level).	40	18	20	10	20	25	1.558	0.212
The IT strategic plan in the company								
Exists.	40	35	20	35	20	35	0	1.000
Is being implemented.	40	8	20	10	20	5	0.360	0.548
Is being updated at least every second year.	40	25	20	35	20	15	2.133	0.144
The role of IT in the company								
IT is NOT considered only as a supporting business service.	40	48	20	40	20	55	0.902	0.342
IT stimulates business processes reengineering.	40	38	- 20	35	20	40	1.067	0.744
IT is a source of competitive advantages for the company.	40	48	20	50	20	45	0.100	0.752

Table 3: Share in percent of positive answers for each sub-question in more and less productive groups of firms for ITC capital

4 Conclusion

Our results indicate that the level of relational, informational, and ITC capital in Albanian firms affects the productivity of firms. Results show that the share of revenues from the domestic market and the share of exports to the EU-15 markets have a statistically significant effect on productivity. However, the effect is not as expected. Firms who export more are less productive, and share of revenues created from developed countries (EU-15) is smaller in more productive firms compared to less productive firms. Anecdotic evidence suggests that this can be due to distressed export (loan operations). Due to low labor costs in less developed countries, firms from developed countries engage firms from less developed countries to do basic operations for them. We found additional support for the above observations mainly in questions dealing with the monitoring of customers and the origin of suppliers. In the case of loan operations, the firm that is producing 'loan' products gets instructions on how to produce from its buyer. As a result the buyers generally provide the firms with raw materials and only inspect if the end products are of good enough quality.

In the long term the distressed export will not be able to persist. As Albania progresses, their low labor cost will increase, which will cause Albanian companies to become uncompetitive. Since export currently represents around 30 percent of Albania's GDP, this will result in a substantial slow down of the Albanian economy. To prevent this scenario from occurring, the Albanian government should focus more on the export-led growth model, in which intangibles play a very important role. The government should create an environment that will be beneficial for establishing firms such as GPG Company and Xherdo Ltd that are described in Redek et al. (2012, in this book).

Our results also show that the typical Albanian firm is inside oriented since it creates most of its revenues from its domestic market. The insider-oriented firm is also more productive since it is able to extract rents. As long as the government does not put substantial effort into improving the business environment, which will result in stronger competition in the domestic market from abroad, Albanian firms that operate mainly in the domestic market will continue to extract rents.

References

Aghion, P., Griffith, R., & Howitt, P. (2006). U-shaped relationship between vertical integration and competition: Theory and evidence. *International Journal of Economic Theory, 2(3-4),* 351–363.

Andersson, M., Lööf, H. & Johansson, S. (2008). Productivity and International Trade: Firm Level Evidence from a Small Open Economy. *Review of World Economics/Weltwirtschaftliches* Archiv, 144 (4), 774–801.

Barney, J. B., Wright, M., & Ketchen, D. J. Jr. (2001). The resource-based view of the firm: ten years after 1991. *Journal of Management*, 27(6), 625–641.

Clemons, E. K., & Row, M. C. (1991). Sustaining IT advantage: the role of structural differences. *MIS Quarterly*, 15(3), 275–294.

Cohen, W. M., & Levinthal, D. (1989). Innovation and Learning: The Two Faces of R&D. *Economic Journal*, 99(397), 569-596.

Domadenik, P., Farčnik, D., & Koman, M. (2010). Intangible investment in ITC in Slovenian manufacturing firms 2006-2009. In Prašnikar, J. (ed.), *The role of intangible assets in exiting the crisis*, (141-148). Ljubljana: Časnik Finance.

Griffith, D. A., Yalcinkaya, G., & Calantone, R. J. (2010). Marketing capabilities consistently mediate effects of firm intangible capital on performance across institutional environments? *Journal of World Business*, 45(3), 217–227.

Halpern, L., Koren, M., & Szeidl, A. (2009). Imported Inputs and Productivity. *Center for Firms in the Global Economy (CeFiG) Working Papers No. 8.*

Huang, Y., Li, E. Y., & Chen, J. S. (2009). Information synergy as the catalyst between information technology capability and innovativeness: empirical evidence from the financial service sector. *Information Research*, 4(3-4), 351–363.

Hunt, S. D. (2000). A general theory of competition. Thousand Oaks, CA: Sage.

Kasahara, H., & Rodrigue, J. (2008). Does the use of imported intermediates increase productivity? Plant-level evidence. *Journal of Development Economics*, 87(1), 106-118.

Koman, M, Filić, R., Flerin, M., & Juriševič, Ž. (2010). Relational and informational capital in Slovene manufacturing firms. In Prašnikar, J. (ed.), *The role of intangible assets in exiting the crisis*, (pp.187-202). Ljubljana: Časnik Finance.

Powell, T. C., & Dent-Micallef, A. (1997). Information technology as competitive advantage: the role of human, business, and technology resources. *Strategic Management Journal*, 18(5), 375 405.

Prašnikar, J., & Knežević Cvelbar, L. (eds.) (2012). Intangible assets as a potential for growth in Republic of Srbska. Ljubljana: Ekonomska fakulteta.

Redek, T., Memaj, F., Prašnikar, J., & Trobec, D. (2012). Albania: Two decades of economic development at a glance. In Prašnikar, J., Redek, T. and Memaj, F. (eds.), *Albania: The Role of Intangible Capital in Future Growth*, (4-59). Ljubljana: Ekonomska fakulteta.

Sampler, J. L. (1998). Redefining industry structure for the information age. *Strategic Management Journal*, 19(4), 343-355.

Tippins, M. J., & Ravipreet, S. S. (2003). IT Competency And Firm Performance: Is Organizational Learning A Missing Link? *Strategic Management Journal*, 24(8), 745–761.

Wagner, J. (2007). Exports and Productivity: A Survey of the Evidence from Firm-Level Data. *The World Economy*, 30 (1), 60-82.

Van Ark, B. (2004). The Measurement of Productivity: What Do the Numbers Mean? In Gelauff, G., Klomp, L., Raes, S., & Roelandt. T. (eds.), *Fostering Productivity Patterns, Determinants and Policy Implications.* Elsevier.

Van Ark, B., Inklaar, R., & McGuckin, R. (2002). *Changing Gear: productivity, ICT and services: Europe and the United States.* Groningen Growth and Development Centre, Research Memorandum GD-60.

World Bank. (2010). Country Partnership Strategy for Albania for the period FY11-FY14. Retrieved April 2nd 2012 from http://www-wds.worldbank.org/external/default/WDSContentServer/WDSP/IB/2010/06/25/000334 955_20100625012857/Rendered/PDF/541880CAS0P118101Official0Use0Only1.pdf.

BRANDING AND BRAND CAPITAL

1 Introduction

Albania is a country little known to consumers from more developed countries, which typically hold prejudiced, convenient and shallow narrative about the country. Modest progress and growing stability do not receive mass media attention outside the country. Therefore, Albania's reputation and perceptions lag significantly behind the reality of the country's notable economic and social development since the early nineties (see Redek et al., 2012, in this book). The exceptions are probably 'professional' audiences such as business investors, bankers, and tour operators in tourism, who seem to be better informed and enthusiastic about the country's prospects (Anholt, 2007). As a former Eastern-bloc country, the country has been rebranding itself as 'a new Mediterranean love,' was named to the top of Lonely Planet's (2011) 'countries to visit,' and was named as an 'emerging destination' by the Financial Times (2011), although the majority of visitors still come from neighboring states (Balkans.com, 2011).

Before discussing branding and brand capital in companies in Albania, it is necessary to explain the country in terms of its development of marketing infrastructure and support services. Namely, distribution and promotion

infrastructure reflect the ease of marketing goods in a country and for businesses, successful execution of distribution and promotion decisions greatly depend on the marketing infrastructure of the country. Among Central and Eastern European countries, Albania was ranked among low attractiveness countries together with Bosnia-Herzegovina and Macedonia in 2001 (Manrai et al., 2001). More recent sources (e.g. Pici and Goga, 2007) show tremendous growth in terms of development of marketing infrastructure. For example, the availability of advertising media improved significantly from low initial numbers in less than a 10year period (Table 1). Pici and Goga (2007) report that the media market in Albania included more than 260 media outlets (outdoors not included). Major media groups have established an Internet presence and actively pursue audience interaction and engagement through prize competitions, on-line forums, surveys etc. Sevrani and Gorica (2011) and Sevrani et al. (2010) report that the Internet is seen as a way of building an image in the long run and as a complementary tool to traditional marketing methods. Pici and Goga (2007) point to a paradox where on one hand the country has very low landline telephone penetration, but with growing PC ownership &internet usage, the country is, on the other hand, an early adopter of mobile TV and HDTV. According to Bedalli (2012), the present advertising structure is the following: in TV more than 60-65 percent, outdoor more than 18-22 percent, press up to 5 percent, and radio about 1.5-2 percent.

Year	Advertising media availability in Albania								
	Nev	vspapers	Radio		Television				
	#	Circulation (per 1000)	#Stations	#Radio sets	#Stations	TV sets			
				(per 1000)		(per 1000)			
1998	4	39.59	18	151.03	9	74.78			
2007	27		54		73*				

Table 1: Advertising media availability in Albania

*includes free-over-the air TV stations, does not include additional 47 cable TV platforms, A DTH platform and 1 mobile TV platform

Source: for 1998: Manrai et al., 2001, for 2007: Pici and Goga, 2007.

⁸⁴

In terms of advertising spending, Pici and Goga (2007) provided an estimation of 23-25 million euros spent on print, electronic and outdoors. This estimation covers the top 50 major ad spenders for the first half of 2007. Advertisers were mostly from industries with more intensive competition, e.g. producers of fast-moving consumer goods and retailing, as well as advertisers from newly established banking organizations and mobile telecommunication operators. Manrai et al. (2001) considered the retailing sector in Albania, particularly in the capital city of Tirana, as already well developed due to large-scale foreign aid and remittances from friends and relatives abroad, combined with a speedy privatization.

Regarding the development of the advertising industry and acceptance of advertising, Manrai et al. (2001) had no information available on the number of advertising agencies available in the country, advertising expenditures by media, or consumers' attitudes toward advertising in Albania. The advertising industry in Albania was considered to be one of the least developed. Pici and Goga (2007) report that the first Albanian advertising agencies developed no earlier than the mid-1990s as small sole proprietary shops owned and operated by practitioners with limited marketing education and advertising experience. In the late 1990s, some of these small 'boutiques', focusing mainly on creative products, were contracted by international full-service agencies to support the local needs of major advertisers. More intensive competition after the year 2000 contributed to the development of a new specialized group of agencies (sales promotion, PR) and advertising groups providing integrated marketing services. New players affiliates of large international agency groups were established, e.g. McCann Erickson Tirana (2012). Pici and Goga (2007) report a growing need to establish trade associations for professional advancement and adoption of common ethical guidelines and widely accepted best practices. Also, there was a lack of use of any consumer insight, brand valuation, or media research tools although such tools are available.

Most the major domestic advertisers tend to rely on the marketing expertise of their employees, while most of the major international advertising brands rely on lead advertising agencies. The market lacks experience and expertise in the part of marketing and media services departments.

The purpose of this chapter is to present branding and brand capital as part of intangible investments in Albanian companies, and to discuss specifics for companies in Albania. According to Corrado et al. (2009), advertising expenditures represent a large part of investments in brand equity and business investments in intangibles. Advertising expenditures have asset/investment-like characteristics: advertising effects on firm intangible assets are accumulative and sustainable (Wang et al., 2009). Our main hypothesis is that branding and brand capital positively affect a firm's performance, which we measure by value added per employee.

Branding is seen as an essential tool that enables price premiums in the market, leading market positions, or market penetration (Kapferer, 2008). Since there is no active market in brands that would provide comparable values, brands are very special intangible assets (Interbrand, 2004). In many businesses, brands influence choices that are crucial to firm performance and the creation of shareholder value (Morgan and Rego, 2009). In competitive markets, branding includes a number of decisions relating to brand expansion, co-branding, participation in trade brands, and the overall brand architecture (Petromilli et al., 2002). Thus, branding includes various aspects of brand awareness (recall and recognition), perceived quality, and level of loyalty (Keller, 2003).

Brand management strategy refers to brand development, brand measurement and control, as well as concern for brand value. *Brand development* includes several aspects of brand management activities, including whether the company develops its own brands (product/services brands and corporate brands) and whether it has

developed brand architecture. *Brand value* related activities start with legal protection of company brands (with patents, trademarks), and build with activities to increase brand value and measure brand value (brand investment or the share of sales for activities to increase the value of brands). Innovations are related to branding and company growth and productivity (Tellis et al., 2009, O'Mahony and Vecchi, 2009). Companies that rank higher on branding and brand-related activities should also implement more innovation within their marketing mix. In terms of a company's future, different channels and media impact company sales and brand equity effects (Keller, 2010).

The chapter presents an overview of brand-related marketing activities. The structure is the following: following a brief explanation of Albanian specifics and development in terms of the development of marketing infrastructure and supporting services, activities related to brand capital, brand development, and brand value in Albania are examined together with reported marketing activities to sustain branding activities and ensure a company's future. A comparison of companies from different industries, with legal forms, export orientation, size, and value added are reported. Results are discussed and conclusions are taken from the findings.

2 Data analysis of branding and brand capital in Albania

The analysis of branding and brand capital in companies in Albania relies on responses from 40 Albanian managers representing companies of different size, industry, legal form and export orientation. Brand development was measured through brand management activities, including the development of own brands, corporate brands in addition to brands for own products/services, and the

development of brand architecture. Based on the answers provided in the questionnaires, there are two distinct clusters of companies: companies with at least two of the above three aspects are ranked higher in brand development (50 percent of companies in the sample, see Table 2), while the others are ranked lower on brand development (50 percent of companies in the sample).

Frequency			Percent
Brand	Lower	20	50.0
development	Higher	20	50.0
activities	Total	40	100.0

Table 2: Brand development marketing activities for companies in Albania

Brand management activities related to brand value were estimated through investigation whether companies have legally protected company brands (with patents, trademarks), whether they finance activities to increase brand value, either corporate brands or product/services brands and, finally, whether they measure brand value, either corporate brands or product/services brands. Again, companies with at least two of the three activities related to brand value were classified as companies high on brand value measurement (38 percent of companies in the sample, see Table 3), while companies with one or none of brand management activities related to brand value were classified as low on brand value measurement (62 percent of companies). Brand development and brand value activities are correlated (symmetric measure phi has a value of 0.671, sign. 0.00). Only 15 percent of companies are high on brand development and low on brand value, while only 2.5 percent of companies are high on brand value and low on brand development. For all other companies there is a congruency between both dimensions of brand management (both low are 48 percent of companies, both high at 35 percent of companies).

Table 3: Brand value marketing activities for companies in Albania

	Clusters	Frequency	Percent
Brand	Lower	25	62.5
value	Higher	15	37.5
activities	Total	40	100.0

Brand investment or share of sales for activities to increase the value of brands (including external costs of advertising and marketing activities of advertising agencies and media) was examined. Only a fraction of companies responded (Table 4). The reported investments are in the range of 0 to 30 percent and are growing in the period 2006-2009.

Table 4: Brand investments for companies in Albania (as percentage of sales)

Year	Ν	Mean	Std. Deviation
2006	18	6.7	7.9
2007	18	7.6	9.6
2008	19	9.6	11.3
2009	19	9.5	11.4

In terms of marketing innovations, 68 percent of companies reported innovations in terms of marketing communications (new media or promotion techniques), the same percentage (68 percent) in product design or packaging, 65 percent in marketing channels and 83 percent in new forms of pricing, while 12.5 percent of companies reported no marketing innovation. Regarding marketing innovation, clusters of companies according to brand development activities and brand value activities were compared. Companies in the cluster with more developed brand development activities reported significantly more areas of innovation and also, regarding brand value clusters, companies in the cluster with more brand values activities reported significantly more areas of innovation and also,

	Cluster	N	Mean	Std. Deviation	Std. Error Mean	Т	Sig. (2- tailed)
Brand	Lower	- 20	2.15	1.565	.350		
development	Higher	20	3.50	1.000	.223	-3.250	.003
activities	_						
Brand value	Lower	25	2.24	1.535	.307	4 506	000
activities	Higher	15	3.80	.560	.145	-4.390	.000

Table 5: Marketing innovations for companies in Albania*

*Scale for marketing innovations: number of areas for reported innovations (for four 4Ps), range 0-4

Preparations for future marketing seem to be very high for companies in Albania: 73 percent of companies reported that they have a strategy about the further development of brands, 63 percent of companies do see possibilities for expanding their brands to new markets, 70 percent of companies see possibilities for establishing a leading market position with their brands in the future, while only 22.5 percent of companies did not report any of the above activity. Again, regarding preparations for future marketing the clusters of companies according to brand development activities and brand value activities were compared. Again, companies in the cluster with more developed brand development activities reported significantly better preparation for future marketing. Also, regarding brand value clusters, companies in the cluster with more brand values activities reported significantly better preparation for future marketing (see Table 6 for reported mean values and t-test statistics).

T l l l D	· · · ·	<i>с</i> .	1	C		· 11	1 • 🗸
I able 6. Pro	naration to	Nt1111100	marboting	tor com	DAM10C	$m \Delta h$	namar
1 404 0.11	paraion p	minic	mandung	01 6011	panies .	$m \perp m$	mu
	/ /	/	01				

				Std.	Std. Error	t	Sig.
	Cluster	Ν	Mean	Deviation	Mean		(2-tailed)
Brand development	Lower	20	1.45	1.394	.312	3 3 2 5	002
activities	Higher	20	2.65	.812	.182	-3.325	.002
Brand value activities	Lower	25	1.60	1.384	.277	2 0 4 1	001
	Higher	15	2.80	.560	.144	-3.841	.001

* Scale for preparation for future marketing orientation includes brand development strategy, new markets planning or leading market position; range 0-3.

Furthermore, differences in branding and brand capital are analyzed for companies according to their value added. As a measure for company value added, a dummy variable for company value added in a selected year (2009) was taken, where the median value added per employee was taken as a cut-off value (0=below median, 1= median and above). When comparing clusters of companies according to their brand development activities (Table 7), the analysis showed significant negative correlation between the brand development cluster membership and value added per employee (Phi=-.300, sign.=0.05).

 Table 7: Brand development marketing activities and value added per employee for companies in
 Albania

			Value	e added	
			lower	higher	Total
Brand development	lower	Count	7	13	20
activities		% of Total	17.5%	32.5%	50.0%
	higher	Count	13	7	20
		% of Total	32.5%	17.5%	50.0%
Total		Count	20	20	40
		% of Total	50.0%	50.0%	100.0%

A comparison of value added to brand value activities (see Table 8) revealed that two clusters of companies according to their brand value activities do not differ significantly in terms of value added (Phi=-.155, sign.=0.327).

Table 8: Brand value marketing activities and value added per employee for companies in Albania

		Value added			
			lower	higher	Total
Brand value	lower	Count	11	14	25
activities		% of Total	27.5%	35.0%	62.5%
	higher	Count	9	6	15
		% of Total	22.5%	15.0%	37.5%
Total		Count	20	20	40
		% of Total	50.0%	50.0%	100.0%

Similarly, a comparison of marketing innovations and preparation for the future in terms of branding and brand capital reveals that the differences between companies with lower and higher value added are not significant (Table 9).

	Value			Std.	Std. Error	t	Sig.
	added	Ν	Mean	Deviation	Mean		(2-tailed)
Marketing innovations	Lower	21	2.95	1.283	.280	.565	.576
	Higher	19	2.68	1.668	.383		
Preparation for	Lower	21	2.33	1.155	.251	.148	.147
marketing future	Higher	19	1.74	1.367	.314		

Table 9: Marketing innovations and preparation for the future for companies with different value added per employee for companies in Albania

Finally, the effects of industry, company legal form, export orientation, and size on branding and brand capital were examined. Companies were divided into two industry groups: manufacturing and services companies. The analysis did not reveal significant differences in terms of industry structure for companies with lower and higher brand development and brand value activities in terms of their marketing innovations or preparation for future marketing. Although services companies seem to report more marketing innovations (mean for manufacturing = 2.7, for services 3.0), t-test for means does not report significant differences (t=0.59, sig. (2-tailed) =.55).

For company legal form, limited companies were compared to joint stock companies. Differences in legal forms structure were significant for companies with lower and higher brand development (Phi=-.327, sign.=0.038) so that joint stock companies are predominantly companies with high brand development activities, while limited companies are to a higher degree low brand development companies (see table 10). For brand value activities, differences are not significant (Phi=-.282, sign.=0.075). Joint stock companies reported more innovations (3.4, compared to 2.6 for limited companies, t=1.91, sig. (2-tailed) =.06).

		Legal form			
			Ltd	Joint stock	Total
Brand	lower	Count	17	3	20
development		% of Total	42.5%	7.5%	50.0%
activities	higher	Count	11	9	20
		% of Total	27.5%	22.5%	50.0%
Total		Count	28	12	40
		% of Total	70.0%	30.0%	100.0%

Table 10: Brand development marketing activities and company legal form for companies in Albania

For export orientation, companies with revenues from exporting were compared to companies that earned most of their revenues in the domestic market. Again, the differences were not significant for companies with lower and higher brand development and brand value activities. There were no significant differences among exporters and non-exporters in terms of their marketing innovations, nor preparations for future marketing.

Finally, the impact of company size on branding and brand capital was analyzed. Two groups of companies were formed: companies with less than 50 employees and companies with more than 50 employees. The analysis revealed no significant correlations between measures of branding and brand capital and company size.

3 Discussion

Overall, our research synthesizes and tests branding and brand capital for Albania. The following conclusions can be taken from the sample of companies that could serve as examples of practice among companies in Albania. A substantial number of companies report activities related to brand development. This means that they report activities related to the company's own brand, corporate brands in addition to brands of product/services, and in-one third of cases even developed brand

architecture as a system of organizing company brands. However, the majority of companies report only a limited amount of activities related to brand value. They lack either legal protection of their brand or do not finance activities to increase brand value, and have no measurement of brand value as such.

In terms of brand capital, the investments in activities to increase the value of brands were examined. The reported investments are very sound, in the range to 30 percent of sales, which highlights the importance of investments in the field in investigated companies. As already stated, the sample of companies included in the analysis includes selected companies that can serve as role models. Many of the selected companies report marketing innovation and preparation for future marketing. Also, it is not surprising to find that brand management activities in these companies are related to continuous improvement, innovation, and the future orientation of companies.

Nevertheless, the comparison of brand development activities for companies with different value added per employee showed higher value added for companies with lower brand development activities. Differences were not significant for other branding activities. Brand development activities as intangible investments are therefore more significant for companies that, in principle, reach lower value added per employee. This could show that these companies recognize the value of branding and brand related activities, while high value added comes from companies enjoying the privilege of an unsaturated market or market niches. Such privileged companies can reach high value added per employee without trouble and worries over brand development, brand capital, or other forms of intangible capital.

No significant differences in terms of industry structure show that branding activities among companies in the sample can be found between services firms and

manufacturing companies. In terms of legal form, joint stock companies are predominantly companies with high brand development activities, while limited companies are to a higher degree low brand development companies. Joint stock companies are in principle larger companies (in terms of their assets and revenues, not necessary in terms of employees), and therefore employ more resources necessary for brand development. The size of companies in terms of assets in revenues could in general play an important role in terms of sources available for brand management activities. Another comparison according to the number of employees did not reveal significant differences. Labor intensive companies could be among those companies that have little branding activities present (e.g. the herb collecting and processing company Xherdo Ltd., see company case in Redek et al., 2012, in this book), or very successful companies(e.g. the plastic products company Golden Pen Generation (GPG), see company case in Redek et al., 2012, in this book) can be in principle very small in terms of their employees. In terms company size, small and medium-sized enterprises (SMEs) represent the majority of all companies located in Albania and employ more than two-thirds of the overall workforce. SMEs are in principle customer-oriented organizations; however they lack branding related knowledge necessary to transmit value to customers. Loca and Ceku (2010) report that the concept of brand and branding becomes more significant for Albanian SMEs since it helps them compete and increase market share. Although their understanding of branding is often limited to a unique name or logo that is shown on their products, and not necessarily related to the attributes or values of the products/company to customers, they do mostly perceive branding as equal to quality.

Consistent to findings for relational, informational, and IT capital in Albanian firms (Koman and Lalović, 2012, in this book), our results show that export orientation does not play a significant role, as the majority of companies in the sample are

oriented to the domestic market: their products and services (agricultural products, patriotic tourism, herb collection) lack the quality necessary for efficient export. These companies perform "refinement operations" without quality and control standards of branded European producers and, therefore, cannot compete for business from these companies (e.g. Stella, a textile company, or Dafinor, a wood panel production company, see company cases in Redek et al., 2012, in this book). However, some export-oriented companies (e.g. the plastic materials company GPG, see Redek et al., 2012, in this book) managed to gain knowledge about foreign markets and a competitive edge through a foreign trade partner.

In general, companies that started in the 1990s and remained in business had to learn about business practices with intense competition in the market. With more exposure to marketing practices from their partners and elsewhere, they are improving their own marketing and branding activities. However, as shown by Strizhakova et al. (2008), global companies possess the advantages of developed and well-managed brands that can be a serious threat to companies that are used to less intense competition. Companies in Albania will need to deliver to the set standards of branding in order to strengthen their market positions and brand equities.

Last but not least, the success of marketing and branding activities will depend on two sources: the supply of qualified people with relevant marketing skills on how to facilitate brand management, and access to reliable data on market development, market share, and media audience. Branding is not only about spending on advertising and media, but mostly about differentiation and reaching out to the target audience.

References

Anholt S. (2007). Simon Anholt Nation Branding Place. Albania, Karen Hughes. Retrieved March 29th 2012 from http://simonanholt.blogspot.com/2007/11/albania-karenhughes.html.

Bedalli Elona. (2012). Publicitetit 2012: E pëson media e shkruar. Monitor. Biznes. Ekonomi. Treg. Retrieved May 15th 2012 from http://www.monitor.al/publicitetit-2012-e-peson-media-e-shkruar/.

Balkans.com. (2012). Albania is re-branding itself 'a new Mediterranean love'. Retrieved March 29th 2012 from http://www.balkans.com/open-news.php?uniquenumber=91807, 29.3.2012.

Corrado, C., Hulten, C., & Sichel, D. (2009). Intangible capital and US economic growth. Review of Income and Wealth, 55(3), 661-685.

Financial times. (2011). Emerging Destinations. Retrieved March 29th 2012 from http://www.ft.com/cms/s/2/8b194f34-19e1-11e0-b92100144feab49a.html#axzz1qar1UUUs.

Interbrand. (2004). Brand valuation. The financial value of brands. Retrieved March 29th 2012 from http://www.brandchannel.com/papers_review.asp?sp_id=357.

Kapferer, J. N. (2008). The new strategic brand management: creating and sustaining brand equity long term. London: Kogan Page Publishers Ltd..

Keller, K.L. (2010). Brand Equity Management in a Multichannel, Multimedia Retail Environment. *Journal of Interactive Marketing*, 24(2), 58-70.

Keller, K. L. (2003). Building, measuring and managing brand equity. New Jersey: Person education, Inc.

Koman, M., & Lalović, G. (2012). Relational, informational and IT capital. In Prašnikar, J., Redek, T., and Memaj, F. (eds.), *Albania: The Role of Intangible Capital in Future Growth*, (68-82). Ljubljana: Ekonomska fakulteta.

Loca, S., & Ceku, B. (2010). Marketing communication and Branding for SMEs. *Journal of Studies in Economics and Society*, 2(1), 67-76.

Lonely planet. (2011). Top 10 countries for 2011: Europe's got the goods. Retrieved March 29th 2012 from http://www.lonelyplanet.com/albania/travel-tips-and-articles/76175.

Manrai, L. A., Manrai, A. K., & Lascu, D.-N. (2001). A country-cluster analysis of the distribution and promotion infrastructure in Central and Eastern Europe. *International Business Review*, 10(5), 517-549.

McCann Erickson Tirana. (2012). Retrieved April 2nd from http://www.mccann.al/view-gallery.php?pg=14.

Morgan, N. A., & Rego, L. L. (2009). Brand Portfolio Strategy and Firm Performance. *Journal of Marketing*, 73(1), 59-74.

O'Mahony, M., & Vecchi, M. (2009). R&D, knowledge spillovers and company productivity performance. *Research Policy*, 38(1), 35-44.

Petromilli, M., Morrison, D., & Million, M. (2002). Brand architecture: building brand portfolio value. *Strategy & Leadership*, 30 (5), 22-28.

Pici A. & Goga, T. (2007): Albanian Gap. 10th SEMPL conference presentation. Retrieved April 2nd from 10.sempl.si/files/presentations/sempl10-adlej-pici.ppt, 2.4.2012.

Redek, T., Memaj, F., Prašnikar, J., & Trobec, D. (2012). Albania: Two decades of economic development at a glance. In Prašnikar, J., Redek, T. and Memaj, F. (eds.), *Albania: The Role of Intangible Capital in Future Growth*, (4-59). Ljubljana: Ekonomska fakulteta.

Sevrani, K., & Gorica, K. (2011). Implications of E-banking in Entrepreneurial Marketing-Case from Albania. *Chinese Business Review*, 10(1), 67-75.

Sevrani, K., Gorica, K., Tolica, E. K., & Ma, E. Z. (2010). The Internet's Impact on the Marketing of Services: The Case of Electronics Companies in Albania. *Journal of Economic Development, Management, IT, Finance and Marketing, 2 (1),* 150-173.

Strizhakova, Y., Coulter, R. A., & Price, L. L. (2008). The meanings of branded products: A cross-national scale development and meaning assessment. *International Journal of Research in Marketing*, 25(2), 82-93.

Tellis, G. J., Prabhu, J. C., & Chandy, R. K. (2009). Radical Innovation across Nations: The Preeminence of Corporate Culture. *Journal of Marketing*, 73(1), 3-23.

Wang, F., Zhang, X.-P., & Ouyang, M. (2009). Does advertising create sustained firm value? The capitalization of brand intangible. *Journal of the Academy of Marketing Science*, 37(2), 130-143.



THE CASE OF R&D

1 Introduction

The key elements of growth and competitiveness in the EU are knowledge and innovation, both very important cornerstones of intangible capital, and today one of the most researched determinants of productivity. Innovation is often related to R&D and economic growth. R&D generates knowledge that enables firms to develop either superior products or more efficient production processes (Ramirez and Hachiya, 2008).

The purpose of this chapter is to examine the characteristics of R&D and innovation in Albania. The study is based on a detailed survey in a sample of 40 Albanian companies. The research applies new methodology, based on the analysis of intangible assets, which employs cascading techniques on modified standardized international questionnaires adapted for the characteristics of developing countries.

At the moment the situation in Albania is grim, but improving.¹ According to the National Strategy of Science, Technology and Innovation 2009-2015 (Republic of Albania, 2009, p. 11), Albania was estimated to have a gross domestic expenditure on R&D of about 15 million euro in 2009, which represented around 0.2 percent of

¹ The data on research and development in Albania was not collected systematically and in accordance with international standards (OECD, 2002, aka Frascatti manual). The lack of reliable data complicates international comparisons.

⁹⁹

GDP. Also, this is funded primarily by the public sector and foreign sources. According to the World Bank data (2011), 1.3 percent of Albanian manufacturing exports are high-tech exports. This is significantly less than the exports in developing Sub-Saharan Africa countries (5.3 percent). The comparison with the OECD (19 percent), Euro area (16 percent) and developing upper middle-income countries (21.4 percent) is even more revealing. The majority of Albanian exports represent clothing, iron and steel industry products and machinery and transport equipment. The World Bank report 'Building Competitiveness' (2009, pp. 26-37) states that 59 percent of manufacturing firms in Albania reported in 2007 to have introduced a new/improved product. This fact ranked Albania second in the region and behind Croatia (67 percent). Primarily firms in the textiles and garment industry are more likely to introduce new products, while firms in the metal industry are more likely to use upgrading. Foreign firms and exporters are more likely to innovate, and also smaller firms are less likely to innovate. In terms of process innovation, the most common method of introducing new technology is through the purchase of new technology, followed by the hiring of skilled personnel. Of the companies that introduced new technology, 95 percent of them did so by purchasing new machinery, 64 percent by hiring skilled personnel, and 29 percent through internal sources. And given that better technology usually indicates foreign technology, larger companies are more likely to introduce technological upgrading (54 percent of larger firms compared to 27 percent of small firms). Exporters also have better access than non-exporter to new technology (65 percent of exporters licensed new technology compared to 29 percent of non-exporters).

Albania has been improving in terms of innovation and the environment for innovation, and some of that progress and focus was provided from governmental programs (Republic of Albania, 2009 and 2011, various funding programs, institutional support and other). Improvement is evident also from international comparisons, although Albania is still ranked low. The Global Innovation Index (2011) ranks Albania as the 80th economy among 125 economies (121st in 2009), and acknowledges the structural reforms implemented during transition, especially those strengthening the legal system and financial system. The country's economic progress is also important and stimulating in terms of innovation. Namely, growth provides additional stimulus and funding for R&D. But the country has significant weaknesses in many relevant aspects. It has low expenditure on education, poor quality of research institutions and low cooperation of universities and industry, low high tech imports, and poor trade and transport infrastructure.

This chapter is structured as follows. First, the theoretical background to the analysis is provided, followed by research design. Third, the results are presented, followed by the conclusion.

2 Research methodology

Innovation activity in Albania reflects the innovation situation in Albanian companies. The study extends the research and results of the 2009 World Bank study. To further examine and find causes of the rather grim overall situation, a study of the microeconomic characteristics of innovation was conducted. The study was a part of research on the characteristics of intangible assets in Albania. Namely, the research stems from the definition of innovation in terms of intangible capital, and the survey is based on a detailed questionnaire adapted for developing countries.

2.1 Research methodology: theoretical background

Innovation is increasingly important in growth literature today. Although innovation and growth have been linked since (pre)classical economics, R&D and innovation is the focus of endogenous growth theory and intangibles research, showing a clear linkage between innovation, R&D and growth (e.g. Romer, 1990, Aghion and Howitt, 1998, Corrado et al., 2009, Fukao et al., 2007, Van Ark et al., 2009). Increased R&D expenditure will lead to increased innovation activity and thus higher economic rents, to the spillover effect of R&D knowledge across the economy and enhanced absorption capacity for acquiring knowledge from the environment (Griffith et al., 2003). Empirical research confirms the importance of R&D and innovation for growth. According to Klenow and Rodriguez-Clare (1997) or Easterly and Levine (2002), countries with larger R&D as a share of GDP grow faster. The firm level effect has also been confirmed via positive influence of innovation on firm productivity (e.g. Wakelin, 2001, Mairesse and Sassenou, 1991, Griliches and Mairesse, 1983).

Is the question of intangible capital, innovation, and R&D relevant also for developing economies? R&D and innovation have direct and indirect effects on the economy. Innovation that is achieved by investment in intangibles is a major change-driver, as innovation primarily stimulates productivity (Griffith et al., 2003). Similarly as other elements of intangible capital, R&D increases the productivity and value added of firms, thereby increasing their (international) competitiveness, opening their potential for exports, and increasing their potential for future growth. This is the most obvious direct impact on companies. Also important is the impact on other companies (e.g. value chain or competition) through spillover.

The nature of innovation naturally depends on country development. According to Forbes and Wield (2000), innovation and technology management are very important and active also in technological followers and in smaller companies, not just global players. Forbes and Wield (2000) also stress that for innovation success the type of R&D and its focus is most important, not the level of R&D spending. In technological followers the focus is on incremental innovation, also process innovation is often more important than product innovation given that competition is more cost oriented. In this context, shop floor innovations are the major source of savings. Last, it is highly important to recognize also the importance of organizational, cultural and managerial innovation to close the loop of incremental innovation.

The majority of developing countries exploit the export-based model of growth, and consequently growth depends on several sectors. Intangible capital in general, not just innovative intangible capital, can increase productivity in general and consequently also value added. In this context, the absorption capacity mentioned by Griffith (2003) is also relevant. Given that knowledge transfer (growth by imitation and adjustment) is very important for technological followers, absorption capacity determines the potential for such growth and convergence based on this model. According to the World Bank (2009), technological development is driven primarily by two factors: exposure to more advanced (foreign) technical knowledge and the capacity (and incentive) to absorb it. Exposure to more advanced technologies happens primarily through foreign trade, FDI and brain circulation. On the other hand, adoption and actually benefiting from it depend on a wider environment, including taxation, regulation, competition, government policies, financial resources to adopt and adapt it, having the necessary skills, and other factors.

Last, it should be acknowledged that despite the fact that growth in developing countries is normally driven by just several sectors, other sectors should not be neglected during the favorable growth cycle. Therefore, innovation is important not just in the tradable sector, but also the non-tradable sector (e.g. Prašnikar et al. 2003, Bole, 2009, Chinn and Johnston, 1999). Balance is crucial for stable and balanced future growth.

2.2 Research design

The questionnaire was developed by using the aforementioned theoretical background and existing standardized questionnaires (CIS by OECD/Eurostat). Also, the questionnaire was adapted to incorporate the characteristics relevant for developing countries (Prašnikar et al., 2011).

Innovation and R&D were examined through a series of statements comprising 10 questions. First, the companies were asked in which domestic and foreign markets they operate, then they were asked about the intensity and success of their innovation in comparison with the competitors in their relevant (most important) market. Product and process innovation were examined later, followed by the organization of the R&D department and the size of R&D expenditure. Technological, marketing and complementary competences as a precondition for successful innovation were examined next, and finally we focused on the source of ideas for innovation, where the sources could be either internal or external. The structure of the questions is revealed in the results section. Thus, a detailed description is omitted here.
3 Results

The analysis of innovation activity as part of intangible capital in Albania focused on the following aspects: product and process innovation, characteristics of R&D in the company, company competences, and capabilities. Also, company characteristics and R&D activity links are examined.

3.1 Overview of R&D activity in the sample companies

Innovation is the essential driver of growth as has been noted by many (Griffith et al., 2000, Rodríguez-Pose, 1999, Baumol, 2000, Romer, 1990 and other). Schumpeter (1942, pp. 82-83) well captured the importance of innovation for growth: "The fundamental impulse that sets and keeps the capitalist engine in motion comes from the new consumers' goods, the new methods of production or transportation, the new markets, ... This process incessantly revolutionizes the economic structure within, incessantly destroying the old one, incessantly creating a new one. This process of creative destruction is the essential fact about capitalism.'

The innovation intensity is driven by many elements, internal and external. One of the most important external elements is the quality and the characteristics of the competition. According to Schumpeter (1942) it is the new products, processes, and innovation themselves that create the competition. The innovation is an everpresent threat. The fiercer the competition in the firm's target market, the more driven the firm is to innovate in order not to lose its market share² (see also Aghion et al., 2002, Blundell et al., 1999, Gilbert, 2006). Also, company innovativeness and

 $^{^2}$ Schumpeter (1942) believed that the standard market structure approach to competition is less important. But nonetheless, he preferred some degree of monopoly. He never made it clear whether innovation leads to monopoly or the opposite, but the literature normally assumes monopoly to be a reward for innovation.

technological advancement depends crucially on the firm's presence in international trade (e.g. Bloom et al., 2009, De Loecker, 2007, Aw et al., 2009, Damijan and Kostevc, 2006).

The companies were therefore first asked in which markets they are present, what their most important target market is, and what percentages of sales they sell in a specific market (Table 2). Thirty-two companies (80 percent) claim the domestic market (including those that listed regional) to be their most important market, 7 companies even listed the local market as their most important market (17.5 percent). On average, the firms that operate in the local and regional markets also sell the vast majority of their products in that market, 8 companies sold their entire produce locally. On average companies sell over 85 percent of products in Albania. The European markets were the most important export regions for the sample with 9.7 percent.

A company's largest, most important market was the focus of our further analysis (referred to as relevant market). The companies were asked to evaluate their position in comparison with other companies in only their relevant market³. For the majority of companies, their relevant market was the local and domestic market. Only two companies explicitly stated that their most important market is abroad: the first company stated the Balkan market and the second company specified the American market.

³ It must be noted that when companies were evaluating their performance, they evaluated themselves only in terms of their relevenat market. To further elaborate: if a company sells primarily in the domestic market, it only compared itself with other domestic companies. In this context also 'globally new products' must be explained or understood as new (only, although not neccessarily) in the national market.

¹⁰⁶

Table 2: Target market*

	Please, mark in which of	
	the following markets did	
	your company sell	Average sales in a
	products/services in	specific market
Target market	2009? (% of companies)	(%)
Local/regional market in Albania?	85.0	58.2**
National market?	60.0	86.3
Other European markets (excluding		
countries of Western Balkan)?	27.5	7.5
Western Balkan markets?	37.5	5.5
Other markets?	5.0	0.75

* Data differ slightly from data in the chapter on relationship and informational capital, due to a slightly different division of markets.

** If companies did not provide data on regional sale, but provided only data on national sales, the information on regional sale was treated as a missing value in calculating the regional sales average. There is no mechanism available to break down national sales. In total, data on regional sales was missing for 15 companies.

One of the first innovation drivers is the quality of competition. Given that the majority of sales are conducted in the domestic market, the nature of competition is largely determined by the characteristics of domestic (even local) competition. Sixty percent of companies feel to be on average at least as successful in introducing new products as their competitors in the past 5 years, one-third feel more successful, while even 27.5 percent (11 companies) feel that they are of the leading companies in the industry. Except for one company, all other companies (10 companies) reported their most important market to be the domestic or even regional market (3 companies) (Table 3).

Table 3: Introducing new products: comparison with competitors (percent of companies)

Introducing new products and competitors	NO	YES	
We were as successful as our competitors were on average in introducing new products in the last five years	40	60	
We were more successful than our competitors were on	67.5	32.5	
We were one of the leading companies in the industry in	72.5	27.5	
introducing new product in the last five years.	72.5	27.5	

Innovation can either denote product or process innovation. Product innovation is defined as an introduction of new goods or services or an introduction of significantly improved goods or services. Process innovation, on the other hand, consists of the implementation of a new or significantly improved production process, distribution method, or support activity for the goods or services (OECD, 2002). Innovation is based on the results of new technological development, new combinations of existing technologies, or the use of other knowledge by the enterprise (OECD, 2002).

Figure 1: Answers to the question 'Please, mark the relevance of the following types of new products in your company.' *+, in percent



Answers on a scale of 0-3: 0 denotes 'not used', while 1-3 denotes low (1), medium (2), and high (3) relevance.

+ Global market should be explained with care. Companies were asked to evaluate themselves only in terms of their relevant market. For some companies it is global, but for the majority it is the national market.

The most important type of product innovation is improving existing products (Figure 1). Sixty percent of companies claim this to be very important for them.

This is followed by extensions to existing product lines and new product lines. But even over 20 percent of companies feel that introducing new products that are novelties also globally is very important for them.⁴

The companies were very active in process innovation (Figure 2). Over 67 percent introduced significant process innovation, over 60 percent significantly improved production, over 70 percent significantly improved support services like maintenance, IT, sales, and similar, and 77.5 percent improved logistics, delivery, distribution of inputs, and outputs.





Characteristics of innovation activity and the outcomes or success of innovation depends primarily on the quality and organization of the R&D department, which is closely related to funding and human capital. We first examine the problems of R&D department existence and R&D activity funding.

⁴ See footnote 2.

¹⁰⁹

Almost half of the companies (45 percent) spent less that 1 percent of revenue on R&D. Fifty-five percent spent at least 1 percent of revenue or more. Close to one-third of companies spent 2 percent or more, while 17.5 percent spent 3 percent or more (Figure 3).



Figure 3: R&D expenditure, in percent of companies

Only 20 percent of companies reported to have an R&D department: that is only 8 out of the sample of 40 companies had an R&D department (Figure 4). In the companies where it does exist, the R&D department is well organized and in the majority of companies it works in an absorption and development capacity.



Figure 4: Organization of Rer D department*, in percent of companies



* If a company reported not to have an R&D department, then it did not answer other subquestions. Therefore, only the companies that have R&D departments provided answers about the characteristics of their R&D departments.

Companies can get information and ideas for innovation from many different sources. These can be divided into internal sources (e.g. suggestions from employees, own research activities, etc.) and external sources (market, institutional, and other sources). Albanian companies obtain the majority of ideas from inside the company (Table 4). Namely, over 67 percent of companies reported internal sources as highly important. It is also important to note that conferences, market fairs and exhibits are a highly important source of information to over 50 percent of the sample, more than competition and other companies in the field (43.6 percent) and suppliers of equipment (42.5 percent). This indicates the increasing importance also of external sources of information and the value chain.



Table 4: Sources of innovation ideas, percent of companies

Mark the relevant sources of information, evaluate		Not used	Low	Medium	High	Mode
importance	importance		1	2	3	
Internal sources	Inside the company	2.5	2.5	27.5	67.5	3
Market sources	Suppliers of equipment	10.0	15.0	32.5	42.5	3
	Suppliers of materials,	12.5	2.5	47.5	37.5	2
	components, program					
	equipment					
	Buyers	5.0	7.5	55.0	32.5	2
	Competitors and other	5.1	15.4	35.9	43.6	3
	companies in the field *					
	Consultants, private	17.5	17.5	27.5	35.0	3
	research or R&D					
	facilities *					
Institutional	Universities or other	25.0	30.0	20.0	25.0	1
sources	higher education					
	institutions					
	Government or public	27.5	32.5	17.5	22.5	1
	research institutions					
Other	Conferences, market	15.0	5.0	27.5	52.5	3
	fairs, exhibits					
	Scientific, commercial	22.5	30.0	27.5	20.0	1
	and technical journals					
	Industrial associations	12.5	12.5	50.0	25.0	2
	and chambers					

* Percentages do not add up to 100 percent, but to 97.5, given that in both cases 1 case is missing.

Competences are crucial for successful innovation. Grant (1991) defines competences as the ability to utilize resources that spread across multiple functions, products and markets in a sustainable and synchronized manner. They are broader and not strictly industry specific. Competences are also defined as a set of related abilities, commitments, knowledge, and skills that enable a person (or an organization) to act effectively in a job or situation (BusinessDictionary.com). One could also say that competences mean that you generally have the knowledge to do something, but do not know whether you could perform a task (merge the information together correctly), while capability means that you actually can perform a specific task. Grant (1991) defined capabilities also as repeatable patterns

of actions in the use of assets to create, produce and/or offer products to a market. Simplified, one could say that competences mean that one possesses the knowledge, but capabilities mean that the knowledge can be actually put together and used in order o create something.

Technological competences										
	1	2	3	4	5	Avg	Mode			
Research and development in the firm is advanced.	42.5	2.5	32.5	7.5	15.0	2.50	1			
Number of available technological capabilities inside the firm or through strategic partnership is quite large.	37.5	5.0	27.5	12.5	17.5	2.68	1			
We are good at predicting technological trends.	42.5	12.5	12.5	17.5	15.0	2.50	1			
Marketi	ing cor	npeten	ices							
Obtaining information about changes of customer preferences and needs	25.0	2.5	12.5	22.5	37.5	3.45	5			
Acquiring real time information about competitors	17.5	5.0	22.5	22.5	32.5	3.48	5			
Establishing and managing long-term customer relations	25.0	5.0	12.5	15.0	42.5	3.45	5			
Establishing and managing long-term relations with suppliers	27.5	2.5	10.0	12.5	47.5	3.50	5			
Compleme	entary of	compe	tences							
Activities of the business units are clearly defined in the corporate strategy of our firm.	25.0	10.0	17.5	15.0	32.5	3.20	5			
Good transfer of technological and marketing knowledge among businesses units.	27.5	10.0	15.0	17.5	30.0	3.13	5			
The intensity, quality and extent of research and development knowledge transfer in co-operation with strategic partners.	22.5	10.0	15.0	22.5	30.0	3.28	5			
Product development is cost efficient.	30.0	2.5	17.5	20.0	30.0	3.18	1 and 5			

Table 5: Firm competences (Answers to ' Evaluate performance of your company compared to your main competitors in the following aspects on a scale from 1 to 5')*, percent of companies

* 1=considerably worse than the main competitors, 2=worse than the main competitors, 3=same as main competitors, 4=better than the main competitors, 5=considerably better than the main competitors

Results reveal that the sample of Albanian companies acknowledges the lack of technological competences, but in terms of marketing and complementary competences, the sample is divided (Table 5). Quite a large percentage (30 or even more) believes to be considerably better than their competitors, but over 20 percent of the sample or more believes to be considerably worse than their main competitors. In terms of future development the lack of technological competences, which can in part explain the nature of product and process innovation, will impede progress. Also, the duality in the quality of the companies could (depending on industry) be an obstacle to creating strong value chains and clusters. Competences and capabilities depend largely on human capital, which is according to the World Bank (2010) one of the major problems of Albanian development.

3.2 R&D and company characteristics

The sample can be broken down by legal type (limited liability and joint stock), sector (construction, manufacturing and services (trade and other)), size, export orientation, and by whether the company operates in the final market or is a business-to-business type of company.

We first focus on the introduction of new products and R&D expenditure related to company characteristics. Interestingly, the results show that the share of export⁵ was not related to the importance of the introduction of new products. Even more, out of the 3 companies that sold more than 50 percent abroad (10 percent of the sample), only 1 claimed to be on par with competition, and only 1 claimed to be among the leading companies in introducing new products in their relevant market.

⁵ Companies were asked to state whether they earn more than 0, 25 and 50% of sales abroad. It is an example of a cascading question, values 1, 2, 3 and 4 were assigned to: not more than 0, more than 0, more than 25 and more than 50, respectively.

¹¹⁴

Those that exported between 25 and 50 percent of sales (4 companies) were similarly unconfident: 3 reported to be on par and 1 to be one of the leading companies. Trade was also not related to R&D expenditure. How could this controversy between trade and innovation be explained? According to Memaj (2012), Albania is a small country and not well known in other countries for innovation and new products. The exporters are trying to be in line with the needs of their clients, and only in few cases are they in a position to introduce new products in the export markets. Albanian companies are usually exporting traditional goods and this is one of the features of Albanian export. Many exporting companies also export due to their sub-contracting deals (Koman and Lalović, 2012, in this book), where value added and productivity are both low (see also the case studies of Stella and Dafinor in Redek et al., 2012, in this book). But in the future the economy will need to shift focus, similarly as some companies are already doing (see the case study of Xherdo in Redek et al., 2012, in this book).

Although overall, companies with higher share of trade do not implement product innovation statistically significantly more often (importance of innovation, see above), and there are differences by specific types of product innovation. Companies that trade were more persuaded that it was more important (evaluate on a scale 1-3) to improve existing products or to extend existing product lines (significances were 0.028 and 0.043) in comparison with those that trade less or not at all. It is also indicated that the relationship with trade could exist also with process innovation⁶. Companies with higher share of trade were more inclined towards having more types of process innovation (production, logistics, etc., and support services). The results are insignificant, but the significance levels are much higher (around 0.3 in cases of logistics, delivery and distribution and support

⁶ Otherwise, the introduction of process innovation could not be systematically linked to any of the other grouping variables.

¹¹⁵

services) than in case of other questions conditional on the share of trade (where significances can be as low as 0.9).

The introduction of new products in comparison with competition is also not related to whether the company operates in the final market or is a B2B company; similar is true also for the size of R&D expenditure. Also, the sector (construction, manufacturing, services other and trade) is not in any way related to the introduction of new products or R&D expenditure. Companies were also divided into two groups by productivity in 2008 and 2009. Both median and average productivity were used to divide the companies. The results indicate that neither R&D expenditure nor the success in introducing new products in comparison with competition had any systematic significant relationship with productivity. According to Memaj (2012) the expenditures in R&D and success of new products have no significant relationship with productivity even because the companies are newly created (have no historical background) and the R&D expenditures are not very high. The fact that there are a few companies (and not highly established in the markets) competing with each other in the same markets (trading similar products) brings the focus of the companies more to competition than to R&D and new products.

The types of new products in relation to different groups were also examined. Repositioning, improving existing products, extensions to existing product lines, new product lines and 'globally'⁷ new products could be selected and their importance evaluated on a scale of 1 to 3 (0 if not relevant). Besides the results conditional on the importance of trade (see above), business-to-business type companies also rate extensions to new product lines as significantly more important

⁷ Again, note that 'globally' refers to company relevant market. Consequently, for the majority of firms,this is the domestic market, but for a few also the foreign market.

¹¹⁶

than the rest. The service sector also claimed new product lines to be significantly more important than the manufacturing sector did (sig. 0.063). There is also a weak inclination towards B2B companies claiming 'globally⁸ new products' to be more important than other companies (final market) (sig. 0.121).

Competences and technological marketing, as well as complementary, are very important in determining R&D potential (companies were asked to evaluate their agreement with the statements on a scale of 1 to 5). Companies that do not trade statistically significantly (0.01) report in a lesser extent that the number of available technological capabilities inside their firm or through strategic partnerships is quite large and also evaluate their ability to predict technological trends lower (0.07). This speaks in favor of the hypothesis that trade and innovation and technological development are positively related. Data also shows that there is a slight inclination of companies in the services sector claiming to have advanced R&D in their firms in comparison to firms in the manufacturing sector (sig. 0.112). Similarly, there is a significant difference between companies that were above average productive in 2008 and those below average. The companies that were above the productivity median in 2008 evaluated the claim to 'higher number of available technological capabilities inside strategic partnership and within the firm' higher than those below the median (sig 0.047). Similarly, also median productivity in 2008 gives a signal of difference, but it is still insignificant (sig. 0.153).

In marketing competences, B2B companies are more inclined towards regarding themselves as capable of establishing and managing long-term relationships with suppliers (0.091). Also, companies that do not trade evaluate their ability in this same context (long-term relationships) lower with both suppliers (0.144) and customers (0.077). In complementary competences, the companies that do not

⁸ Similarly as footnote 6.

¹¹⁷

trade evaluate their 'intensity, quality and extent of research and development knowledge transfer in co-operation with strategic partners' lower than those that trade (0.08).

Overall, the results indicate that innovation characteristics differ among companies, and the market (export-oriented companies seem to be more active) and the type of company in the value chain (B2B seem to be more innovative) determine the differences. Also, there is an indication that innovation is linked to productivity since companies more than average or median productive were in some cases (weakly) more active in R&D.

Although the results are in many cases insignificant, they do nonetheless indicate the existence of the expected relationship.

4 Conclusion

Knowledge and innovation are important also in the development of developing countries like Albania. But the benefits should not be expected to bring the desired results unless sufficient focus in provided in both companies and in the business environment in general, also in terms of policy measures.

According to the World Bank (2011), the Albanian growth strategy is to become an upper-middle-income country. Economic growth and productivity rise until now resulted primarily from inter-sectoral shifts of labor, but in the future this process will become largely dependent on intra-industry changes. And the literature suggests that innovation is an important source of productivity growth.

This chapter focused on the characteristics of R&D activity in a sample of 40 Albanian firms selected from the best companies. The results show that at the

moment, innovation and R&D do not yet carry the weight they could. Overall innovation expenditure in Albania is very low, only 0.2 percent of GDP and funded primarily by the public sector and foreign funds. The result is a reflection of the Albanian economic structure, with the majority of exports representing low value added products like textiles, leather products, and ore.

Sample results confirm that R&D and innovation in Albania at the moment are still not at satisfactory levels. The majority of responding companies chose the domestic market as their prime market. The domestic market is still offering significant growth (rent) opportunities with rather insufficient competition. The exporting companies are involved primarily in low value added sub-contracting jobs. These two factors could also explain many of the characteristics of innovative activities in Albania. Companies did introduce product and process innovation, but the majority of companies reported improving existing products, extension to existing product lines and new product lines to be their most important types of product innovation. Almost half of companies spent less than 1 percent of revenue on R&D and only 20 percent of the sample companies had an R&D department. On the other hand, the companies were quite active in process innovation, as two-thirds introduced some process innovation. Research also shows that companies lack competences, primarily technological, while the situation is slightly better in terms of marketing and complementary competences.

Currently, Albanian companies are oriented primarily towards the domestic market. But in order for the companies as well as the economy as a whole to become more export oriented and succeed in the international competition, intangible capital, including innovative property, is crucial. As case studies in the first part show, this fact is already successfully exploited by some companies, and is growing successfully at home and expanding abroad.

References

Aghion, P., & Howitt, P. (1998). Endogenous Growth Theory. Cambridge, MA: MIT Press.

Aghion, P., Carlin, W., & Schaffer, M. (2002). *Competition, Innovation and Growth in Transition: Exploring the Interactions between Policies*. William Davidson Working Paper No.151. Michigan: University of Michigan.

Aw, B.Y., Roberts, M.J., & Xu, D.Y. (2009). R&D Investment, Exporting, and Productivity Dynamics. NBER Working Paper no. 14670. Cambridge, MA: National Bureau of Economic Research.

Baumol, W. F. (2000). The Free Market Innovation Machine: Analyzing the Growth Miracle of Capitalism. Princeton: Princeton University Press.

Bloom, N., Draca, M. & Van Reenen, J. (2009). *Trade Induced Technical Change? The Impact of Chinese Imports on Innovation and Information Technology*. NBER Working Paper No. 16717. Cambridge, MA: National Bureau of Economic Research. Retrieved on 5th January from http://www.nber.org/papers/w16717.

Blundell, R., Griffith, R., & van Reenen, J. (1999). Market Share, Market Value and Innovation in a Panel of British Manufacturing Firms. *Review of Economic Studies*, 6 (3), 529-554.

Bole, V. (2009). Crisis Fiscal Policy in small open economy; the case of Slovenia. Bančni Vestnik, 58(11), 57-65.

BusinessDictionary.com. (2012). Retrieved March 10th 2011 from http://www.businessdictionary.com/definition/competence.html.

Chinn, M., & Johnston, L. (1999). The Impact of Productivity Differentials on real Exchange Rates: Beyond the Balassa-Samuelson Framework. Santa Cruz: University of California.

Corrado, C., Hulten, C., & Sichel, D. (2009). Intangible Capital and U.S. Economic Growth. Review of Income and Wealth, 55, 661-685.

Damijan, J., & Kostevc, Č. (2006). Learning-by-Exporting: Continuous Productivity Improvements or Capacity Utilization Effects? Evidence from Slovenian Firms. *Review of World Economics*, 142 (3), 599-614.

De Loecker, J. (2007). Do Exports Generate Higher Productivity? Evidence from Slovenia. J. Int. Econ. 73 (September), 69-98.

Easterly, W., & Levine, R. (2002). It's Not Factor Accumulation: Stylized Facts and Growth Models. In Loayza, N., Soto, R. & Schmidt-Hebbel, K. (eds) (ed.), *Economic Growth: Sources, Trends, and Cycles*, (61-114), Central Bank of Chile.

Forbes, N., Wield, D. (2000). Managing R&D in technology-followers. Research Policy, 29, 1095-1109.

Fukao, K., Hamagata, S., Miyagawa, T., & Tonogi, K. (2007). Intangible Investment in Japan: Measurement and Contribution to Economic Growth. RIETI Discussion Paper, 07-E-034.

Gilbert, R. (2006). Looking for Mr. Schumpeter: Where Are We in the Competition-Innovation Debate? In Jaffe, A.B., Lerner, J. and Stern, S. (eds.) *Innovation Policy and the Economy*, Cambridge (Mas.): MIT Press. Retrieved January 15th from http://www.nber.org/chapters/c0208.pdf.

Global innovation index. 2011. Retrieved January 24th from http://www.globalinnovationindex.org/gii/main/analysis/rankings.cfm

Grant, R.M. (1991). The resource-based theory of competitive advantage: implications for strategy formulation. *California Management Review, Spring*, 114-35.

Griffith, R, Redding, S., & Van Reenen, J. (2000). *Mapping the two faces of R&D: Productivity growth in a panel of OECD industries*. Centre for Economic Policy Research Discussion Paper, 2457.

Griffith, R., Redding, S., & Van Reenen, J. (2003). R&D and Absorptive Capacity: Theory and Empirical Evidence. *Scandinavian Journal of Economics*, 105(1), 99-118.

Griliches, Z., & Mairesse, J. (1983). Comparing productivity growth: An exploration of French and U.S. industrial and firm data. *European Economic Review, 21(1-2)*, pages 89-119.

Koman, M., & Lalović, G. (2012). Relational, informational and IT capital. In Prašnikar, J., Redek, T. and Memaj, F. (eds.) *Albania: The Role of Intangible Capital in Future Growth*, (73-88). Ljubljana: Ekonomska fakulteta.

Mairesse, J., & Sassenou, M. (1991). R&D Productivity: A Survey of Econometric Studies at the Firm Level. NBER Working Papers 3666. Cambridge, MA: National Bureau of Economic Research.

Memaj, F. (2012). Interview with F. Memaj. Held at Faculty of Economics, University of Ljubljana, 20th March, 2012.

Klenow, P. J., & Rodriguez-Clare, A. (1997). Economic growth: A review essay. *Journal of Monetary Economics*, 40(3), 597-617.

Redek, T., Memaj, F., Prašnikar, J., & Trobec, D. (2012). Albania: Two decades of economic development at a glance. In Prašnikar, J., Redek, T. and Memaj, F. (eds.), *Albania: The Role of Intangible Capital in Future Growth*, (4-59). Ljubljana: Ekonomska fakulteta.

Republic of Albania. (2009). National Strategy of Science, Technology and Innovation 2009-2015. Republic of Albania, Council of Ministers. Retrieved December 10th from http://unesdoc.unesco.org/images/0018/001871/187164e.pdf.

OECD. (2002). Frascati Manual, Proposed Standard Practice for Surveys on Research and Experimental Development. Paris: Organization for Economic Cooperation and Development.

Prašnikar, J., Redek, T., & Drenkovska, M. (2011). *Methodological challenges of measuring intangible capital in developing countries: the case of innovation activity*. Paper presented at the 2011 International conference for entrepreneurship, innovation and regional development, Skopje, 5-7 May 2011.

Ramirez, P., & Hachiya, T. (2008). Assessing the value of intangible assets: evidence from Japanese firms. *International Journal of Applied Management Science*, 1 (1), 55-74, 2008.

Rodríguez-Pose, A. (1999). Innovation prone and innovation-averse societies. Economic performance in Europe. *Growth and Change 30, 1, 75 – 105*.

Romer, P. M. (1990). Endogenous Technological Change. Journal of Political Economy, 98(5), 71-102.

Schumpeter, J. A. (1942). Capitalism, Socialism and Democracy. New York: Harper and Brothers.

Republic of Albania. (2011). Strategic programme for the development of innovation and technology of SMEs for the period 2011-2016. Republic of Albania, Ministry of economy, trade and energy. Retrieved December 10th from http://www.mete.gov.al/doc/20110218092915_english_draft_programi_strategjik_dt._10.0 1.2011.pdf.

VanArk B., Hao, J.X., Corrado, C., & Hulten, C. (2009). Measuring intangible capital and its contribution to economic growth in Europe. European Investment Bank Papers 14-1.

Wakelin, K. (2001). Productivity growth and R&D expenditure in UK manufacturing firms. *Research Policy*, *30*(7), 1079-1090.

World Bank. (2009). Albania. Building competitiveness in Albania. Retrieved December 18th from

http://apps.ubmasia.com/files/mediaobjects/Files/25/PSF/09%20WB%20Albania%20Bu ilding%20Competitiveness.pdf.

World Bank. (2010). Albania: The New Growth Agenda. A Country Economic Memorandum. Retrieved December 18th from http://wwwwds.worldbank.org/external/default/WDSContentServer/WDSP/IB/2010/11 /21/000333037_20101121235513/Rendered/PDF/535990ESW0gray1e0only1910BOX353 785B.pdf.

World Bank. (2011). *Albania at a glance*. Retrieved December 18th from http://devdata.worldbank.org/AAG/alb_aag.pdf.

SOCIAL CAPITAL AND CORPORATE GOVERNANCE

1 Introduction

In this chapter we analyze social capital as a part of intangible capital in firms. More precisely, we study the relationship between internal cohesion among the main stakeholders in firms and connect it with firm productivity. The main stakeholders are workers, owners, and managers, where managers play the role of intermediaries between the interests of workers and owners. Each group of stakeholders has a different role within the firm, and also has different goals and expectations. Merely having goals though is not enough; a group must also have the power to achieve its goals. As the common goals of groups are not disputable, the challenge is rather in finding a joint solution when these goals diverge. Each group tries to achieve its own goals over other groups, but to what extent a group is able to achieve its goals is up to that group's bargaining power. By understanding the goals and bargaining power of the three main interest groups, we can further understand firm behavior.

This chapter consists of three sections. First we elaborate on social capital and how to measure it. In the next section we interpret the results of our analysis. The last section concludes.

2 Measuring social capital

A cooperative game of internal bargaining (Aoki, 1984, 2010) allows workers, managers, and owners to determine firm goals together and, therefore, achieve higher productivity, efficiency, and organizational rent than would be achieved in a competitive market. Cooperation can only be achieved through informing the other stakeholders about own interests. Internal cohesion can be understood as a basic principle of social capital theory. Social capital is defined by norms, trust, and networks that help achieve higher firm efficiency (Putnam, 1993). If workers want to actively participate in the bargaining game, they must also be prepared to accept a part of business risk (Williamson, 1975, Ricketts, 2002). The main hypothesis of this chapter is that building a cooperative game between the main constituencies of a firm (building social capital) is positively correlated to firm productivity.

The questionnaire for studying social capital in Albanian firms consists of eight question sets. Forty firms were selected from different industries, sizes, export orientation, and customer orientation (B2B vs. B2C) in order to generate a representative sample. We were able to get responses from the top managers in selected firms.

The first question set (*the decision making*) focuses on the fundamental division between owners (control rights, residual rights) and managers (decision rights). Strategic function is usually in the hands of top management, and everyday operational function is given to middle and lower management levels. The choice for their separation is in the realm of corporate owners (Wheelen and Hunger, 2010). They are also indirectly responsible for the consolidation of owners' and managers' interests and influence cooperative behavior through building trust between main stakeholders of the firm: owners, managers and workers (Aoki, 1984, Essen et al., 2012). Firms must strive to constantly formulate strategies and organizational goals, and position products and services in the market. These have to be jointly discussed and spread through the firm (Holt, 1992, Flamholtz and Randle, 2000).

In the second question set, dynamic firm behavior was taken into consideration by constructing questions pertaining to *labor adjustment*. Firms are able to restructure employment defensively in the short term (employing through agencies, hiring part-time workers, hiring students, using overtime, etc., or strategically in the long term (adjusting the number of full-time employees). Firms that base their competitive advantages on human capital develop core employees employment relationship (Lepak, Takeouchi and Snell, 2003, Aoki, 2010, Zupan et al., 2010).

An important element of bargaining is also *determining wages*, which is the topic of the third question set. Albania has a very liberal type of labor market. Wage size is associated with collective bargaining processes only in rare cases when collective agreements exist and are also enforced. An answer of "No" to the first question in the question set on wages leads to the conclusion that workers are paid at the reservation wage.¹ Furthermore, answering "Yes" to the second question implies a deviation from the earnings assured by collective agreements. This could be the result of either the higher bargaining power of unions, or of firms building their compensation policies on the efficiency wage philosophy. If wages in the firm are

¹ The reservation wage is defined as the second best alternative under which workers would be willing to work.

¹²⁶

among the highest in the country, then either the first or second strategy was escalated to achieve those wages.

In order to achieve higher bargaining power, workers can choose to concentrate their efforts in the form of *unions*. How unions influence firm productivity is an ongoing debate. This question set tested the cooperative behavior of unions. The first question inquires about the existence of unions in the firm, the second about the number of unions organized in the firm, and the third about the unions' concern for the firm's success.² If several unions exist, collaboration among them may be lowered due to competition for membership (Ferner and Hyman, 1998).

There is a lot of literature on the contribution of *workers' participation* to firm success; literature that both neglects their contribution to firm performance (property rights theory, agency theory, transaction cost theory) and supports their contribution to firm performance (better exchange of information between employees and employers, reduced monitoring costs, improved efficiency of resource allocation) (Allen and Gale, 2002), and literature that fosters a culture of consensus and cooperation (Freeman and Lazear, 1995, Aoki, 2010). If workers want to play a role in a firm's decision-making process, they must accept a part of the business risk (Williamson, 1982).

² After the fall of the socialist regime in Albania in 1991, two bigger unions have been formed: Konfederata e Sindikatave të Shqipërisë (KSSH) (*Albanian Trade union Confederation*), as the successor of the Communist Professional Union, founded on 5 June 1991, which is politically more affiliated with the Socialist party, and Bashkimi i Sindikatave të Pavarura të Shqipërisë (BSPSH) (*The Union of Albanian Independent Syndicates*), which was initially founded as a political opposite to the communist regime on 11 March 1991, and is therefore more affiliated with the Democratic Party. Besides these two main unions, there is another smaller union: Federata Sindikale e Tregtisë, Bankave dhe Shërbimeve (FSTBSH) (*The Union Federation of Trade, Banks and Services*).

The fifth question set, *workers' risk aversion*, deals with the employees' inclination toward corporate risk sharing. First, we ask if most of the workers are prepared to do "something more" for the firm. "Something more" is a broad concept that entails deeds and actions by employees that they choose to do willingly, not forcefully (physically or psychologically), for the benefit of the firm. Second, we pose the question, "Do you believe most workers would stay with the firm even if they were offered better employment somewhere else?" This signals the workers' loyalty to the firm. The third question relates to workers' inclination toward accepting a part of the financial risk.

The sixth question set tests *worker participation in decision making*. Bernstein (1982) distinguishes between four degrees of workers' control: 1) employee consultation, 2) employee co-influence, 3) co-determination, and 4) self-management. Since legal acts in Albania do not support workers' participation, questions focus on the first two forms. Employee consultation is examined by asking if workers are informed about key decisions for the firm. Employee co-influence is expressed in the second question by asking if there is an established open dialog with the workers about key decisions for the firm. The third question is informative and inquires about workers' membership in governing bodies.

Lucas (1988) described investment in human capital as a driving force of the endogenous growth theory. It was later recognized as an important source of competitive advantage (Barney, 1991). Firm-specific human and structural resources were later identified as the largest subpart of a firms' intangible investment (Corrado et al., 2009, for the US and the UK; Fukao et al., 2009, for Japan). Bloom and Van Reenen (2010) derived similar conclusions. *Internal training* is thus the topic of the seventh question set. With the first question we can identify a company's collaborative efforts, through the second question we can learn about

the share of employee involvement in training, and with the third question we determine the level of complexity of a firm's measurement system for training effectiveness.

The eighth question set deals with firm-specific human resource practices as well. More precisely, it focuses on *on-the-job training*, where its presence is identified by the first question. The second question reveals the level of a firm's involvement in spreading knowledge within the firm. The third question shows the readiness of a firm to replace key employees quickly, if needed.

3 Results

In the previous section we explained the theory behind every question set and questions in the question sets to show how we measure social capital in the firm. Here follows the interpretation of results from analyzing 40 firms that participated in our study. Results are presented in Table 1, where we can observe the share of positive answers and standard deviations for the total sample, higher productive firms, and lower productive firms (by median value) by individual question. We tested the differences between higher and lower productive firms with Chi square and corresponding P-value.

*	Total			1	More productive			Less productiv	e		
	Ν	% of firms	sd	Ν	% of firms	sd	Ν	% of firms	sd	Chi ²	P-value
1. DESISION-MAKING											
operation/strategic management separation	40	67.5	47.4	- 20	50.0	51.3	20	85.0	36.6	5.584	0.018
managers and owners act unanimously	40	57.5	50.1	20	40.0	50.3	20	75.0	44.4	5.013	0.025
owners, managers and workers coord.	40	50.0	50.6	20	35.0	48.9	20	65.0	48.9	3.600	0.058
2. ADJUSTING EMPLOYMENT											
short-term adjust. to shocks	40	87.5	33.5	20	95.0	22.4	20	80.0	41.0	2.057	0.151
achieving target level of employment	40	75.0	43.9	20	85.0	36.6	20	65.0	48.9	2.133	0.144
core group of employees	40	72.5	45.2	- 20	80.0	41.0	20	65.0	48.9	1.129	0.288
3. DETERMINING WAGES											
higher than alternative wages	40	77.5	42.3	20	85.0	36.6	20	70.0	47.0	1.290	0.256
wages higher than collective agreement	40	50.0	50.6	- 20	50.0	51.3	20	50.0	51.3	0.000	1.000
wages among the highest in the country	40	40.0	49.6	20	45.0	51.0	20	35.0	48.9	0.417	0.519
4. THE UNION ROLE											
workers organized in unions	40	15.0	36.2	- 20	10.0	30.8	20	20.0	41.0	0.784	0.376
one union organization	40	15.0	36.2	20	10.0	30.8	20	20.0	41.0	0.784	0.376
unions concerned with a firm's success	40	10.0	30.4	20	10.0	30.8	20	10.0	30.8	0.000	1.000
5. WORKERS' RISK AVERSION											
prepared to do "more" for the firm	40	85.0	36.2	20	90.0	30.8	20	80.0	41.0	0.784	0.376
would stay with the firm in bad times	40	47.5	50.6	20	50.0	51.3	20	45.0	51.0	0.100	0.752
willing to make finan. Invest. in a firm	40	17.5	38.5	20	30.0	47.0	- 20	5.0	22.4	4.329	0.037

Table 1: Firms in Albania: social capital by productivity

Table 1 (continued): Firms in Albania: social capital by productivity

6. WORKERS PARTICIPATION											
workers are informed	40	70.0	46.4	- 20	70.0	47.0	20	70.0	47.0	0.000	1.000
open dialog with managers	40	55.0	50.4	20	60.0	50.3	20	50.0	51.3	0.404	0.525
workers are members of gov. bodies	40	25.0	43.9	- 20	35.0	48.9	20	15.0	36.6	2.133	0.144
7. INTERNAL TRAINING											
existence of organized forms in the firm	40	57.5	50.1	20	55.0	51.0	20	60.0	50.3	0.102	0.749
more than 50% of workers participate	40	47.5	50.6	20	45.0	51.0	20	50.0	51.3	0.100	0.752
other methods of evaluation than survey	40	42.5	50.1	20	40.0	50.3	20	45.0	51.0	0.102	0.749
8. ON-THE-JOB TRAINING											
existence of organized forms in the firm	40	77.5	42.3	20	75.0	44.4	20	80.0	41.0	0.143	0.705
knowledge transfer among employees	40	67.5	47.4	20	75.0	44.4	20	60.0	50.3	1.026	0.311
successors for most of key employees	40	60.0	49.6	20	70.0	47.0	20	50.0	51.3	1.667	0.197
ADDITIONAL INFORMATION											
B2B	40	40.0	49.6	20	20.0	41.0	20	60.0	50.3	6.667	0.010
more than 25% of export	40	17.5	38.5	20	5.0	22.4	20	30.0	47.0	4.329	0.037
more than 100 employees	40	20.0	40.5	- 20	15.0	36.6	20	25.0	44.4	0.625	0.429
services vs. Manufacturing	40	37.5	49.0	20	30.0	47.0	20	45.0	51.0	0.960	0.327
state ownership	40	0.0	0.0	20	0.0	0.0	20	0.0	0.0		
blockholdings	40	85.0	36.2	20	85.0	36.6	20	85.0	36.6	0.000	1.000

Decision-making. The first question set is the only one where answers are clearly differentiable between the groups of more and less productive firms in our sample. Fifty percent of higher productive firms and 85 percent of lower productive firms have confirmed that the decision-making process about strategic questions of the firm is separated from the operational decision-making process at different levels of the firm. Forty percent of higher productive firms and 75 percent of lower productive firms have answered positive to the second question that top managers and owners made strategic decisions unanimously in the last five years. The third question was answered "yes" by 35 percent of more productive firms and 65 percent of less productive firms, that basic strategic decisions in the firm are coordinated among owners, managers and workers. Differences in the shares of positive answers between the groups of firms are statistically significantly different in all three cases. These results can partly be explained with the difference in size of the firms in the groups. More productive firms are somewhat smaller in terms of how many people they employ, and owners as entrepreneurs do not wish to lose control over key decision-making areas. This way management is delegated tasks nominally, not given any effective empowerment (Wilson and Bates, 2003). Many studies so far have shown that owners of SMEs are often very reluctant to delegating their powers and in the process surrendering a level of control (Smith, 2003, Boeker and Karichalil, 2002).

Adjusting employment. We can observe 88 percent of all firms' adjusted short-term employment to fluctuations in demand in the last five years. Seventy-five percent already approached the desired number of workers. Seventy-three percent of all firms recognize core groups of employees as their competitive advantage. In terms of groups of firms, nearly all higher productive firms (95 percent) used short-term adjustment of employment versus 80 percent of lower productive firms, 85 percent of higher productive firms and 65 percent of lower productive firms approached

the desired number of employees and 80 percent of higher productive firms distinguish the core group of employees as their competitive advantage compared to 65 percent of lower productive firms. Even though we can clearly see that more productive firms have a higher share of positive answers for all three questions, the first two answers are different at close to a 15 percent statistical significance level, while the third one is not statistically significantly different between less and more productive firms. This shows that high productive firms are more effective in adjusting employment to desired levels.

Determining wages. Seventy-eight percent of all firms in the sample have higher wages than the reservation wage. Since the level of unemployment compensation, which is low in Albania, determines the level of reservation wage, this might be the reason for a high share of positive answers in the above category. Workers in 50 percent of all firms also get higher payment than required by the collective agreement contract for the industry. Forty percent of sample firms claim that wages are among the highest in the country. Differences in answers between groups are small and statistically not significant.

These results reflect very dynamic labor market conditions in Albania. Collective agreement contracts exist on a national or regional level for separate industries, but more as exceptions than rules. They are formed between the representative union branch (sometimes a federation agreement between two of the biggest unions is made) on one side and business associations on the other. These contracts determine the minimum wage, compensations for lay-offs, overtime hourly wage rate, length of contract validity and freedom of union activity, on a general level. Enterprises that have been privatized and remained a monopoly (i.e. the energy sector) are not included in these contracts, but instead bargain individually with unions and the government. There are only a few cases of such firms in our sample.

The unions' role. This question set is interesting because the position of unions in Albania is different compared to most of the other countries in the Western Balkan region. Let us first briefly explain the situation in the past to better understand the present situation: During the communist period (until 1991), labor relations were managed in a centralized manner by one labor (union) organization under the lead of the Communist Party, which (supposedly) represented the voice of workers and their interests. The union's function was also to transmit the Party's ideas and instructions to the working class. Labor Law, which managed labor relations, did exist. The main principle was collectivism, but there were also some individualist tendencies in everyday life. A considerable number of workers tried working privately (illegally) subject to their qualifications (tailors, carpenters, bricklayers, painters, etc.).

Not many firms have unions operating as worker representatives at present, which is clearly evident from the results in Table 1. Only 15 percent of firm has workers organized in unions, and all of these firms have workers organized in only one union.³ In 10 percent of cases the union leadership is concerned with increasing productivity and, thus, the firm's competitive position. There are two main reasons for low union membership across the country: 1) workers have no desire to be part of a bureaucratically functioning union organization (this view stems from past experience), and 2) workers do not understand the purpose and importance of joining a union. Unions were oriented more toward political struggles of the past instead of the protection of labor relations. It is also not surprising that unions met great resistance from company owners when they started to depend on private companies.

³If there is more than one union operating in an enterprise, the union that bargains over the collective agreement is the one that has more members. Of course it is prudent for the biggest union to join other unions in the firm, in order to increase bargaining power in bargaining with employers.

Workers' risk aversion. Workers in Albania seem to be very motivated to do "something more" for their firm. This is true for 85 percent of firms. Forty-eight percent of firms have employees that would stay with the firm even if they were offered better employment elsewhere. There are no differences between productivity groups. The last question asked was whether or not most workers would be willing to accept part of their firm's business risk. Workers in both groups were very unsupportive of this notion. Thirty percent of higher productive firms have workers that would be willing to accept part of their firm's business risk, while only 5 percent of lower productive firms have workers willing to accept this type of proposition. The difference is statistically significant at the 5 percent level.

Workers' participation. Seventy percent of firms in the whole sample inform workers about key decisions for the firm. Fifty-five percent have an established open dialogue with workers about key decisions for the firm, and 25 percent of total firms have workers' representatives as members of governing bodies. These percentages are lower than in Slovenian firms, where 94 percent of firms are estimated to regularly inform workers, 83 percent claim to have an established open dialogue with workers, and almost 60 percent of firms have worker representatives in governing bodies (Prašnikar et al., 2010).

The decision makers in Albanian firms are usually managers and owners, rather than workers. Generally, employee-employer relations in a competitive environment are more short-term oriented because workers are often treated (and regard themselves) as a production factor. They generally do not participate in discussions and are not commonly in governing bodies to deliberate on performance and progress of the company. Splitting the sample of Albanian firms, we find 35 percent of higher productive firms and 15 percent of lower productive firms have workers involved in governing bodies, with a statistically significant

difference at the 15 percent level. Previous legislation regarding the management of companies and various public organizations assured a level of workers' participation in governing bodies, but this was later changed.

Although the Labor Law regulates labor relations well, it is commonly violated (i.e. large scale of informal economy). The European Commission demands many improvements in national legislation, also in laws regulating labor relations. In focus are specifically the laws defined by EU directives that regulate the rights of workers' representation in the firm (Communication from the Commission to the European Parliament and the Council, 2010).

Internal training. Almost 58 percent of all firms claim to provide organized training to their employees based on identified needs of the company. When it comes to involving more than half of employees in training programs annually, 48 percent of firms do so. Forty-three percent of firms measure training effectiveness with other methods rather than solely conducting a survey at the end of the training program. There are no statistically significant differences between the groups of higher productive firms and lower productive firms on the topic of internal training.

On-the-job training. Both groups, more and less productive firms, answered very similarly that providing regular on-the-job training (apprenticeship, mentorship, job rotation, etc.) is done in 78 percent of cases, on average. Sixty-eight percent of firms in the whole sample systematically induce knowledge transfer among employees. Out of all firms, 60 percent have successors for most of their key employees, allowing successors to swiftly and effectively replace employees in key positions as needed. Higher productive firms have a share of 70 percent of such firms and lower productive firms have a share of 50 percent of such firms, but these differences are not statistically significant.

By observing additional information in Table 1, we can see some differences in the export orientation and customer orientation of the firms, where higher productive firms export less and make more revenues by selling to end customers (B2C) than do lower productive firms. Only five percent of higher productive firms earn more than 25 percent of total revenues from exports and 20 percent of total revenues from selling to other businesses (B2B). On the other hand, 30 percent of less productive firms make at least 25 percent of revenues from exports and 60 percent of total revenues from selling to other businesses. Thirty percent of higher productive firms belong to the service industry versus 45 percent of less productive firms. There is no diversification in ownership structures because no state-owned companies are in the sample.

4 Discussion

Higher productive companies are operating mostly in the domestic market, exposed to less competition, which may be one of the reasons for achieving higher productivity, if productivity is expressed by value added per employee. Taking into account that higher productive firms in Albania are to some extent smaller in terms of how many people they employ, owners might operate as entrepreneurs that control key decision-making areas.

Differences in wages between the groups of more and less productive firms are not significant. An explanation for such a result might be that firms in Albania operate under very dynamic labor market conditions with owners and/or managers having higher bargaining power than workers, which is also evident in low workers' corporate governance participation. Workers have small bargaining power due to high unemployment and low unemployment benefits (low alternative wage) on one side, and the lack of union representation on the other side. Consequently, the decision makers in Albanian firms are usually owners and/or managers. Workers are often treated solely as a production factor and generally do not participate in discussions on company performance and progress. Despite this situation, our study has shown that higher productive firms have less (business) risk averse workers, and also involve workers in governing bodies more often than lower productive firms, suggesting that workers in smaller (middle-sized) firms are none the less more intertwined with the life of the (entrepreneurial type of) firm.

There are no significant differences between the groups of higher and lower productive firms on the topic of internal training and on-the-job training even though both groups claim they invest in firm-specific human capital.

References

Aoki, M. (1984). The co-operative game theory of the firm. Oxford: Claredon Press.

Aoki, M. (2010). Corporations in Evolving Diversity. Oxford: Oxford University Press.

Allen F. & Gale, D. (2002). Asset Price Bubbles and Stock Market Interlinkages. *Wharton Financial Institutions Center Working Paper* No. 03-27.

Barney, J. (1991). Firm Resources and Sustained competitive advantage. Journal of Management, 17(1), 99-120.

Bernstein, P. (1982). Necessary Elements for Effective Worker Participation in Decision-Making. Workplace Democracy and Social Change. Boston: Porter Sargeant Publishers Inc.

Bloom, N., & Van Rennen, J. (2010). Why do management practices differ across firms and countries. *Journal of Economic Perspectives, 1*, 203-224.

Boeker, W., & Karichalil, R. (2002). Entrepreneurial transitions: factors influencing founder departure. *Academy of Management Journal, 45 (4)*, 818-826.

Communication from the Commission to the European Parliament and the Council: Commission opinion on Albania's application for membership of the European Union. (2010). Brussels: European Commission. Corrado, C., Hulten, C., & Sichel, D. (2009). Intangible Capital and U.S. Economic Growth. Review of Income and Wealth, 55, 661-685.

Essen, M., Oosterhout, J., & Heugens, P. (2012). Competition and cooperation in corporate governance: the effects of labor institutions on blockholder effectiveness in 23 European countries. *Organization Science* (forthcoming).

Ferner, A., & Hyman, R. (1998). Changing industrial relations in Europe. Oxford: Wiley-Blackwell.

Freeman, R., & Lazear, E. (1995). An Economic Analysis of Works Councils. NBER Chapters, In: Rogers, J., & Streecks, W. (eds.), *Works Councils: Consultation, Representation, and Cooperation in Industrial Relations (pp. XX)*. Chicago: National Bureau of Economic Research, Inc.

Flamholtz, E. G., & Randle, Y. (2000). *Growing Pains: Transitioning from an Entrepreneurship to a Professionally Managed Firm.* San Francisco: Jossey-Bass Inc.

Fukao, K., Miyagawa, T., Mukai, K., Shinoda, Y., & Tonogi, K. (2009). Intangible Investment in Japan: Measurement and Contribution to Economic Growth. *Review of Income and Wealth*, 55, 717-736.

Holt, D. H. (1992). Entrepreneurship: New venture creation. Englewood Cliffs: Prentice Hall.

Lepak, D.P., Takeuchi, R., & Snell, S. A. (2003). Employment Flexibility and Firm Performance: Examining the Interaction Effects of Employment Mode, Environmental Dynamism, and Technological Intensity. *Journal of Management, 29(5)*, 681-703.

Lucas, R. (1988). On the Mechanics of Economic Development. Journal of Monetary Economics, 22, 3-42.

Prašnikar, J., Voje, D., Lesjak, J. D., Gjibexhi, L., Raičević, M. (2010). Social Capital as Intangibles in Slovenian Firms. In Prašnikar, J. (ed.), *The role of intangible assets to exit the crisis*, (215-234). Ljubljana: Časnik Finance.

Putnam, R. D., Leonardi, R., & Nanetti, R. Y. (1993). *Making democracy work: Civic Tradition in Modern Italy*. Princeton: Princeton University Press.

Ricketts, M. (2002). The Economics of Business Enterprise: An Introduction to Economic Organisation and the Theory of the Firm. New York: Edward Elgar Publishing.

Smith, M. (2003). From Entrepreneurial to Professional Management: A Transition Model from Australian Manufacturing SME's. *Small Enterprise Research*, 11(1), 3-21.

Wheelen, T. L., & Hunger, D. L. (2010). *Strategic management and business policy*. New Jersey: Prentice Hall.

Williamson, E. O. (1975). Markets and Hierarchies, Analysis and Antitrust Implications: A Study in the Economics of Internal Organization. New York: Free Press.

Williamson, E. O. (1982). Efficient Labor Organizations. Mimeographed.

Wilson, P., & Bates, S. (2003). The Essential Guide to Managing Small Business Growth. Chichester: John Wiley & Sons.

Zupan, N., Farčnik, D., Fišer, J., Stančič Kodarin, L., & Valenčič, G. (2010). Investment in people and organizational flexibility in Slovenian manufacturing firms in 2006-2009. In Prašnikar, J. (ed.), *The role of intangible assets to exit the crisis*, (173-186). Ljubljana: Časnik Finance.
INVESTMENT IN HUMAN CAPITAL

1 Introduction

As Albania is one of the least developed European countries that entered transition from a closed communist system to a market economy two decades ago, the importance of human capital development for economic growth came to be recognized as a top priority by most international organizations, such as the World Bank, EBRD, IMF, and OECD. Many of these organizations prepared country reports and looked at macroeconomic data, which revealed a lack of skilled labor and unsatisfactory levels of education (e.g. World Bank, 2006; EBRD, 2009). Reforms in the education system and the expansion of vocational training have been seen as necessary means for human capital development (Nikolovska, 2008). Conversely, not much has been done to analyze human capital investments and HRM practices at the micro level in Albanian enterprises, even though it has been proven that these practices are positively linked to productivity and sustainable competitiveness (Siebers et al., 2008; Combs et al., 2006).

Therefore the aim of this chapter is to close this gap and shed light on human capital investment at the enterprise level. Following a theoretical framework, which is explained in the next section, we will first look at employment and training practices in Albanian enterprises. Then we will evaluate HRM, management

practices and organizational flexibility, which have also not yet been properly discussed in published research due mostly to difficulties in acquiring the data and the informality of these practices in small size endeavors. While performing the literature review, we found only one systematic study of enterprise level HRM practices by Cucllari et al. (2010) who analyzed enterprises in one region (Korça). They report that the main HRM challenges for enterprises with regard to employment seem to be replacing workers (about 50 percent of enterprises), then hiring new employees (27 percent), and reducing the number of employees (the remainder), which would suggest that enterprises are not satisfied with current employee performance and/or knowledge and skills. They observed low levels of motivation, which they explained to be linked to high turnover. While the majority of enterprises (70 percent) reported that finding, recruiting, and retaining skilled employees is crucial to them, most companies did not provide any training to upgrade employee skills (but training seemed to increases with the size of companies). Rather than invest in training, enterprises prefer to hire trained employees. They also found that small enterprises use mostly informal HRM methods.

These results suggest that HRM is not well developed among Albanian enterprises, and thus gives validity to our approach to study the best companies across economic sectors to explore the link between human capital investment (and supporting HRM, management practices and organizational flexibility) and productivity¹. Our sample consists of 40 Albanian enterprises, of which 25 are manufacturing companies and the remaining 15 are service companies. The average number of employees was 147, with the smallest company employing 4 workers and the largest employing 990 workers (median is 47 workers). Ten companies are

¹ The link between human capital investment (and supporting HRM, management practices and organizational flexibility) and productivity should be investigated with caution, which is later explained.

¹⁴²

foreign owned and 17 companies sell more than half of their products on foreign markets. Besides surveying 40 enterprises to gather both qualitative and quantitative data, we greatly appreciate the insight and feedback from prof. Fatmir Memaj, University of Tirana, who served as a local expert and helped us understand the context and behavior of enterprises. This was especially useful due to the lack of published materials in the English language.

2 Theoretical background

There are many definitions of human capital, but most of these definitions share commonality in employee knowledge and skills, which can be obtained through education and training activities, and also through experience from performing different work-related activities. In the broadest sense, human capital can be defined as all the capabilities (knowledge, skills and abilities) of individuals that are the source of innovation and renewal within enterprises (Stewart, 1997). As such, human capital was recognized as an important source of sustainable competitive advantage (e.g. Barney, 1991) and several researchers in different parts of the world have identified human capital to be an element of the largest subcategory of businesses intangible investment (e.g. Corrado et al., 2009, for the US and the UK; Fukao et al., 2009, for Japan). However, according to Fukao et al. (2009) the effectiveness of these investments largely depends on management practices at the firm level. Thus, we may speculate that if investments are supported by HRM and management practices, which promote further development and productive use of human capital, we may find a stronger impact on productivity and competitiveness. The findings of Bloom and Van Reenen (2007), who gathered firm-specific human capital and organizational change data and explored effects of firm-level activities on the link between human capital and productivity, support this line of thinking.

Another theoretical concept that is useful in exploring human capital and performance link is the widely used AMO (abilities, motivation, opportunity) (Boxall and Purcell, 2008). This model suggests that the greatest effects on performance seem to be achieved when high human capital (abilities) is combined with high levels of motivation and opportunities, which are available to employees to use their human capital effectively. Therefore human resource and general management practices are crucial for increasing the value of human capital through first effective recruitment and selection of highly capable employees, then through training (on and off-the-job) and development, and finally by the retention of valuable employees, especially those with high firm-specific human capital. Also, HRM and other managerial practices such as performance management, motivation, compensation and rewards, leadership, communication, organizational climate, and culture play an important role for effectively managing the productive use of human capital.

Miyagawa et al. (2010) took the research one step further and suggest that besides management and human resource management, practices related to human capital investment and organizational reform as an element of building organizational flexibility may contribute significantly to firm productivity, as was the case for the Japanese firms in their sample. According to the Amiri et al. (2010), with the increase in organizational flexibility and change in organizational structure towards more organic forms, human capital and knowledge productivity also increase.

3 The context of human capital in Albania

With its 2.8² million people, Albania belongs to a group of small European countries. Due to declining mortality and fertility rates, the age structure has changed significantly in the past decades, but compared to other European countries it still has a large share of young population. The median age in Albania is 28.3 years compared to 39 years for the EU (World Bank, 2006). An important phenomenon with regard to population relates to large internal (from rural to urban areas) and external migratory waves (Nikolovska, 2008). Although figures for Albanian emigrants vary (from 600,000 to almost 1,100,000), the fact remains that it is much higher than in other European countries (Vullnetari, 2007). Also, Albania is heavily affected by brain drain since highly educated and skilled employees represent a significant portion of external migrants (Aliaj, 2010).

Of course, migration is closely linked to poverty (as described in the macroeconomic outlook in Redek et al., 2012, in this book) and the labor market, which is marked by relatively high unemployment with the official number of registered unemployed at around 14 percent (World Bank, 2012). It seems that young people and women have the most difficulties in finding jobs, and there is a great deal of long-term unemployment. Although the number of unemployed has been fairly stable over the last decade, local expert suggest that actual unemployment of the working age population is much higher because rural areas

² Preliminary Census of Population and Housing 2011 results show that Albania's population is 2,831,741 inhabitants. The population from the 2001 Census was 3,069,275. It is noted that the population in 2011 decreased by 7.7% in 10 years. It is assumed that the main causes of population decline are large-scale migration and fertility decline (Institute of Statistics, 2012).

¹⁴⁵

are not well covered by statistics, and many people do not even declare themselves as job seekers because they do not believe in existing job opportunities. According to the European Training Foundation's (EFT) labor market analysis, it seems that job creation in formal sectors is rather slow, and labor demand is mostly restricted to the public sector and some private businesses such as banking and, to a smaller extent, construction and tourism (Nikolovska, 2008).

In general most people, especially educated people, consider the public sector attractive for employment because of job security, stability, and better training opportunities, despite lower salaries than in the private sector (Kasimati, 2011). The private sector, on the other hand, consists mostly of micro and small enterprises with simple structures and strong owner influence (usually the owner is also acting as the manager), which diminishes career prospects for educated and skilled workers (Lame and Cela, 2004). The private sector also has a large proportion of informal employment, which is another important characteristic of the Albanian labor market (Nikolovska 2008). It is believed that the majority of workers in the private sector are employed in informal arrangements without proper working contracts and outside the coverage of labor legislation (including getting paid less than minimum wage according to a local expert's observations) and social insurance. Balliu (2010) prepared a report on undeclared labor for ILO and found that three-quarters of people surveyed support informal employment. Only 33 percent of the respondents have concluded employment contracts with their employers (in this estimation is included also the public sector), 28 percent are working without employment contracts, and the rest are unemployed or selfemployed. As was explained by the local expert, formality and concluding contracts are not a part of the Albanian culture, and this may importantly contribute to the fact that 'the Albanian business environment is characterized by the co-existence of rules and spontaneous behavior, informality and formality ...' (Lame and Cela, 2004,

p. 127). It seems that this culture of informality is also maintained by the labor legislation, as there is a provision that employment contracts can be either oral or written. If the contract is oral, the employer has to provide a written contract in 30 days, and if not, the oral contract bears the same validity as a written contract and the employer is fined for not providing a written contract (ICGL, 2011).

Compared to other European countries, Albania has a very high proportion of selfemployed individuals. According to EFT labor market survey data, two-thirds of employment pertains to self-employed, and only one-third is remunerated by an employer (Nikolovska, 2008). Self-employment has family business characteristics and can be found mainly in trade and crafts handed down through generations of families. Albania also stands out regarding the proportion of temporary to permanent employees, which has been found to be 3.68:1, while most other transition countries in the region had fewer temporary than permanent employees (Sahadev and Denmirbag, 2010). One reason for this can be that there is no time limit for temporary contracts in legislative provisions (World Bank, 2011).

Albanian labor legislation has evolved throughout the transition period and in many ways emulates ILO standards and provides a relevant basis for the protection of employee rights and employee well being. The new Labor Code was adopted in 1995, and later amended in 1995 and 2004. Some other important provisions are stipulated in the Law on Employment Promotion adopted in 2006, the amended Health and Safety Law in 2010, and the ratification of several ILO conventions (ICGL, 2011). Despite extensive legislation, the local expert has emphasized that many provisions are not easily enforced due to prevailing tradition and practices, as well as due to a large portion of informal or undeclared labor that is not even covered by these legislations. Nowadays, the leading institution for developing policy and legislation in the area of employment is the Ministry of Labor and Social

Affairs and Equal Opportunities (MoLSAEO). This institution tries to enforce better adherence to legislation by organizing supervision and inspection bodies. However, the effects are still rather limited due to corruption in inspection practices (Balliu, 2010).

With regard to employment protection, Albania's legislation is fairly flexible compared to other South East European countries, especially when we look at hiring (for non-permanent employees) and firing practices (World Bank, 2006). Institutions of industrial relations are evolving slowly and there is not much evidence about the effectiveness of a social dialogue at the national level, or about collective bargaining and management – trade union relations at the company level. According to the opinion of our local expert, the effectiveness of industrial relations at both macro and micro levels is relatively low, mostly due to the lack of tradition of trade unions and employer associations and specifics of the business context (small size of companies and high levels of informality with regard to employment relations).

4 Investments in human capital

Qualitative data was collected for 40 firms in three areas considering HRM and organization: (1) data on human capital and motivation that include data on employee training, regular on-the-job training and knowledge transfer, and employee performance measures and rewards. (2) Data on the organizational climate include data on the general level of satisfaction and loyalty, whereas data on (3) organization flexibility include data on developments in the last five years with respect to the number of hierarchical levels, number of managers, rules and tasks prescribed, job specialization, and autonomy at work.

Regarding the number of employees, data are available from 2002 to 2010 and range from data on 9 firms (in 2002) to data on 40 firms (2009 and 2010). On average we can observe employment growth during the years, except for the year 2007 when the number of employees in the observed firms on average decreased. For example, in 2002 the average number of employees in the sample was 78, where the smallest company employed only 4 employees and the largest 461. In 2010 the average number of employees was 148, the smallest company again employed 4 employees and the largest 990. In 2010 the total number of employees in the sample was 5,901. Due to a very small number of companies in the sample in 2002, we rather focus on the last few years. Overall, 19 companies in the database increased the number of employees in all of the observed years and, out of those companies, more than half (11) were small companies. Twenty-one companies decreased the number of employees in at least one of the observed years (half of them were small companies).

First, we were interested in how investment in human capital differs with respect to independent variables: legal status (limited company, joint stock), industry (manufacturing, service), ownership (domestic, foreign), size (less than 50 employees or more, less than 100 employees and more), market orientation (domestic, foreign), and availability of resources to invest in people (here we presume that those companies that have their employees among the best paid in Albania also have in general more resources to invest in people). Second, we explore how supporting practices differ in the sample of 40 Albanian firms. We focus on performance measurements, organizational climate, and organization flexibility. These management practices are again investigated with respect to independent variables. Third, we investigate the productivity of the firms in the sample and the relationship of independent variables, as well as investments in human capital, supporting management practices, organizational climate, and flexibility with productivity. We report only the significant effects and relationships.

As mentioned above, we observe 40 companies that together employed 5,901 employees in 2010, and the majority of these companies experienced a steady growth in number of employees in the observed period. The majority of companies (85 percent of them) used some kind of flexible employment arrangements, mostly employees working overtime or hiring part-time workers. Because institutional framework allows high levels of flexibility in part-time employment, some companies, such as Golden Pen Generation (see the case study of GPG in Redek et al., 2012, in this book) use employment flexibility as the main element of their competitiveness. With the help of these flexible employment measures, about three-quarters of companies (29) claim that they have achieved the target number of employees. Nearly all companies in the sample (37 of them or 93 percent) claim to use the core employees' model (for more discussion on this model see Prašnikar et al., 2012, in this book).

<i>inacpendient i an tae tee</i> .	-		
Type of investment activity	NO	YES	t-stat
	38%	62%	
		Not among the Among the	
		highest paid in highest paid in	
Organized training based on		Albania Albania	
company needs		0.388 0.818	-3.030***
		(0.501) (0.394)	
		Domestic owners Foreign owners	
		0.566 0.800	-1.315*
		(0.504) (0.421)	
	57%	43%	
		Not among the Among the	
More than 50% of		highest paid in highest paid in	
employees participate in		Albania Albania	
training programs		0.222 0.590	-2.463***
		(0.427) (0.503)	
		Less than 50% trade More than 50%	
		abroad trade abroad	
		0.304 0.588	-1.825**
		(0.470) (0.507)	

Table 1: Investment in organized training and on-the-job training with respect to different independent variables.

SD in brackets. *** significant at 1%, ** significant at 5%, * significant at 10%.

Type of investment			
activity	NO	YES	t-stat
	47%	53%	
		Not among the Among the highest	
Measuring effectiveness		highest paid in paid in Albania	
of training programs		Albania	
		0.277 0.727	-3.087***
		(0.460) (0.455)	
Regular on-the-job			
training	23%	77%	
	35%	65%	
Systematic knowledge		Less than 100 More than 100	
transfer		employees employees	
		0.562 1.000	-2.431***
		(0.504) (0.000)	
	23%	77%	
		Not among the Among the highest	
		highest paid in paid in Albania	
Successors for most of		Albania	
the key employees		0.666 0.863	-1.488*
		(0.485) (0.351)	

Table 1 (continued): Investment in organized training and on-the-job training with respect to different independent variables.

SD in brackets. *** significant at 1%, ** significant at 5%, * significant at 10%.

In investigating organized training and on-the-job training, we observe that companies are more devoted to on-the-job training (Table 1). Around two-thirds of the companies in the sample (62 percent) are aware of the importance of investment in external employee training to assure long-term competitiveness. The companies that invest in training are also very keen to measure the training effectiveness, as half of the companies (53 percent) also measure the effectiveness of training programs. How companies with an ambitious vision are aware of the need for training to increase value added has been described in the cases of EWH and Stella companies in Redek et al. (2012, in this book).

Measuring the effectiveness of training is mostly done through different tests or actual implementation of gained knowledge at work. However, looking at the scope

151

of training results are less encouraging, as only a small proportion of employees are involved in the training activities. Less than half of the companies (43 percent) annually involve more than half of their employees in training programs.

When we look at on the job training, there are even more companies which regularly provide this kind of training to their employees (77 percent). Sixty-five percent of the companies also report having a systematic way of knowledge transfer among employees, mostly through assigning supervisors to new employees in order to guide them and help them acquire necessary experience. From the listed descriptions of knowledge transfer practices, we could speculate that companies mostly use them for new hires and that there is not much organized knowledge transfer activities for other employees who have already been in the company for some time. Interestingly, a very high proportion of 77 percent of the companies believe they have qualified successors for key employees. This would suggest that they do not see systematic knowledge transfer as an important practice for developing successors.

In comparing investment in organized and on-the-job training, we find that there are not many significant differences with respect to independent variables, especially regarding the industry. The most significant differences are regarding the wages of employees. We find that companies with workers who are among the highest paid in the industry invest more in organized training, which is also more often organized for more than half of the employees (in 60 percent of companies more than half of the employees are included in organized training). They also more often measure training effectiveness and have successors for key employees. With respect to ownership, we find that foreign-owned companies are on average more likely to invest in organized training based on the company's needs. Companies that sell more than 50 percent of their products abroad on average

invest in a higher proportion of employees, as we find that those companies invest significantly more to train more than half of employees. Regarding the size of the company, we find that bigger companies, measured by the number of employees, have developed systematic knowledge transfer more than smaller companies.

5 Supporting HRM and management practices

Following research findings by Fukao et al. (2009) and Bloom and Van Renen (2007), it is important to explore human resource and management practices, which are supposed to support the effective development, allocation, and deployment of human capital. In our study we focus on three people-related elements that may affect employee motivation and commitment: performance management, reward systems, and organizational climate (especially employee loyalty).³ The fourth element is organizational flexibility, where we explore if organizational characteristics are evolving into more organic forms, which are characterized by reduced hierarchy and number of managers, less specialization and formalization, and more autonomy.

³ One of the supporting management practices is also participative leadership style (discussed in Prašnikar et al., 2012, and omitted from this analysis) where workers have the option of giving comments, are involved in open dialog when deciding on key issues, or have a representative in a governing body in the firm. We find that a higher proportion of firms that have a more participative leadership style pay higher salaries to their workers.

¹⁵³

5.1 Performance management and reward system

From Table 2 we can observe that 75 percent of companies claim that they measure performance in such a way that they can clearly distinguish between high and low performers.

Measuring performance	NO	YES		t-stat
Performance evaluation to be able to				
distinguish between high and poor				
performers (n=40)	25%	75%		
	10%	90%		
Rewarding high performance (n=40)		Less than 50%	More than 50%	
		trade abroad	trade abroad	
		1.000	0.764	2.592***
		(0.000)	(0.437)	
Using other measures than oral notice for	13%	87%		
poor performers (n=40)		Less than 50	More than 50	
		employees	employees	
		0.809	0.947	-1.311**
		(0.402)	(0.229)	
		Domestic owners	Foreign owners	
		0.833	1.000	-1.378*
		(0.379)	(0.000)	
		Less than 50%	More than 50%	
		trade abroad	trade abroad	
		1.000	0.705	3.017***
		(0.000)	(0.469)	

Table 2: Performance management and rewards

SD in brackets. *** significant at 1%, ** significant at 5%, * significant at 10%.

Interestingly, 90 percent of companies claim to reward better performing workers, although some of those firms have no systematic measures of performance. When it comes to low performance, a very high proportion of companies (88 percent or 35 companies) apply warning signs other than oral reprimands for low performers to let them know of their substandard performance. These are especially companies with more than 50 employees, which have foreign owners and sell in the domestic market. Companies that sell in the domestic market are also more likely to reward high performers. However, we find no significant differences about measuring

154

performance by some standard procedure of evaluation for different sized companies, different ownership, or legal status.

5.2 Organizational climate

With regard to organizational climate, 36 companies, or 90 percent as reported in Table 3, claim that their employee satisfaction is at similar levels as in comparable companies. Companies that are bigger and are more domestically oriented report that the general level of satisfaction would not reach the level of comparable companies. Managers of 31 companies think that their employees are willing to do "something more" for the company, mostly by working overtime or working during holidays. These are mostly companies in the service industry. However, the reported loyalty of employees is relatively low. Namely, in only 15 companies does management believe that employees would remain working even if they received an offer for better employment elsewhere. It seems that employees would be on average more likely to stay in large domestic companies in the service industry and quit jobs in companies that are smaller and with foreign ownership.⁴ As it has been described in case studies in Redek et al. (2012, in this book), many companies intentionally pay employees above the Albanian average to maintain good relations and employee satisfaction, but because even higher than average salaries are still relatively low, it is understandable that perceived loyalty is also low and employees would move to better paying jobs (in or out of Albania) should the opportunity arise.

⁴ If we compare the organization climate of Albanian companies with Slovenian companies, we find that the general level of satisfaction and loyalty perceptions are very similar, but the willingness to do something more is perceived at a slightly lower level in Albania (Zupan et al., 2010). When compared to companies in the Republic of Srpska (BIH), results of Albanian companies are a little better for all three variables (Petković and Zupan, 2012).

¹⁵⁵

Table 3: Organizational climate

Organizational climate	NO	YES		
General level of satisfaction in the				
company is similar to that in other				
similar companies	10%	90%		
		Less than 100	More than	
		employees	100	
			employees	
		0.937	0.750	1.591*
		(0.245)	(0.163)	
		Less than 50%	More than	
		trade abroad	50% trade	
			abroad	
		0.956	0.823	1.384*
		(0.208)	(0.392)	
Most of the employees are willing to do			• • •	
"something more" for the company	23%	77%		
		Manufacturing	Service	
		0.680	0.933	-1.894**
		(0.476)	(0.258)	
Most of the employees would remain			• • •	
working for the company even if they				
got an offer for better (for example				
better paid) employment elsewhere	63%	37%		
		Manufacturing	Service	
		0.280	0.533	-1.614*
		(0.458)	(0.516)	
		Less than 100	More than	
		employees		
		0.312	0.625	-1.647*
		(0.470)	(0.517)	
		Domestic	Foreign	
		owners	_	
		0.433	0.200	1.315*
		(0.504)	(0.421)	

SD in brackets. *** significant at 1%, ** significant at 5%, * significant at 10%.

5.3 Organizational flexibility

Table 4 shows investments in organizational flexibility through changes in the characteristics of organizational structure. On average, companies tended to not

change much or even decrease organizational flexibility by increasing the number of hierarchical levels (40 percent of companies) and increasing formalization through imposing more rules. The majority of companies increased the number of tasks prescribed and also increased job specialization. In nearly half of the companies the number of managers remained about the same, and in nearly half of the companies the number increased. Thus, the overall picture does not show that companies would believe in organizational flexibility as an important element of increasing productivity.⁵

Table 4: Changes in organizational characteristics related to organizational flexibility

Organizational characteristic	Lower/less	About the same	Higher/more
Number of hierarchical levels (n=40)	2	22	16
Number of managers (n=40)	2	19	19
Job specialization $(n=40)$	2	12	26
Number of rules $(n=40)$	4	12	24
Number of tasks prescribed (n=40)	1	13	26
Autonomy at work $(n=40)$	3	22	15

We find that flexibility in the observed years deteriorated the most for domestically oriented companies with more than 50 employees (Table 5). Job specialization, number of rules, and number of tasks prescribed increased for larger companies (also those with some kind of employee participation and lower payment). We do not observe other independent variables that would influence changes in organizational flexibility.

⁵ These results are very similar to those for the Republic of Srpska (BIH), where organizational flexibility has also not improved in the observed period (Petković and Zupan, 2012). However if we compare these results with research on Slovenian manufacturing companies, for example, we find that in contrary organizational flexibility is one of the core management practices supporting HRM (Zupan et al. 2010). In this previous literature we find that in Slovenian manufacturing companies the number of hierarchical levels from 2006 to 2009 had on average decreased, and workers autonomy increased.

T 11	- 0		1 (7 ·	7 ·	7•.
abla	5.1	mannanationa	1 +	102 111	ni	ata
<i>i une</i>	J. ()	ryanizailonai	. 11	e.x.u	111	uv
	- • ~	· S	.11			

Number of hierarchical levels		
Less than 50 employees	More than 50 employees	t-stat
1.190	1.526	-1.889**
(0.511)	(0.611)	
Less than 50% trade aboard	More than 50% trade abroad	t-stat
1.521	1.117	2.296**
(0.593)	(0.485)	
Job specialization		
Less than 50 employees	More than 50 employees	t-stat
1.363	1.689	-1.589*
(0.674)	(0.541)	
Average payment	Significantly higher payment	t-stat
1.739	1.411	1.780**
(0.448)	(0.712)	
Number of rules		
Less than 50 employees	More than 50 employees	t-stat
1.238	1.789	-2.774***
(0.768)	(0.418)	
Average payment	Significantly higher payment	t-stat
2.000	1.411	2.032**
(0.000)	(0.701)	
Not among the highest paid in Albania	Among the highest paid in Albania	t-stat
1.666	1.363	1.421*
(0.594)	(0.726)	
Number of tasks prescribed		
Less than 50 employees	More than 50 employees	t-stat
1.476	1.789	-1.891**
(0.601)	(0.418)	
Less than 50 employees	More than 50 employees	t-stat
1.521	1.764	-1.425*
(0.593)	(0.437)	
Average payment	Significantly higher payment	t-stat
1.782	1.411	2.256**
(0.421)	(0.149)	

SD in brackets. *** significant at 1%, ** significant at 5%, * significant at 10%.

6 Investments in human capital and productivity

Investigating the relationship between productivity measured as the mean productivity in 2008 and 2009 and classifying the companies below and under this mean productivity and medium productivity in 2008 and 2009, we find that above average productive companies in 2008 had more loyal employees than below average productive companies.⁶ Those companies are also more likely to use measures other than oral notices for poor performers. Companies that had above median productivity in 2008 invested more in systematic knowledge transfer than other companies. When investigating productivity in 2009, we find that on average less productive firms in 2009 invested more in regular on-the-job training than more productive, and the employees of these companies were more willing to do something more for the company. We find that none of the measures of organizational flexibility has any relationship with the productivity.

⁶ The investigation of productivity should be explored with caution. As mentioned, our sample consists of the more productive companies, and among those are companies that are operating in niche markets and thus have higher returns and profits. Most of them are export oriented, and the company Golden Pen Generation (GPG) described in Redek et al. (2012, in this book) can serve as an example of one such company. On the other hand, companies are less productive due to refraining from lay-offs. This is mostly motivated by bad publicity and for foreign-owned or privatized companies. One of these examples is the steel producing company Kürüm International described in Redek et al. (2012, in this book).

Below mean						
productivity in						
2008	Above mean productivity in 2008					
65.79%	34.21%					
	Using only oral notice for poor	Using other measures than	t-stat			
	performers	oral notice for poor				
	1	performers				
	0.960	0.769	1.852**			
	(0.200)	(0.438)				
	Employees would leave a	Loval employees				
	company for better paid job	5 1 5				
	0.280	0.615	-2.066**			
	(0.458)	(0.506)				
Below median						
productivity in						
2008	Above median productivity in 200	8				
	No systematic knowledge					
	transfer	Systematic knowledge transfer	t-stat			
	0.578	0.789	-1.395*			
	(0.507)	(0.418)				
Below mean	•	· · · ·				
productivity in						
2009	Above mean productivity in 2009					
52.50%	47.50%					
	No on-the-job training	Regular on-the-job training	t-stat			
	0.857	0.684	1.303*			
	(0.358)	(0.477)				
	No willingness to do something	Willingness to do something	t-stat			
	more for the company	more for the company				
	0.857	0.684	1.303*			
	(0.358)	(0.109)				

Table 6 Relationship between productivity and HRM, motivation, and organizational climate

SD in brackets. *** significant at 1%, ** significant at 5%, * significant at 10%.

7 Discussion and conclusions

The aim of this chapter was to assess the investment in human capital in Albanian enterprises. In our sample we deal with some of the leading companies in their respective sectors, thus we may speculate that on average Albanian companies are lagging behind our observations. On the other hand, looking at the best companies proves advantageous because we are able to depict some patterns of intangible investment with regard to independent variables.

In observing direct investments in human capital, there is a bit less activity with regard to organized training compared with on-the-job training. Around two-thirds of companies claim to provide organized training, and 75 percent provide on-the-job training. However, involvement in training programs does not seem to be wide spread across companies. Only 40 percent of companies reported that they involve more than half of employees in training programs. An inclination towards informality, which we mentioned while describing the context, can also be shown in our results. For example, many companies report to have qualified successors for key positions, even though some of them do not organize training or support formal knowledge transfer. Not surprisingly, we found that larger companies and companies with foreign ownership, or companies exposed more to foreign markets, invest more in human capital. This would suggest that some good practices are already in place and other companies can benefit through sharing knowledge and experience. Here the newly established association of HR professionals (Albanian Human Resource Society) can play an important role (see AHRS, 2012).

Pervasive informality and a reliance on intuition when making people-related decisions are especially evident with reported performance management and reward systems. Namely, almost all companies claim that they reward better performers,

however the number of companies that report to have a performance evaluation mechanism in place to help them distinguish between high and low performers is much lower. Besides rewarding high performance, almost the same high proportion of companies (close to 90 percent) uses more than just an oral notice to deal with poor performers. A similar group of companies that is investing more in training (with foreign ownership, markets abroad and of larger size) also seem to handle poor performance more promptly than others.

A participative leadership style as an important practice supporting human capital use is discussed in Prašnikar et al. (2012, in this book). Here we would like to comment that it is not surprising that the service sector is more open to a participative leadership style than the industry sector. However, it is a bit unexpected that smaller companies (with less than 50 employees), which usually have more informal relationships and more open lines of communication, report lower levels of employee involvement in decision making. It can be that the power of the owner, who is usually also acting as the director in small companies and making important decisions, could present an obstacle for employee involvement. Nevertheless, a general level of satisfaction has been reported as being at a similar level to comparable companies, and even more frequently for smaller companies than for larger ones, which may suggest that employees who go to work for small business owners accept their authority and do not perceive not being involved in decision making as negative. However, investing in human capital without then providing opportunities to employees to express their opinions may not bring full returns on these investments, and even owners of small businesses should consider implementing less authoritarian managerial styles, especially if they deal with highly skilled employees.

Despite a fairly high assessment of employee satisfaction and commitment (willingness of employees to do something more for the company), employee loyalty is not perceived to be high, as about two-thirds of employees would change employers if they received an opportunity for a better paying job. It seems that employees are less loyal to foreign owners than they are to domestic owners. This suggests opportunistic behavior in the Albanian labor force, as laborers would easily change jobs if a better-paid option emerged. Therefore retaining employees obviously presents one important challenge for companies investing in human capital. Or, alternatively, companies should consider options to minimize the risk of losing employees in which they invested.

Among all supporting practices, the most evident obstacle for the more efficient use of human capital lies in company organization. Here we can observe trends pushing organizations further from organic structures. In one way this can be explained by the growing size of companies, which then need more formalization, specialization, and hierarchy (including higher number of managers). But nevertheless, companies should envision ways to increase organizational flexibility. At the moment, the only positive trend pertains to employees getting more autonomy at work.

When we explore the links between productivity and investment in human capital and supporting HRM/management practices, results are mixed and only a few correlations can be depicted. There are some indications that investment in human capital and performance management improve productivity, but further research would be needed to make a stronger case.

We may conclude that the more advanced Albanian firms seem to recognize the need for investment in human capital and that they are also slowly developing

supporting HRM/management practices. Exposure to the foreign business environment and practices (either through foreign ownership or by serving foreign markets) seems to intensify these intangible investments. At the same time, companies still follow the tradition of informality and intuition-based decision making, which could present a problem when businesses experience further growth. Owners/managers will not be able to control their businesses in the same way, and a more professional management approach will be needed. That is why it is important that there is also investment in management training and development in Albania, as well as development in the HRM profession. At this time we did not explore how the HRM function is organized in companies and how it follows professional standards, so this remains one of the areas for further study. Because we have also experienced difficulties in gathering data, maybe another means of obtaining quality information would be through a qualitative case study approach. Namely, there are certainly many opportunities for Albanian companies to further improve effective management of human resources and turn investments in human capital into improved business results. However, due to many specifics of their business context and strong traditions, it would be important not only to import knowledge from the international milieu (as many international organizations are trying to do), but also to identify and transfer Albanian best practices. Our research suggests that there certainly are cases from which other companies can learn.

References

AHRS (Albanian Human Resource Society). (2012). Retrieved April 20th from www.albaniahr.org.

Aliaj, E. (2010). From brain drain to brain gain: The case of Albania. Conference proceedings: An Enterprise Odyssey. Zagreb: University of Zagreb, Faculty of Economics and Business, 402-408.

Amiri, A. N., Ramazan, M., & Omrani, A. (2010). Studying the Impacts of Organizational Organic Structure on Knowledge Productivity Effective Factors Case study: Manufacturing Units in a Domestic Large Industrial Group. *European Journal of Scientific Research, 40 (1)*, 91-101.

Balliu, Q. (2010). *Combating Undeclared Work through Social Partnership in Albania*. ILO report. Geneva: International Labour Organisation.

Barney, J. (1991). Firm Resources and Sustained competitive advantage. Journal of Management, 17(1), 99-120.

Bloom, N., & Van Reenen, J. (2007). Measuring and Explaining Management Practices Across Firms and Countries. *The Quarterly Journal of Economics*, 122(4), 1351–1408.

Boxall, P., & Purcell, J. (2008). *Strategy and Human Resource Management*. New York: Palgrave Macmillan.

Chin-Ju, T., Edwards, P., & Sangupta, S. (2010). The association between organizational performance, employee attitudes and human resource management practices. *Journal of General Management*, 36(1), 1-20.

Code of Labor of Republic of Albania. (1995). Unofficial translation prepared for the ILO. Retrieved March 24th 2012 from http://www.ilo.org/dyn/natlex/docs/SERIAL/41344/63433/F1167646799/ALB41344.P DF.

Combs, C., Liu, Y., Hall, A., & Ketchen, D. (2006). How much do high performance work systems matter? A meta analysis of their effects on organizational performance. *Personnel Psychology*, 59(3), 501-528.

Corrado, C., Hulten, C., & Sichel, D. (2009). Intangible Capital and U.S. Economic Growth. Review of Income and Wealth 55, 661–685.

Çuçllari, F., Cini, M., & Gabeta, A. (2010). Labour Market and Human Resources Development: An Analysis of Findings from the Albania Survey of Human Resources Development. *European Research Studies*, 13(1), 119-128.

EBRD. (2009). *Strategy for Albania 2009-2012*. Retrieved March 24th 2012 from http://www.ebrd.com/downloads/country/strategy/albania.pdf.

Fukao, K., Miyagawa, T., Mukai, K., Shinoda, Y., & Tonogi, K. (2009). Intangible Investment in Japan: Measurement and Contribution to Economic Growth. *Review of Income and Wealth*, 55, 717–736.

ICGL. (2011). The international comparative legal guide: Employment & Labour Law. Retrieved March 24th 2012 from http://www.iclg.co.uk/khadmin/Publications/pdf/4358.pdf.

Kasimati, M. (2011). Job satisfaction and turnover under the effect of person organization fit in Albanian public organizations. *Journal of East European Management Studies, 16 (4)*, 315-337.

Lame, P., & Çela, H. (2004). Human resource development – A key factor for the sustainable development of Albania. *Management*, 9(2), 117-129.

Miyagawa, T., Lee, K., Kabe, S., Lee, J., Kim, H., Kim, Y., & Edamura, K. (2010). *Management Practices and Firm Performance in Japanese and Korean Firms -An Empirical Study Using Interview Surveys.* RIETI Discussion Paper Series 10-E-013.

Musaraj, A. (2011). Albania, the human factor and sustainable development: Lessons from the present. Academicus. International Scientific Journal, 35-41.

Nikolovska, M. (2008). Human resource development: country analysis – Albania. European Training Foundation Working Paper.

Petković, S., & Zupan, N. (2012). HRM and organization. In Prašnikar, J., & Knežević Cvelbar, L., *Intangible assets as a potential for growth in Republic of Srbska*, (61-76). Ljubljana: Ekonomska fakulteta.

Prašnikar, J., Memaj, F., & Voje, D. (2012). Social capital and corporate governance. In Prašnikar, J., Redek, T. and Memaj, F. (eds.), *Albania: The Role of Intangible Capital in Future Growth*, (73-88). Ljubljana: Ekonomska fakulteta.

Redek, T., Memaj, F., Prašnikar, J., & Trobec, D. (2012). Albania: Two decades of economic development at a glance. In Prašnikar, J., Redek, T. and Memaj, F. (eds.), *Albania: The Role of Intangible Capital in Future Growth*, (4-59). Ljubljana: Ekonomska fakulteta.

Sahadev, S., & Demirbag, M. (2010). A comparative analysis of employment practices among post-communist and capitalist countries in South Eastern Europe. *Employee Relations, 32(3)*, 248-261.

Siebers, P. A., Aickelin, U., Battisti, G., Celia, H., Clegg, C., Fu, X., De Hoyos, R., Iona, A., Petrescu, A., & Peixoto, A. (2008). *Enhancing Productivity: The Role of Management Practices*. Nottingam: The University of Nottingam.

Stewart, T.A. (1997). Intellectual Capital. London: Nicholas Brearley Publishing.

World Bank (2011). *Doing business, Employing workers in Albania (2011)*. Retrieved March 24th 2012 from http://www.doingbusiness.org/data/exploreeconomies/albania/employing-workers/.

World Bank. (2006). Albania: Labour market assessment. World Bank Report No. 34597-AL.

World Bank. (2012). *World Databank*. Retrieved March 24th 2012 from http://databank.worldbank.org/.

World Economic Forum. (2011). *The Global Competitiveness Report 2010-2011*. Retrieved March 18th 2012 from http://www.weforum.org/reports/global-competitiveness-report-2011-2012.

Vullnetari, J. (2007). Albanian migration and development: State of the art review. IMISCOE Working paper No. 18.

Zupan, N., Farčnik, D., Fišer, J., Stančič Kodarin, L., Valenčič, L. (2010). Investment in people and organizational flexibility in Slovenian manufacturing firms in 2006-2009. In Prašnikar, J. (ed.) *The role of intangible assets in exiting the crisis.* (173-186). Ljubljana: Časnik Finance.



FIRMS' FINANCIAL POLICIES

1 Introduction

This chapter presents an analysis of financial policies of a sample of 40 firms in Albania characterized by market imperfections that encompass concepts such as asymmetric information and imperfect capital markets that are widely present in developing and peripheral economies such as Albania, and reflect the level of development of financial markets.

The main elements of the financial policies studied were capital structure and capital budgeting. Using the analysis of financial policies, the goal of the study was to explain the financial behavior of the observed firms and to suggest the level of development of financial markets in Albania. The chapter is structured as follows.

Firstly we searched for evidence of the pecking order hypothesis and financial hierarchy of sources for financing future investments.

Secondly, we observed the importance of capital budgeting methods used in financing decisions and imposed concerns over ranking and usage of methods.

Thirdly, we compared the financial behavior of Albanian firms with the behavior of firms in the Republic of Srpska based on observations in a similar study. The goal was to compare elements of financial policies to observe where Albania can place itself in terms of financial sophistication and development of financial markets compared to the Republic of Srpska, which has also recently undergone transition to a market economy.

2 Pecking order hypothesis and financial hierarchy

Modern capital structure theory began with Modigliani and Miller in 1958. Until then, capital structure theory consisted of loose assertions about investor behavior rather than constructed models that could be statistically tested (Brigham and Daves, 2004). In their study, Modigliani and Miller (1958) addressed capital structure in a rigorous and scientific way and, under several simplifying assumptions¹, proved that a firm's value is not unaffected by its capital structure.

Although both academicians and practitioners have addressed concerns over the validity of their models, Modigliani and Miller have set off a series of studies and research concerning capital structure. Recognizing and considering factors disregarded by Modigliani and Miller, such as asymmetric information, bankruptcy costs, agency costs, and imperfect capital markets, modern capital structure theory has developed into two categories: trade-off theory and pecking order hypothesis.

¹ Assumptions encompass the following: no taxes, no bankruptcy costs, perfect capital markets, symmetric information, riskless debt, firm's EBIT not affected by use of debt. After the first study published in 1958, Modigliani and Miller have altered the original assumptions and incorporated corporate tax into their study published in 1963.

¹⁶⁹

The trade-off theory implies trading off the benefits of debt financing (tax advantage) against agency costs and the risk of bankruptcy (Brigham and Daves, 2004).

The pecking order hypothesis implies financial hierarchy in the use of raising funds, but does not explain a firm's optimal capital structure. It originated from Donaldson's description of financial practices in 1961, in which he observed that firms preferred internal financing and avoided issuing stock. Myers (1984) and Myers and Majluf (1984) theoretically justified the pecking order hypothesis using concepts such as asymmetry of information. According to the pecking order hypothesis, a firm will follow a financial hierarchy due to information asymmetry, where internally generated financing is preferred to external financing, and when external financing is required, debt is preferred to equity.

2.1 Evidence of financial hierarchy and the pecking order hypothesis in Albania

Our purpose was to observe financial hierarchy and evidence of the pecking order hypothesis in the sample of 40 firms that returned questionnaires with a completed capital structure section. In it respondents were asked to evaluate the attractiveness of 6 long-term financing sources for financing future investment projects on a 1 to 5 scale, 5 meaning the most attractive. Results are presented in Table 1.

Table 1: Financial hierarchy of sources for financing future investments

Sources of capital by order of preference	Average importance
(40 firms)	on a 1 to 5 scale
Internal equity (retained earnings)	3.00
Long-term bank loans	2.00
External equity (issuance of new shares)	1.73
Convertible debt	1.40
Preferred shares	1.25
Convertible preferred shares	1.20

On average respondents ranked retained earnings as the most attractive long-term financing source for financing future investment projects, which is according to the pecking order hypothesis (internal sources are preferred to external sources). Longterm bank loans were on average ranked as the second most attractive source with an average importance significantly lagging behind internal equity, and external equity was ranked as the third most attractive source on average, which is again in accordance with the hypothesis (external debt before external equity). Other remaining sources, convertible debt, preferred shares, and convertible preferred shares follow in a corresponding order, each having a low average importance. Results imply the existence of financial hierarchy in the financing decisions of sample firms in Albania in complete accordance with the pecking order hypothesis.

2.2 Capital budgeting procedures

Capital budgeting is an important tool in deciding which investments add to the firm's value and in choosing the right investment option given other alternatives. Capital budgeting is therefore the most important task for financial managers and their staff. Poor capital budgeting can have severe financial consequences: if a firm invests too much, it will incur unnecessarily high depreciation and other expenses, however if it does not invest enough, inadequate capacity and outdated equipment can lead to a reduction in market share and the loss of customers and competitiveness (Brigham and Daves, 2004). Therefore, it is of crucial importance that firms use appropriate methods to screen for projects that add value, and to reject projects that undermine firm value.

Our purpose was to observe the importance of the evaluation methods used in financing decisions in the sample of 40 firms. We asked the respondents to state which of the 4 evaluation methods (internal rate of return (IRR), net present value

(NPV), payback period and accounting-based methods) are in use in their companies. If a particular method was used, respondents were asked to mark the importance of the method on a 1 to 5 scale, 5 meaning the most important. Results are presented as a frequency distribution in a following figure.



Figure 1: Frequency distribution of importance of evaluation methods used in financing decisions

Using the frequency distribution data in Figure 1, we evaluated the average importance for each evaluation method. Results are presented in the following table.

Table 2: Average importance of evaluation methods used in financing decisions

Capital budgeting method (40 firms)	IRR	NPV	Payback period	Accounting-based methods
Average importance of a method on a 1 to 5 scale	4.00	2.94	3.89	4.24
Percentage of companies using a method	45.0	40.0	45.0	52.5

Results indicate that the most frequently used methods are accounting based with 52.5 percent, or 21 firms, using the method, followed by internal rate of return and payback period, both used in 45.0 percent, or 18 firms. The least frequently used method was net present value, which was used in 40.0 percent, or 16 firms. Only 2 firms stated that they also use other types of evaluation methods when evaluating financing decisions.

Accounting-based methods are the most important methods with an average importance of 4.24, followed by internal rate of return with an average of 4.00, payback period with an average of 3.89, and net present value with an average importance of 2.94. The results of our study raise three concerns.

The first concern is the relatively low percentage of firms that use methods in general. Looking at similar recent studies, carried out in countries in the same Balkan region, concerning the importance of capital budgeting methods, Albania falls behind both: the Republic of Srpska and Slovenia. The study of 58 companies in the Republic of Srpska carried out by Prašnikar et al. (2012a), as presented in the table below, shows that 63.8 percent to 72.4 percent (depending on the method) of observed firms on average use evaluation methods. The study of Valentinčič et al. (2010) of 56 Slovenian manufacturing firms shows a considerably higher percentage of evaluation method usage²; internal rate of return and net present value are both used by 87.2 percent of firms on average, and payback period is used on average by 94.4 percent of firms.

 $^{^2}$ Only relatively less important accounting methods have a relatively lower percentage (60.0 %).

¹⁷³

Percentage of companies using a method by country	IRR	NPV	Payback period	Accounting- based methods
Albania	45.0	40.0	45.0	52.5
Bosnia (Republic of Srpska)	63.8	63.8	72.4	63.8
Slovenia	87.2	87.2	94.4	60.0

Table 3: Comparison of the usage of capital budgeting methods in firms by countries of the Balkan region

The study of capital budgeting methods used by the Fortune 500³ industrial companies, carried out by Bierman (1993), indicate a relatively high usage of capital budgeting methods. According to the study, payback period was used by 84 percent of surveyed companies, net present value was used by 85 percent of companies, and internal rate of return was used by 99 percent of the surveyed companies. Furthermore, most companies gave the highest weight to the discounted cash flow methods (NPV and IRR). Similarly, Kester at al. (1998) in a multinational study of the Asia-Pacific found that internal rate of return and net present value were the most popular capital budgeting methods for large firms in that region.

The second concern is the popularity of accounting-based methods compared to internal rate of return and net present value in terms of average importance and percentage of companies using a method. Accounting-based methods (such as accounting rate of return) are not considered superior evaluation methods in neoclassical financial theory⁴, namely accounting-based methods ignore the time value of money and often focus on net income rather than cash flow and, therefore, is considered a flawed measure. Most early studies, like the one by Miller

³ The Fortune 500 is a list that ranks the top 500 U.S. closely held and public corporations as ranked by their gross revenue after adjustments. The list is constructed and published by Fortune magazine annually; the first list was published in 1955.

⁴ Neoclassical financial theory regards net present value as the best single measure, followed by internal rate of return, which many decision makers prefer. Literature nevertheless points out that firms should consider all of the measures, which nowadays, with the use of modern technology, can easily be obtained since each gives a different piece of important and relevant information.

(1960) and Schall et al. (1978), report payback technique as the most popular and preferred capital budgeting method, which was attributed to the lack of financial sophistication and limited use of computer technology, while the early study of Istvan (1961) reports a preference for accounting rate of return as a superior method of capital budgeting. Later studies, like the one by Bierman (1993), indicate that no firms used any accounting-based methods, nor did firms use payback period as their primary method. Firms instead gave the greatest weight to discounted cash flow methods such as internal rate of return and net present value.

Although Ryan and Ryan (2002) claim that throughout the literature net present value has always trailed internal rate of return in management preference our survey shows a large discrepancy between the methods' average importance (IRR having average importance of 4.0 and NPV having average importance of 2.9). Our survey also shows that the correlation between the usage of internal rate of return and net present value, measured as Pearson's correlation coefficient, is 0.8, indicating a strong, linear and positive correlation. In other words, not every company that uses internal rate of return uses net present value and vice versa, which is odd considering the conceptual nature of both methods and the calculation procedures of obtaining outcomes based on these two methods. Other studies, like the one of Prašnikar et al. (2012a) show that the correlation between the usage of internal rate of return and net present value, measured as Pearson's correlation coefficient, is 1, meaning a perfect, linear and positive correlation. In other words, all firms in the sample that used internal rate of return also used net present value and vice versa, which is in accordance with financial theory and the conceptual nature of both methods.

Our study shows a relatively low percentage of companies that use methods in general. When using methods, firms on average give relative importance on

accounting-based methods. Our study also shows a discrepancy in the correlation between the firms' usage of internal rate of return and net present value, which poses a concern over the conceptual understanding of both methods by firms. The findings reflect the level of development of financial markets in Albania. While firms in the United States in 1993 showed a relatively high usage of capital budgeting methods and emphasized the importance of discounted cash flow methods, as indicated by Bierman (1993), which is attributed to the long period of market economy that existed in the United States even as early as 1993, our analysis of capital budgeting methods in Albania suggest a level of development of financial markets closer to transition economies in the Balkan region. The next section compares the elements of financial policies of firms in Albania to a similar study of firms in the Republic of Srpska to observe where Albania can place itself in terms of financial sophistication and development of financial markets.

3 Financial policies in Albania in comparison with the Republic of Srpska

This section compares the financial behavior of Albanian firms with behavior of firms in the Republic of Srpska that were observed in a similar study carried out by Prašnikar et al. (2012a). Our purpose was to compare elements of financial policies to observe where Albania can place itself in terms of financial sophistication and development of financial markets compared to the Republic of Srpska, which has also recently undergone transition to a market economy.

Comparisons of firms' financial policies in Albania to policies of firms in the Republic of Srpska are summarized in Tables 4 and 5.
Sources of capital by order of preference (average importance on a 1 to 5 scale)	Albania	The Republic of Srpska
Internal equity (retained earnings)	3.00	4.00
Long-term bank loans	2.00	2.64
External equity (new shares)	1.73	2.14
Convertible debt	1.40	1.93
Preferred shares	1.25	1.53
Convertible preferred shares	1.20	1.52

Table 4: Comparison of financial hierarchy of sources for financing future investments by country

Source: Prašnikar et al. (2012a); own calculations.

Table 4 shows a comparison of financial hierarchy of sources for financing future investments. Albanian firms show equal financial hierarchy and presence of the pecking order hypothesis as the sample of firms in the Republic of Srpska. In terms of contrasts in average importance, Albanian firms and firms in the Republic of Srpska give, by far, the highest preference to internal equity, and much less to long-term bank loans and external equity capital.

Table 5: Comparison of usage of evaluation methods used in financing decisions

Percentage of companies using a method	Albania	The Republic of Srpska	Cluster (manuf. firms)
IRR	45.0	63.8	44.4
NPV	40.0	63.8	44.4
Payback period	45.0	72.4	61.1
Acc. based methods	52.5	63.8	55.6

Source: Prašnikar et al. (2012a); own calculations.

Table 5 shows a comparison of usage of evaluation methods used in financing decisions. Albanian firms on average use evaluation methods much less than firms in the Republic of Srpska. This suggests that Albanian financial markets are less developed than markets in the Republic of Srpska. Even more, the usage of evaluation methods in Albania resembles the usage of methods of a cluster of

manufacturing firms⁵ in the Republic of Srpska identified by Prašnikar et al. (2012b) and presented in the fourth column of Table 5. The cluster represents the poorest performing group of firms among all identified clusters in the Republic of Srpska today, and illustrates the strongest form of market imperfections and lowest end in the level of development of financial markets in the Republic of Srpska. Of the firms in the cluster, 44.4 percent used internal rate of return and net present value, 61.1 percent used payback period, and 55.6 percent of firms used accounting-based methods as evaluation methods, which is closer to Albanian firms than the whole sample of firms in the Republic of Srpska. This implies that the level of development of financial markets in Albania resembles the lowest end in the level of development of financial markets in the Republic of Srpska as noted in its cluster of worst performing firms.

Average importance of a method (on a 1 to 5 scale)	Albania	The Republic of Srpska
IRR	4.00	4.00
NPV	2.94	3.59
Payback period	3.89	4.05
Acc. based methods	4.24	3.59

Table 6: Comparison of average importance of evaluation methods used in financing decisions

Source: Prašnikar et al. (2012a); own calculations.

Table 6 shows a comparison of average importance of evaluation methods used in financing decisions. While payback period is the most important method in the Republic of Srpska, on average, and accounting-based methods are the most

⁵ The cluster of manufacturing firms consists of mainly manufacturing firms that, on average, have lower productivity rates, lower investment in human capital, lower worker wages, and show high short-term adjustment of labor to shocks. A common characteristic of a cluster is low coordination of basic strategic decisions between owners, managers, and workers on average. The prevailing legal status of firms in a cluster is limited company. The ownership structure shows that the majority of firms are privatized, former socially owned firms.

¹⁷⁸

important in Albania, on average, both samples attribute less importance to discounted cash flow methods.

Albania, originating from a very rigid social model with strict central controls, was one of the last countries that stepped on the path of transition. The first serious steps were initiated in 1991 with price liberalization, investment deregulation, macroeconomic stabilization, land distribution and small scale privatization, followed by the privatization of larger firms in 1995 and invitation of foreign capital at the end of the decade. The second decade was marked by a continuance of reforms improving the business environment and rule of law. Even today, the process of transformation has yet to be finished, especially in the field of liberalization, privatization, structural reforms, business environment, and infrastructure (Redek et al., 2012, in this book). Although Albanian firms show resemblance to firms in the Republic of Srpska in terms of financial behavior, the comparison, especially of the usage of evaluation methods, implies that Albanian financial markets are less developed and market imperfections are more widely present. This is attributed to the fact that Albania is still in the process of transition, and according to Redek et al. (2012, in this book) has a comparative closed market and underdeveloped financial system.

4 Conclusion

Our analysis of financial policies of 40 firms in Albania shows that an average firm does not pursue the goal of maximization of shareholder value; rather it shows the existence of market imperfections that are widely present in developing and peripheral economies such as Albania. Two main elements imply such financial behavior.

Observing capital structure, firms rank internal equity (retained earnings) as the most attractive long-term financing source for future investment projects on average, followed by long-term bank loans and external equity (issuance of new shares). Other sources, convertible debt, preferred shares, and convertible preferred shares follow in a corresponding order, each having low average importance. Results indicate the existence of financial hierarchy of sources and existence of the pecking order hypothesis that contradicts the goal of maximization of shareholder value because reliance on costly equity does not lead to minimal cost of capital, but rather implies information asymmetry.

In observing capital budgeting and the importance of four evaluation methods that were used in financing decisions, an average firm ranks accounting-based methods as the most important and most frequently used methods (52.5 percent of all firms use it). In terms of importance, internal rate of return was ranked second, payback period third, and net present value fourth. In terms of usage, internal rate of return and payback period were used in 45.0 percent of firms, while net present value was used in only 40.0 percent of firms in the sample. Our study indicates a relatively low percentage of firms that use methods in general; popularity of accounting-based methods compared to internal rate of return and net present value in terms of average importance and percentage of companies using a method; and discrepancies between usage and importance of internal rate of return and net present value. Our conclusions reflect the level of development of financial markets in Albania.

Finally, we compared the financial behavior of Albanian firms with the behavior of firms in the Republic of Srpska and found that although Albanian firms show resemblance to firms in the republic of Srpska in terms of financial behavior, especially in the usage of evaluation methods, the comparison implies, however,

180

that Albanian financial markets are even less developed and market imperfection is more widely present. Albanian firms have a difficult time making financial decisions. It seems that the firms rely mainly on the capital of their owners who are striving to achieve the highest short-term profit.

References

Bierman, H. (1993). Capital Budgeting in 1993: A Survey. Financial management, 24, Autumn.,

Brigham, E. F., & Daves, P. R. (2004). Intermediate Financial Management (8th ed.). Independence, K.Y.: Thomson South Western.

Donaldson, G. (1961). Corporate Debt Capacity, Division of Research. Graduate School of Business Administration, Harvard University.

Kester, G. W., Chang, R. P., Echanis, E. S., Haikal, S., Isa, M. M., Skully, T. M., Tsui, K., & Wang, C. (1998). Capital Budgeting Practices in the Asia-Pacific Region: Australia, Hong, Kong, Indonesia, Malasia, Philippines, and Singapore. *Financial Practice and Education, (Spring/Summer)*, 25-33.

Istvan, D. F. (1961). *Capital-Expenditure Decisions: How They Are Made in Large Corporations*. Bureau of Business Research, Indiana University.

Miller, James H. (1960). A Glimpse at Practice in Calculating and Using Return on Investment. N.A.A. Bulletin (now Management Accounting), June, 65-76.

Modigliani, F., & Miller, M. H. (1958). The Cost of Capital, Corporation Finance, and the Theory of Investment. *American Economic Review, June 1958*, 261-297.

Modigliani, F., & Miller, M. H. (1963). Corporate Income Taxes and the Cost of Capital: A correction. *American Economic Review, June 1963*, 433-443.

Myers, S. (1984). The capital structure puzzle. Journal of Finance, 39, 575-592.

Myers, S. & Majluf, N. (1984). Corporate financing and information decisions when firms have information that investors do not have. *Journal of Financial Economics*, 13, 187-221.

Prašnikar, J., Valentinčič, A., Mikerović, D., & Ušlakar, A. (2012a). Financial policies of firms in the Republic of Srbska. In Prašnikar, J., & Knežević Cvelbar, L., *Intangible assets as a potential for growth in Republic of Srbska*, (138-158). Ljubljana: Ekonomska fakulteta.

181

Prašnikar, J., Kondić, N., Mikerević, D., & Voje, D. (2012b). Social capital and corporate governance. In Prašnikar, J., & Knežević Cvelbar, L., *Intangible assets as a potential for growth in Republic of Srbska*, (47-60). Ljubljana: Ekonomska fakulteta.

Redek, T., Memaj, F., Prašnikar, J., & Trobec, D. (2012). Albania: Two decades of economic development at a glance. In Prašnikar, J., Redek, T. and Memaj, F. (eds.), *Albania: The Role of Intangible Capital in Future Growth*, (4-59). Ljubljana: Ekonomska fakulteta.

Ryan, P. A., & Ryan, G. P. (2002). Capital Budgeting Practices of the Fortune 1000: How Have Things Changed? *Journal of Business and Management, 8(4), 355-364*.

Schall, L. D., Sundem, G. L., & Geijsbeek, W. R. (1978). Survey and Analysis of Capital Budgeting Methods. *Journal of Finance 33 (1),* 281-288.

Valentinčič, A., Marinšek, D., Mally Buh, G., & Simoneti, N. (2010). Financial policies of manufacturing firms in Slovenia in 2006-2009. In Prašnikar, J. (ed.), *The role of intangible assets in exiting the crisis*. Ljubljana: Časnik Finance.

182